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Chapter 6
Gasifier Fabrication and Manufacture

6.1 Introduction

Gasifier construction is arelatively simple task and can
be accomplished in any well-equipped shop using
basic sheet metal and welding assembly techniques. In-
deed, the task is so simple that it was possible for the
countries of wartime Europe to construct almost one
million gasifiers in just a few years in spite of wartime
shortages (Egloff 1943; Gengas 1950). Nevertheless, a
number of new materials and fabrication techniques
have become available since World War I, and we shall
call attention to these improvements in this discussion.

According to Kaupp (1984a), “the construction of a
small gasifier, including the purification system, does
not require sophisticated equipment or highly skilled
mechanics. It can be built in workshops comparable to
the auto repair shops found in most third world
countries.”

Fabrication refers to the construction of a single
gasifier for use or for an experiment. Manufacture com-
mences when one undertakes the construction of a
number of identical units.

A general discussion of drilling, welding, and assem-
bly procedures pertinent to gasifier fabrication and
manufacture is beyond the scope of this manual. In-
stead, we shall comment upon specific techniques of
fabrication, as well as the wide range of materials
specifically applicable to gasifiers.

6.2 Materials of Construction

Gasifiers are usually constructed from commercially
available materials such as steel pipe, sheet, and plate.
When choosing materials, one should (where possible)
select those that are readily available and use off-the-
shelf equipment and materials that are available in bulk
quantities. One should avoid exotic alloys, special
shapes, and custom fabrication techniques that require
large initial setup and tooling costs, except in cases
where their use is justified.

Smaller, atmospheric-pressure gasifiers require a min-
imum metal thickness of 20-gauge, with double-thick-
ness reinforcements extending a few centimeters (1 in.)
around all fittings and fastenings (Freeth 1939). The
maximum mild-steel service temperature is 480°C
(900°F) (MASEC). Although the metal temperatures en-
countered in well-designed air gasifiers do not usually
exceed the softening point of mild steel, certain stain-
less steels or inconel may give the extra temperature
resistance necessary for critical areas such as the grate,
hearth, or nozzles.

Some of the mild-steel components may suffer chemi-
cal corrosion in certain parts. Corrosion is likely to
occur in areas where water condenses or collects since
gasifier water often contains organic acids. Water col-
lection is especially a problem in regions such as the
upper magazine of Imbert gasifiers, as well as in some
wet-scrubber systems. In these instances, the steel
should be replaced by corrosion-resistant materials
such as copper, brass, epoxy lined steel, or stainless
steel as required. Stainless steel usually costs two to
three times as much as mild steel and requires inert gas
welding techniques. Copper and brass cost five times
as much as mild steel but can be joined by brazing or
hard-soldering using an acetylene torch. Aluminum is
particularly vulnerable to corrosion in alkaline
environments, and its use should be avoided there.

Some wartime gasifiers, particularly the stationary
ones, contained massive but fragile firebrick insula-
tion. We are fortunate today to have lightweight in-
sulating materials based on spun alumino-silicate that
are capable of withstanding temperatures up to 1500°C,
far beyond the requirements of gasifiers. The alumino-
silicate insulation also offers many times the durability
and heat-flow resistance of firebrick, at a fraction of the
weight (Perry 1973). It is relatively inexpensive and is
available in a wide variety of forms. The 2- to 5-cm-
thick felt blankets and vacuum preformed cylinders (or
“risa sleeves”) are particularly recommended for in-
sulating the reaction zone. “Moldable ceramics” that
come as a wet putty can be shaped to corners and edges,
and thus are also very useful.

Finally, plastics can be used in certain applications.
Some plastic pipe will perform acceptably up to the
boiling point of water, is more flexible than metal pipe,
and will not corrode. Plastic liners, such as epoxy
paints, can sometimes provide the corrosion resistance
needed in critical areas, provided temperatures are not
greater than 120°C.

6.3 Methods of Construction

A gasifier is built much like a water heater, and the
same methods of construction are used. The workshop
should be equipped with tools for performing tasks
such as shearing sheet metal, rolling cylinders and
cones, drilling, riveting, grinding, painting, sawing,
tube cutting, and pipe threading.

An oxyacetylene torch is valuable for cutting and weld-
ing tasks, but an arc welder is preferred for mild-steel
welding. When aluminum, stainless steel, or inconel is

48 Handbook of Biomass Downdraft Gasifier Engine Systems



used, it will be necessary to use either TIG (tungsten-
inert gas) or MIG (metal-inert gas) welding techniques.

All seals must be made gas-tight; threaded and welded
fittings are preferred at all points, and
exhaust-pipe-type gaskets can be used if necessary.
High-temperature, anti-sieze pipe dope should be used
on all pipe joints. High-temperature applications will
require ceramic fiber or asbestos gaskets. Silicone
sealant is appropriate at temperatures below 300°C and
rubber or Viton “O” rings and gaskets will perform
excellently at room temperature. The system should be
leak-tested before the initial startup, as well as after
modifications. Leak-testing is accomplished by
plugging the system and pressurizing it to 25 cm
(10 in.) of water with a blower. A thick soap solution is
applied to all fittings and joints, and they are checked
for emerging soap bubbles. Leak-testing should also be
performed as a standard test in the regular maintenance
schedule.

6.4 Sizing and Laying out the Pipes

When designing a gasifier, it is important to keep the
pressure drop in the system as small as possible.
Because there are unavoidable pressure drops
associated with the gasifier, the cyclone separator, and
the cleanup system, it is very important to use
adequately sized pipe. The pressure drop associated
with standard runs of pipes is shown in Fig. 6-1 (Perry
1973). Engineering and plumbing handbooks also list
the pressure drops associated with pipe fittings such as
elbows and couplings.

On the other hand, gas velocities within the pipes
should be adequate so that entrained solids will be con-
veyed to their proper point of removal, as shown in
Table 6-1 rather than deposited inside the pipe.

When laying out pipe connections for a gasifier system,
it is important to allow access to various parts that may
require cleaning or adjustment. It is recommended that
new systems be assembled with a large number of pipe
unions to facilitate cleaning out the pipes, as well as
future design modifications. In general, it is better to
use a pipe “T” with a plug rather than an elbow, to
allow for instrument mounting and other additions at

Table 6-1. Gas Velocity
Requirements for Conveying Solids

The conveying velocities in pipes are dependent upon the na-
ture of the contaminant. Recommended minimum gas velocities
are;

Contaminant Velocity
Smoke, fumes, very light dust 10m/s
Dry medium density dust (sawdust, grain) 15 m/s
Heavy dust (metal turnings) 25 m/s

Source: Kaupp 1984a

a later time. After a unit has been thoroughly tested and
is in production, such provisions can be omitted.

6.5 Instruments and Controls

The gasifiers of the past were crude, inconvenient
devices. Today’s gasifiers are evolving toward safer,
automated processes that make use of a wide range of
present-day instruments and controls. An extended
discussion of the system instrumentation and control
requirements is found in Chapter 10.

6.5.1 Temperature

Thermocouples (such as chromel-alumel type K)
should be used to measure various gasifier tempera-
tures, especially below the grate, as a check for normal
or abnormal operation. Temperatures at the grate
should not exceed 800°C; higher temperatures indicate
abnormal function. Consequently, the signal from the
thermocouple can be used by a control system or an
alarm system.

6.5.2 Pressure

Manometers are required to measure pressure drops
across the bed, cyclone, filters, and other components.
(Usually, these pressure drops amount to only a few
centimeters of water pressure.) The manometers are
available as tubes filled with colored liquid or, more
conveniently, bellows manometers (such as a Dwyer
Magnehelic gauge). Both types give a direct reading of
the pressure drop and can be equipped with limit
switches that will sound an alarm to warn when preset
flow levels have been violated, and/or activate control
valves to regulate those flows. Also, electrical
transducers are available that convert pressure dif-
ference into an electrical signal suitable for readout or
control processing.

6.5.3 Gas Mixture

Oxygen sensors have been developed by the automo-
tive industry to measure the small changes in oxygen
concentration required to control the air/fuel ratio for
clean, efficient combustion. They are relatively inex-
pensive and, in principle, can be adapted to gasifier
systems for similar functions (though this has not yet
been done).

6.5.4 Automatic Controls

The fact that an operator will be required for both large-
and small-scale gasifiers is a fixed-cost scale factor that
naturally causes larger systems to be favored.
Automatic unattended operation is therefore essential
to the economic viability of small gasifier systems.
Automatic fuel feed and char-ash removal equipment
are already well-developed for stoker-fed boilers and
could be adapted to automatic gasifier operation.
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Finally, we can look forward to gasifier systems of the
future that will use inexpensive microprocessors to in-
tegrate the signals from these sensors into the

automatic and unattended operation of highly efficient
gasifiers, making gasifiers as simple to use as a car or
home furnace.
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Chapter 7
Gas Testing

7.1 Introduction

It is relatively simple to build a gasifier and operate an
engine for a short time. However, the commercial suc-
cess of gasification ultimately depends on long-term,
reliable operation of gasifier systems. Many gasifier
systems have failed after less than 100 h of operation
because of tar buildup in either the system or the en-
gine. Destroying an engine is a costly method for deter-
mining whether a gas is sufficiently clean for engine
operation. A quantitative knowledge of gas quality and
cleanliness is necessary for the designer, developer,
buyer, and user of gasifier equipment.

This chapter will describe simple and inexpensive
tests of the physical and chemical properties of
producer gas. Using these tests will allow one to
determine whether the gas is suitable for its intended
purpose.

7.2 Gas-Quality Measurements and
Requirements

During gasifier system development, one may need to
be able to measure:

¢ Gas composition: The volume percent of CO, CO,,
H,, H,0, CH,, C, and higher hydrocarbons, and N,
to calculate the gas energy content or to analyze
gasifier operation.

» GGas energy content: Can be calculated from gas com-
position, or it can also be measured calorimetrically
without the need to know composition. The gas must
have an energy content greater than 4 MJ/Nm?3
{100 Btu/scf) for most applications. (See Appendix
for definition of scf and Nm3.)

¢ Quantity of tars: The quantity of condensible or-
ganics in raw gas is a measure of gasifier performance
and determines whether the gas can be cleaned.
Above 5000 mg/Nm? tars, the gas is difficult to clean
up and is suitable only for close coupled direct com-
bustion. Gas cleanup equipment should reduce the
tar level to below 10 mg/Nm3.

¢ Quantity and size of particulates: The nature and
quantity of char-ash and soot entrained in the gas
stream can help to design filters. Particles larger than
10 um must be removed to a level below 10 mg/Nm?
for engine applications.

e Water content of gas: The water content of the gas
helps to calculate cooling requirements.

7.3 Description of Producer Gas and
Its Contaminants

7.3.1 The Gas Analysis

A typical raw gas analysis from a recent SERI test of
corncob gasification is given in Table 7-1. This analysis
includes volume concentrations of each major chemi-
cal constituent, as well as the physical contaminants of
the gas. The energy content of the gas can be calculated
from the energy content of the components using the
high or low heating values (HHV or LHV) for each gas,
as shown in Tables 7-1 and 7-2. The LHV also can be
determined graphically from Fig. 7-1.

7.3.2 Particulates

Some particulate levels reported for wood and charcoal
gasifiers are listed in Figs. 7-2 and 7-3.

In order to remove particles with the appropriate equip-
ment, it is necessary to know their nature and size dis-
tribution. Particle size distributions shown in Fig. 7-4,
Table 7-3, and Fig. 7-5 were obtained by mechanical
screen separation of the cyclone contents for SERI and
Imbert tests. The results of both tests are plotted in
Fig. 7-4 on log probability paper for ease of analysis,
displaying the distinctive slope common to most fine
powders produced by fragmentation.

The potential particle-size range of a wide variety of
particles and their characteristics are shown in Fig. 7-6,

Table 7-1. Composition of Producer Gas from
Corn Cobs after Cyclone Separation

Physical Composition

Tar content 1300 mg/m3 1300 ppm@
Particulate 330 mg/m3 330 ppm
Ash content of particulate 30 mg/m3 30 ppm
H>0O 7Awt% 71,000 ppm

Chemical Composition

CcO 19 Vol % x 322 Btu/scf = 61

CO2 14Vol%x O

Ho 17 Vol % x 325 Btu/scf = 55

CHg4 2 Vol % x 1031 Btu/scf = 20

No 48Vol % x 0 o
136

Dry Gas (HHV)P 136 Btu/scf (60°F, 30 in. Hg Dry)

a1 Nm3 of gas weighs about 1 kg, so that 1 mg/Nm3 = 1 ppm

bThe gas heating value may be calculated from the gas analysis using
Table 7-2.
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along with the equipment appropriate for separation of
each size range.

Itis important to distinguish between the various forms
of particulates that result from biomass combustion
and gasification. Starting with full-sized biomass fuel
and 0.5% ash, we can use this ash as a tracer to follow
conversion in the gasification process.

Although combustion and updraft gasifiers leave a
white ash, downdraft gasifiers do not produce this
white mineral ash because there is no oxygen present
when the final charcoal breakup occurs. Freshly
produced charcoal, just after it has finished flaming
pyrolysis, is only slightly smaller than it started out and
should not be able to pass through the grate. As
gasification of the charcoal proceeds, carbon is
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Fig. 7-1. Nomogram for lower heating value of producer gas (Source: Kaupp 1984a)
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gas generators (Source: Gengas 1950, Fig. 86)

52 Handbook of Biomass Downdraft Gasifier Engine Systems



removed from both the surface and interior of the char
particle by motion of the hot gases. The char particle
cracks and crumbles as carbon is converted to ash.

The term char-ash refers to the black dust that falls
naturally through the grate in a downdraft gasifier
when gasification is as complete as it will go. Char-ash
is produced during the final breakdown of the charcoal
mechanical structure as the charcoal reacts with
pyrolysis gases. Char-ash from downdraft gasifiers still
contains 50% to 80% carbon {Fig. 3-3), which is
enough carbon to give char-ash a black color. Char-ash
usually is collected below the grate or in the cyclone
separators.
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Fig. 7-4. Imbert and SERI downdraft particle size distribution of
producer gas char dusts

If the fine char-ash from downdraft gasifiers is about
20% ash, then this represents a 95% mass conversion
of the fuel since we started with 0.5% ash biomass. We
can also see from Fig. 3-3 that it is desirable to keep the
char larger than 500 pm (0.5mm) in the gasifier to boost
efficiency. Also, particles under 500 pm have
completed their task and should be removed as
thoroughly as possible.
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Fig. 7-5. Residue curve for the screening of Imbert generator gas
(Source: Gengas 1950, Fig. 88}

Although the unconverted carbon found in char-ash
represents an energy loss, it also has several benefits.
The final ash from combustion is less than 1 pm in size
and can be captured only in expensive bag house
filters. The char-ash holds the ash in a 10 pm matrix
which is captured by cyclones. The char-ash may have
considerable value as a charcoal or as a soil
conditioner.

Char-ash particles smaller than the cyclone separator’s
cut-point pass through the cyclone separator. Smaller
particles normally are higher in ash content, as shown
by Fig. 3-3. Higher ash content is more abrasive;
however, solids smaller than the oil film thickness do
not cause major engine wear. Ash that has been sub-
jected to slagging is much harder and more abrasive
than nonslagging ash, which crumbles easily.

All biomass contains some ash (typically a few per-
cent), but some fuels, such as rice hulls or MSW, can

Table 7-2. High Heating Value and Low
Heating Value of Gas Components?

Component Symbol HHVP (MJ/Nm3) HHVE (Btu/scf) LHVP (MJ/Nm3) LHVC (Btu/scf)
Hydrogen Hp 13.2 325 1.2 275
Carbon monoxide CcO 13.1 322 13.1 322
Methane CH4 41.2 1013 37.1 913

a1 Btu/scf = 8.26 kcal/Nm3 = 40.672 kJ/Nm3
bStandard conditions, 0°C and 760 mm Hg Dry
CStandard conditions, 60°F and 30 in. Hg Dry
Source: Adapted from Perry 1973, Table 9-18.
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Table 7-3. Analysis of Wood Gas Dust

% Ash
Over 1000 um (1 mm screen) 1.7
1000 - 250 um 24.7
250 - 102 um 23.7
102 - 75 um 7.1
75 - 60 um 8.3
Under 60 um 30.3
Losses 4.2

100.0
Water content 3.2
Ash content, dry sample 10.6
Loss due to burning, dry sample 15.7
Content of FeoO3 in the ashes 11.0
Content of SiO» in the ashes 7.7

Source: Gengas 1950, Table 2-9

contain 20% ash or more. During flaming pyrolysis of
the original biomass, the organic molecules break
down to form a very finely divided soot (carbon black),
such as that seen in oil or candle flames. Soot particles
are much smaller than char-ash particles (ordinarily
less than 1 pm). The soot is so fine that it can be ex-
pected to pass harmlessly with the gas and burn in the
engine without harm.

Carbon monoxide is unstable below 700°C and, given
enough time, will decompose in the presence of certain
catalytic metal surfaces to form carbon and carbon
dioxide, according to the reaction

2C0—-C+CO,. (7-1)

This carbon, known as Boudouard carbon, is slippery
to the touch and nonabrasive. Below about 500°C, the
reaction is very slow. Normally, Boudouard carbon
does not form in gasifiers because the gas cools quickly
through this temperature range.

Char-ash, because of its high mineral content and
abrasive potential, is the main cause of engine wear in
engine systems and understandably is a main focus in
gas cleanup. Similarly, soot and Boudouard carbon are
inherently ash free, nonabrasive, and possibly lubricat-
ing. Despite their small size and difficulty of capture,
they are not seen as a significant factor in engine wear.

Typically, the largest particles that pass through the
grate can be extracted mechanically, for instance, with
an auger. Subsequent to that step, removing the
suspended small particles is the principal problem in
gas cleanup. The particulates in Table 7-1 were
collected after the cyclone separator, which caught
50% of 50-um particles. These smaller particles are
composed of very fine char-ash, soot, and tar mists.

The type of gas cleanup equipment required is deter-
mined by the particle sizes that must be removed, and
therefore it is important to determine the particle-size

distribution as well as the total quantity of particles.
Relatively little information is available in the litera-
ture (Gengas 1950), so complete gas cleanup design
necessitates measurement or knowledge of particle-
size distribution. Particle-size measurement is
discussed further in Section 7.8.

7.3.3 Tars

Initial biomass pyrolysis can produce up to 60% “wood
oil,” composed of the monomers, oligomers, and frag-
ments of the biomass polymers cellulose, hemicel-
lulose, and lignin. Subsequent high-temperature
cracking (over 700°C) of these large molecules results
mostly in gas, but also polymerization to form 5% to
10% of heavier polynuclear aromatic molecules that
are similar to coal tars. Up to 20% of these tars and oils
can be carried through with the gas from updraft
gasifiers.

In downdraft gasification, oxygen is available to burn
these oils during pyrolysis. Although flaming pyrolysis
burns most of the tars and oils, 0.1% to 1% (depending
on the gasifier design) can be expected to survive.
These tars and oils are troublesome in the
gas-processing system and the engine, so they must be
thoroughly removed by scrubbing.

Tars occur mostly as a mist or fog composed of fine
droplets that may be less than 1 pm in diameter (see
Fig. 7-8). Tar mists continually agglomerate into larger
droplets and tend to saturate and coat solid particles.
If not removed, tar mist forms deposits that cause en-
gine intake valves and other moving parts to stick.
Before a gasifier is considered suitable for operating an
engine, it is imperative that one test the producer gas
for tars and particulates,

7.4 Gas Sampling
7.4.1 Sample Ports

A temporary or permanent port must be provided at
each point on the gasifier where samples are desired,
as shown in Fig. 7-7, such as downstream from the
cyclone and before the burner or engine, as well as at
each stage along the gas cleanup train when it is desired
to determine the effectiveness of each system
component. It is important that the gas sample is
representative of the gas at each point. The port and tap
may need to be heat traced to prevent premature
condensation (see below).

Permanent sampling ports should be closed off with
gate or ball valves, which provide a straight through
passage. Needle valves and sill cock type water valves
should be avoided, since much of the material being
sampled will deposit within the twisted passages of
these types of valves.
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Fig. 7-7. Static tap sampling port (Source: ASME 1969, Fig. 1)

The temperature, pressure, and moisture content of gas
at the nozzle must be accounted for when designing a
sampling train, and measurements should be reduced
to standard conditions. For average samples, the test
duration should be long enough to average the reading
over at least one cycle of the equipment being tested—
for instance, the fuel-feed cycle, the scrubber service
cycle, or the shaking cycle. For snap samples or where
a transient phenomenon is being observed, then the
most rapid sampling method and small samples should
be used.

Hot, raw gas emerging from any gasifier will contain
tar, char-ash, soot, and water vapor, and it is relatively
simple to measure these quantities in a small sample
of raw gas. After the gas has been cleaned and condi-
tioned, the measurements become more difficult. Im-
purity levels are much lower, so it is necessary to
handle much larger gas samples in order to accumulate
a measurable-sized sample. The principles of measure-
ment remain the same, but the measurements require
more time to accomplish.

The physical analysis of producer gas is based on the
weight of tar, particulates, and water in a measured
quantity of gas. Therefore, a positive-displacement gas-
testing meter (such as those made by Singer, Rockwell,
and the American Gas Association) should be available
for calibrating flowmeters, pumps, and similar com-
ponents. An analytical balance capable of weighing to
0.1 mg is also required for accurate measurement of tars
and particulates.

Approximate results can be obtained by comparing the
volume of sample required for a particular depth of
color deposit such as 50% grey measured on a standard
grey scale such as that used for smoke testing (Dwyer
1960). Tests at SERI found that a color of 50% grey on
a 47-mm filter disc, taken from a standard grey scale,
represented approximately 0.12 to 0.5 mg of collected
contaminants (Das 1985). A quantity of gas for chemi-
cal analysis can be collected at the same time as the
sample to be used for physical analysis.

7.4.2 Isokinetic Sampling

Isokinetic (equal gas-velocity) conditions in the flow
chamber and the sampling tube should be ensured
where particle sizes exceed 10 um. Otherwise particle-
size distribution will not be the same in the chamber
and sampling tube. The design of sampling-tube parts
and their placement within the gas stream are shown
in Figs. 7-8 and 7-9, respectively. Figure 7-10 shows
velocity streamlines for a sampling tube in a flow
chamber. Part “a” illustrates isokinetic conditions; that
is, the streamlines are equally spaced within the duct
and tube. In “b,” sampling-tube velocity is less than in
the duct (indicated by the wider streamline spacing in
the tube), and proportionately more gas must flow
around the tube than through it. However, the inertia
of large particles impedes their being carried by the gas
that deflects around the tube. Large particles in line
with and immediately upstream of the sampling tube
continue their flight into the tube. Hence, propor-
tionately more large particles exist in the tube than flow
in the chamber. The opposite holds true where tube
velocity is greater than the velocity in the chamber
(Fig. 7-10(c)); i.e., large particles are underrepresented
in the tube.

Nonisokinetic sampling erroris plotted in Fig. 7-11. We
can see that for under 10 pm-particles, the concentra-
tion error is within £10% over a wide range of sampling
velocity from half to double the gas velocity for a
velocity ratio us/u, between 1/2 and 2.

In practice, the high efficiency cyclone separator of a
gasifier system will remove most particles larger than
10 pm, so for the smaller particles remaining in the gas
stream, the error due to nonisokinetic sampling can be
ignored.

The sampling error for nonisokinetic sampling condi-
tions also can be neglected for tar mists and other very
fine aerosols. In fact, the sampling port of Fig. 7-7 can
serve as a convenient 10-um coarse inertial prefilter.
The particle-sample probe of Fig. 7-9 tends to accumu-
late impacted large-particle deposits and eventually
clogs, so it should be adjusted to face downstream
during periods that it is not in use.
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The isokinetic flow rate can be calculated as where

Q=0Q (DHZ/DPZ] (7-2) Q = flow rate,
D = diameter.
5 The subscripts n and p refer to nozzle and pipe,
g respectively.
g The flow rate of a dirty-gas stream can be measured ir-

respective of temperature and molecular weight of the
gas by using a balanced-tube, null-type apparatus.
Such a system in effect measures chamber flow by col-
lecting a portion of the gas flow through a sampling
tube, cleaning it, and measuring it. The mass flow
through the entire chamber is then calculated, using
the ratio of chamber area to sampling tube area.
Measurements are made once the velocity in the sam-
pling tube has been adjusted (via a vacuum pump in
the system} to be equal to that in the flow chamber.
(a) Velocities within the tube and chamber are equilibrated
by using a differential manometer to balance the static
pressures for the tube and chamber. The best placement

T~ for the probe within the chamber can be checked by
! ! testing the flow profile across the chamber. The probe
! ! should be located where the flow is average for the
2 D min. ': ! chamber. Where necessary, flow straighteners should
straight | ¢ - be used to ensure accurate readings. Balance-tube,
length 11 ' / null-type sampling, without gas-cleanup equipment,
‘ ': E ‘ / can be used for clean gas.
! A
L S 7.5 Physical Gas-Composition Testing
D 0.25 mm
] ] 300 max. radius 7.5.1 Raw Gas
Sampling train options for measuring the range of
B A levels of tar, char-ash, and water are shown in Fig. 7-12,
90° 0 , « | and gasifier test-train component options are presented
ote 3 ! X ; in Tables 7-4, 7-5, and 7-6.
[} . .
' 2 Certain basic procedures must be followed whenever
‘ f sampling producer gas—be it for tar, particulates, heat
] \ WR vy ! content, etc.:
i _____{ I'Nole 2
e e Jlo o .
End view Side view E
(b) o)
Not p
otes: . <
1. The area upstream of this 90° sector shall be Sggig:;ng Pt?bee Oust SS'J.E’E'\'” Fowmeter /VOIVe ¢
free of obstructions between planes A-A and
B-B. \E \ P_’__“ -
T
2. No portion of probe shall project upstream of — such(c))n fon
nozzle entrance within a distance from nozzle
centerline of 15 cm or 5 nozzle diameters,
whichever is greater. T T T
D = Nozzle diameter Y = 4 x D (minimum) Uncleaned
R =2 x D (minimum) X =6 x D {minimum) qases
Fig.'7-8. (a) Typical holder for flat round filters and (b) recommended Fig. 7-9. Large particle sampling probe (Source: Strauss 1975,
design for sampling nozzle tip (Source: ASME 1980) Fig. 2.9. © 1975. Used with permission of Pergamon Press)
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¢ The gas, being noxious, should be either burned off Figure 7-12(a) shows a setup for measuring particulates

or returned to the pipe downstream from the sam- and tar, or moisture, or gas composition, or gas
pling point. production rate.
» The gas should be cleaned. Figure 7-12(b) is a setup for continuously measuring
¢ The gas should be dried. change in water content and change in heat content. It
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Fig. 7-10. Gas stream lines at the entrance to sampling probes (Source: Adapted from Strauss 1975, Fig. 2.12. © 1975. Used with permission of
Pergamon Press)

Table 7-4. Components for Raw Gas Contaminant Test Train

Item Requirements Price Rank (1=Lowest)
Sample Probe Tar-only measurements permit any suitable fitting with pipe thread access 1
Tar and particulates require tubing with 90° bend facing the gas stream 2
Shut Off Valve Ball Valve 1/4 in. —
Filter Holder 47 mm reusable hoider
Polycarbonate 1
Aluminum 2
Stainless 3
Filter Discs Glass fibers 99.9% efficiency at 0.3 um —
Gas Sample Pump Hand operated rubber bulb 1
Plastic piston pumps (36 cm/stroke) 3
Water-powered aspirator pump 2
Motor driven vacuum pump 4
Gas Sample Flow Flow meter, 2 in. scale, 4% full scale 2
Gas flow indicator 1

—_

Gas Test Meter Positive displacement meter to indicate accumulated sample volume
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also can be used to measure gas flow and to collect
moisture and tars and particulates over a measured
time.

Figure 7-12(c) shows a system train for measuring gas
flow and tars and particulates.

The ball valve at the sample port (Fig. 7-12(a) and (c))
permits changing the filter disc without danger of
releasing gas or admitting air. The filter holder in
Fig. 7-12(a) and (¢) must be maintained, by electric
heating or locating close to the hot gas pipe to keep the
filter hot enough, above the water dew point, typically
80°C, to avoid water condensation in the filter disc.

The desiccant-drying section (Fig. 7-12(a)-(c)) should
be constructed so that it can be disconnected for weigh-
ing. We have found that an indicating desiccant as-
sembly can be fabricated by containing the desiccant
between two glass-wool plugs in Tygon or glass “U”-
tubes (Fig. 7-12(b)). (Drierite is a commercial form of
anhydrous CaSO, containing cobalt sulfate, which
changes from blue to pink when it becomes hydrated.)
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Fig. 7-11. Errors in concentrations of 5 and 10 um particles (Source:
Strauss 1975, Fig. 2.13. ® 1975. Used with permission of Pergamon
Press)

Table 7-5. Additional Components for Continuous Gas Quality Test Train

Function Specifications Manufacturer
Sample Port 1/4 in. FPT Fitting Common compression fittings
1/4 in. MPT —1/4 in. tube compression (e.g., Swagelock)
Hygrometer Cross fitting bushed down to accept 1/4 in. tubing User fabricated
Dry Bulb Water reservoir in bottom
Wet Bulb 2 hole stopper with thermometers
Dry bulb bare
Wet bulb wrapped with wicking that dips into water reservoir
Dryer Indicating desiccant Drierite or silica gel in a container large User fabricated
enough for several hours sample time 1.5 cm dia x 40 cm
U-tube
Pump Aquarium pump modified for suction and pressure service Whisper (User modified)
capable of 50 in. WG, 0.01 scfm
Flow Meter Floating ball rotameter—RMA Dwyer
Burner Diffusion flame non-mixed See Fig. 7-16 or user fabricated
1/4 in. tubing with 1/8 in. ID opening
Readout Chromel-alumel thermocouple Omega, etc.
Table 7-6. Additional Components for Condensible Collection Test Train
ltem Purpose Price Rank
Gas Dryer For small sample moisture determination 1
8 in. length of 3/8 in. tubing with fittings filled with indicating desiccant weighed
before and after each test 2
For large sample pump protection
250 ml flask filled with indicating desiccant or ice bath, bubbler, impinger, condenser 3
Filter Heater To prevent condensation at probe 1

Heater tape around sample lines and filter holder heated chamber
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The desiccant container should be sealed for transport
and weighing. Raw-gas moisture measurement is
essential to mass balance calculations.

Finally, where volume is measured (Fig. 7-12(a) and
(¢)), means must be provided to pull a known quantity
of gas through this train. Hand-held positive displace-
ment vacuum pumps are made by a number of sup-
pliers (e.g., Mine Safety, Draeger, and Gelman). We
have also used a hand-held rubber aspirator bulb and
found that 70 strokes collected 3 L of gas (0.1 ft3). We
also have used a Dwyer smoke test pump. The gas meter
is required only for initially calibrating the sampling
train and pump, since counting strokes yields adequate
precision for measuring the test-gas volume. A decision
on the amount of gas to be sampled should be based on
the anticipated impurities in the gas and the con-
taminant sample quantity required for the specific
analysis methods available. For instance, 50% grey
scale analysis requires a 0.5 mg sample on a 47 mm
filter disc.

Weighed samples require a 5 to 30 mg sample size for
analytical balances with 0.1 mg readability.

7.5.2 Cleaned Gas

If the gas is cleaned sufficiently for engine use, it will
be necessary to pass a much larger sample (usually
1 m3) through the filter. A mechanical pump capable
of pulling a moderate vacuum, such as a motor-driven
vacuum pump or a calibrated air-sample pump, is
recommended. The positive-displacement meter can
also be located in the collection train between the
pump and the gas return if the system pressure is close
to atmospheric pressure. It is imperative to protect the
pump and meter with a large absolute filter because any
tar or particulates entering the pump or meter will
rapidly affect their performance.

7.6 Chemical Gas Composition

7.6.1 Gas Samples for Chemical Analysis

The gas composition can be measured either con-
tinuously (on-line) or through discrete samples taken
periodically from the gas stream. These methods will
be discussed separately. Before the gas is analyzed, it
must be drawn from the system and cleansed of tar and
particulate contaminants, as described previously.

Batch-sampling requires collecting a sample of gas in
a suitable container {e.g., glass cylinder, metal cylinder,
Tedlar gas sample bag or syringe), as shown in Fig. 7-13.
The subsequent analysis is only as good as the sample,
and it is easy for gas leaks to spoil a sample after it has
been taken. Therefore, it is important to use extra care
to avoid leaks either into the sampling train while the
sample is being taken or out of the sample bulb before
the analysis is made.
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Fig. 7-13. Gas sample containers (Source: Strauss 1975, p. 13.
© 1975. Used with permission of Pergamon Press)

When possible, the sample cylinder should be
evacuated or, alternatively, should be very thoroughly
flushed. The cylinder should be filled to at least a small
positive pressure from the pump (Fig. 7-14), so that air
cannot leak in before analysis. A positive pressure
sample can be collected without a pump by chilling the
cylinder before the gas is taken, so that a positive pres-
sure develops as the gas in the cylinder warms to room
temperature. Gas samples should be drawn from a
point as close as practical to the gasifier outlet, in order
to avoid errors due to air leaks in the gasifier piping.

Usually, any oxygen found in the gas can be attributed
to air leaks, since oxygen is completely removed in the
gasifier. When oxygen is found in the gas, the composi-
tion can be converted to an “air-free” basis by subtract-
ing the oxygen and the corresponding ratio of nitrogen
(the N,/O, ratio in air is 79/21).

Some gas sample containers are shown in Fig. 7-13. A
rubber septum is a desirable feature that permits one to
extract the gas sample with a hypodermic syringe for
injection into a gas-chromatograph without opening
the stopcocks. The hypodermic syringe for injecting
samples into the gas chromatograph should have a
valve at the needle that can be closed between filling
the syringe and analysis. Valved syringes are available
as accessories from gas chromatograph manufacturers.
The metal cylinder of Fig. 7-13 can contain gas at a
much higher pressure than the glass system. It is im-
portant to use leak-proof valves rather than needle
valves on this container and to avoid stopcock grease,
which has a high hydrogen solubility. A syringe also
can be used to collect a gas sample. If standard gas
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sample containers are unavailable, gas samples can be
collected in glass bottles by water displacement, insert-
ing a stopper while the bottle is submerged and sealing
by dipping the stoppered opening in paraffin.
Whichever container is used, the samples should be
tested as soon as possible, since hydrogen can rapidly
diffuse through rubber seals and stopcocks, thereby
changing the gas composition in a few hours.

7.6.2 Methods of Analysis

7.6.2.1 Gas Chromatography

Gas chromatography (GC) is the most widely used
method of gas analysis. It depends on the ability of cer-
tain adsorbent materials to selectively slow the rate of
gas passage through a column packed with the adsor-
bent. Hydrogen is slowed least, CO, N,, and O, are

(b)

BA-GO201755

Fig. 7-14. Apparatus for drawing gas samples: (a) Filling sample containers by liquid displacement; (b) hand-operated piston vacuum pump; (¢) motor-
driven rotary vacuum pump; (d) rubber bulb hand aspirator; (e) Chapman filter pump (Source: (a, d, e) ASME 1969, Figs. 6 and 7)
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slowed to a greater extent, and water and CO, are
slowed to the greatest degree. The gas sample is mixed
with a carrier gas; usually, helium is used because it
does not occur naturally in the sample. A detector,
which is inserted into the gas stream at the end of the
column, records on a chart recorder both the time of
passage and the quantity of each component. The
presence of a particular gas component is indicated by

a peak. The quantity of that gas is then determined by
integrating the area under the peak in the curve and
compared with that in a calibration gas of known com-
position. More advanced recorders include automati-
cally controlled valving, integration of the response
curves, calculation of gas quantity from calibration fac-
tors, and a printout of the composition results. Such a
printout is shown in Fig. 7-15.
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[HP] 27: MANUAL INJECTION AT 15:49 FEB 7, 1984
RUN E7 SAMPLE W6
RUN TIME AMOUNT NAME
2.87 25.723 HYDROGEN
4.25 0.15 PROPYLENE
9.46 0.08219 TRANSBUTENE
15.26 14.592 CO2
15.84 1.318 ETHYLENE
16.37 0.136 ETHANE
17.09 0.367 ACETYLENE
18.41 0.49 OXYGEN
18.76 1.949 NITROGEN
19.58 3.9303 METHANE
23.92 0.002827 BACKFLUSH

Sum

Fig. 7-15. Typical gas chromatography printout
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The most common GC detector performs analyses by
measuring the thermal conductivity of the gas (TC
detector) and is the most suitable for producer gas
measurement. With this type of detector, helium (or a
hydrogen-helium mixture, see below) is often used
because of its abnormally high thermal conductivity
relative to other gases.

The flame-ionization detector (FID) measures the num-
ber of ions produced in a flame and is particularly use-
tul for detecting hydrocarbon species. The FID is not
particularly useful for producer gas, since producer gas
contains few hydrocarbons other than methane.

The response of the TC detector to low levels of
hydrogen in the inert carrier gas is nonlinear, and this
leads to ambiguous results. There are two effective
solutions to this problem. A heated palladium tube at
the inlet can be used to selectively diffuse the hydrogen
out of the sample into a separate nitrogen gas stream;
in this secondary stream, hydrogen yields a linear
response (Carle method). Alternatively, adding
hydrogen to the helium carrier gas will move the
baseline onto the linear region of the TC-detector
response curve.

The position of a peak on the time scale of the recorder
chart indicates the time of retention and is characteris-
tic of each particular gas component. The area under
the peak, obtained by analog or digital integration, in-
dicates the volume of each gas present. Although reten-
tion times and sensitivities are listed for each adsorbent
material, aging and drift are common to column pack-
ings, so it is necessary to calibrate the instrument daily
to obtain an accuracy on the order of 1%. For this pur-
pose, it is necessary to have a cylinder of previously
analyzed standard gas. These cylinders are available
from GC equipment manufacturers,

Although samples are usually collected as needed, it is
possible to use automatic sampling with the GC to give
a measurement of gas composition at regular intervals.
The GC analysis cycle time depends on both the reten-
tion time of the columns used and the number of
species analyzed. This time is typically 30 minutes, but
note that the warmup time for the GC is one day.

A number of companies, including Carle, Hewlett-
Packard, and Perkin-Elmer, manufacture satisfactory
units for $3000 to $30,000 and provide excellent
instruction and service.

7.6.2.2 Orsat Gas Analysis

The Orsat gas analysis system was developed to
measure the gases CO,, CO, O,, H,, and CH,. It was the
principal measurement method used before the GC was
developed in the 1950s and is more reliable and less
costly than GC; however, it requires more time (typi-
cally 30 minutes of full operator attention per analysis)
and more skill.

The Orsat analysis depends on the ability of certain
chemicals to react selectively with each gas component
of the producer gas mixture. The components are ab-
sorbed in the order of CO,, O,, CO, then H, and CH,,
and the analysis reports the volume percent of each
component directly.

Orsat analysis equipment is portable, does not require
AC power, has no warmup time, and can be purchased
(along with the required chemicals) from scientific
supply houses for $500 to $1000.

7.6.2.3 On-Line Gas Measurement

It is convenient to have continuous “on-line” measure-
ment of all the gas components to show instantaneous
changes in composition that otherwise would not be
shown by batch sampling. Methods for on-line gas
analysis include flame observation, combustion
calorimetry, infrared absorption, thermal conductivity,
and mass spectrometry.

The heat content of the gas is a measure of a gasifier’s
performance and can be calculated from the gas com-
position (see Fig. 7-1 and Table 7-2). Most gasifier
facilities, if they have gas analysis equipment, use an
Orsat analyzer or a gas chromatograph, so that normal-
ly a value is available only after a considerable time
delay (10-30 min). It is desirable to have a continuous
indication of gas quality.

Continuous immediate readout of producer gas com-
position, however, has been achieved in two ways. One
method, used at U. C. Davis, uses infrared (IR) absorp-
tion for continuous CO, CO,, and CH, analysis with a
thermal conductivity detector for continuous H, deter-
mination. The second method uses a mass spectrom-
eter to give immediate on-line digital readout of all
gases present, CO, CO,, H,, O,, H,0, CH,, and high
hydrocarbons (Graboski 1986).

The calorimeter shown in Fig. 7-16 is a precise primary
standard for measuring HHV of the gas. Combustion
air, fuel rates, delivery temperatures, and pressures are
carefully measured. Heat-transfer air is also metered for
inlet flow, temperature, and pressure. A counterflow
heat exchanger cools the combustion products to the
air inlet temperature (60°F) and simultaneously con-
denses water vapor to a liquid. The temperature rise of
the heat-transfer air is directly proportional to the HHV
of the fuel gas. The equipment pictured in Fig. 7-16 was
designed for gas with a HHV of 1000 Btu/scf and may
require modification of the burner to use producer gas
with a HHV of 150 Btu/scf.

Other simpler, more relative methods are available and
may be sufficient for many applications. It is informa-
tive simply to observe the gas flame during operation.

Flame length tends to increase with the gas heating
value; flame luminance increases with hydrocarbon
and tar content. After the operator has gained
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Fig. 7-16. Gas calorimeter combustion chamber (Source: Adapted
from ASTM 1977)

experience, the flame can reveal a good deal on the
functioning or malfunctioning system.

One author (Das) has used a small burner/thermo-
couple monitor shown in Fig. 7-12(b), which produces
a temperature signal roughly proportional to the heat
content of the clean gas.

The accuracy of continuous sampling equipment is
subject to accumulation of gas contaminants, so
prefiltration should be used as in Fig. 7-12(b) of

adequate capacity for prolonged use and adequate
efficiency for equipment protection.

7.6.3 Water Vapor Analysis

Water vapor can be determined by many methods. The
three most suited to producer gas are psychrometry,
condenser outlet temperature, and gravimetric
methods.

7.6.3.1 Psychrometry

Water content can be determined by measuring the wet-
and dry-bulb temperature of the gas as in Fig. 7-17. The
moisture content is then calculated from a
psychrometric chart (Fig. 7-18 or 7-19, depending on
gas temperature) to find the moisture as absolute
humidity.

Moisture wt % = Absolute humidity x 100 (7-3)
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Fig. 7-17. Gas stream hygrometer (Source: RAC, p. 7)
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Adapted from ASHRAE 1981)
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Fig. 7-19. Psychrometric chart for high temperatures (Source: ASME 1980, Fig. K-3)

7.6.3.2 Gravimetric

Gas moisture also can be determined by passing a
measured volume of gas through a preweighed dryer
assembly containing desiccant. Moisture can be
calculated as

100 x Weight gain

- 0, p—
Moisture wt % = Sample vol. x Gas density

(7-4)

7.7 Analysis of Test Data

7.7.1 Mass Balances and Energy Balances

A typical gasifier mass balance, shown in Table 7-7, is
an accounting of all mass inputs to the gasifier (or
gasifier system) and all mass ouputs over a given time.
Since the law of conservation of mass requires that
mass be conserved in any process, the total mass input
must equal the total mass output. Any deviation from
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an exact balance indicates either an error in measure-
ment, or that some important flow has been
overlooked.

The total inlet and outlet mass flows must not only
balance each other, but also the inlet and outlet mass
flows of each element (in this case, carbon, hydrogen,
and oxygen) must balance. This elemental mass
balance is a more rigorous test of measurement
procedure; the sources of error in the global (overall
inlet and outlet) mass balance may be pinpointed by
the elemental mass balance.

Table 7-8 shows an energy balance, which is obtained
by tabulating the energy associated with all input and
output streams. The law of conservation of energy,
which requires that energy be conserved, therefore
provides a means for evaluating efficiency, finding in-
strumental errors, or calculating quantities that cannot
be measured directly.
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Table 7-7. Mass Balance

Inputs (kg/h) Outputs (kg/h)

Run Wet Chips Dry Air H,0 Total Dry Gas Char Tar H,0 Total Closure (%)
98 32.0 43.1 0.5 75.6 66.3 0.9 0.09 7.4 74.7 98.9
910 32.0 452 0.5 77.7 68.0 0.9 0.14 7.4 76.4 98.3
920 35.7 62.1 0.4 98.2 92.9 1.4 0.09 71 101.5 96.6
101 52.4 76.9 0.8 130.1 116.9 1.7 0.09 13.3 132.0 98.6
929 52.7 74.0 1.0 127.7 113.0 1.8 0.14 12.2 127.1 99.5
106 58.1 76.8 0.5 135.4 117.0 1.2 0.09 10.8 129.1 95.4
1119 89.1 1121 1.0 202.2 173.8 3.0 0.27 19.7 196.8 97.3
1117 96.2 140.4 0.7 237.3 218.9 25 0.18 229 2445 97.0
1221 140.5 202.1 1.1 343.7 302.1 41 0.54 43.3 350.0 98.2

Source: Walawender 1985, p. 918.

Mass and energy balances have only been applied oc-
casionally to gasifier development because of the dif-
ficulty and expense of measuring all flow streams.
Detailed mass and energy balances usually can be per-
formed at universities in chemical engineering
laboratories or at major research laboratories, and only
a few have been performed on air gasifiers. If gasifica-
tion is to become a developed field, it is necessary to
perform mass and energy balances.

7.7.2 Flow Rate Characterization

The variation of gas quality of a gasifier with flow rate
helps determine optimum sizing parameters. Note that
in Fig. 7-20 total hydrocarbons and tars steadily
decrease with increased flow. In Fig. 7-21 we see that
maximum heating value, peak CO and H,, and maxi-
mum efficiency do not coincide. Peak efficiency occurs
at more than twice the flow rate for maximum heating
value.

Actual hearth load (see Section 5.7.3) for sizing
depends on the application. Heating applications that
need maximum flame temperature should be sized
with lower hearth rate load to provide maximum heat-
ing value. Heating applications that need maximum
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Producer gas flow rate, m3/h (25°C, 1 atm)

Fig. 7-20. Tar versus flow for rice hull gasifier (Source: Kaupp 1984b,
Fig. 10-88)

fuel economy should be sized with higher hearth load
to coincide with the peak efficiency curve. Engine ap-
plications should be sized for maximum hearth load to
coincide with peak of the efficiency curve in order to
allow maximum room for turndown.

7.8 Particle-Size Measurement

Knowledge of the size distribution and other charac-
teristics of gas contaminants is helpful for cleanup
design. Table 7-9 presents particle-size analysis
methods and examples of equipment available for
characterizing particle size.

7.8.1 Typical Particle-Size Distributions

The particle-size distribution of solid char and ash for
raw gas shown in Fig. 7-12 was produced by mechani-
cal screen separation for both the Imbert (tuyere and
hearth constriction) and the SERI unconstricted gas
producers. The difference between the two gasifiers is
caused by the grate design. Gas from the Imbert gasifier
exits upward through a settling space that retains larger
particles. On the other hand, the SERI oxygen gasifier
passes all solids through the gas outlet. Note that the
overall slope is the same for both size distributions.

7.8.2 Sieve Analysis

The distribution of large solid particles greater than
40 um may be determined using sieves. Table 7-10 lists
actual sieve size for various mesh size.

7.8.3 Microscopic Size Analysis

Particles captured on a filter disc can be counted by
microscopic examination. However, particles smaller
than 10 um are difficult to see under a light microscope,
and liquid-droplet sizes cannot be determined by this
method because droplets, once captured, coalesce,
leaving no evidence of their original size.

7.8.4 Aerodynamic Size Analysis

Aerodynamic sizing can be accomplished with either
a cascade impactor or a cascade cyclone. Particles and
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Table 7-8. Energy Balance

Dry Chip Matl. Dry Gas Char Tar Tar Dry HO Energy Total Mass con- Cold gas
chip moisture balance gas yield yield yield in air in out yield energy version effici-
rate (%> wet  closure HHV (m¥kg  (Kg/100 (kg/1000 dry gas (kg/ (kg/ (MJ/kg out effici- encyP
Run (kg/h) basis) (%) (MJ/m3) DC?) kg DC) kg DC) (ppm) kg DC) kg DC) DC) (MJ/h) ency (%) (%)
98 26.7 16.52 98.9 5.33 2.22 3.28 2.55 1370 1.61 0.277 11.82 315 87.7 70.2
910 26.7 16.52 98.3 5.18 2.28 3.21 4.59 2003 1.69 0.277 11.82 315 87.5 70.2
920 33.6 6.01 96.6 5.33 2.48 4.06 2.7 977 1.85 0.212 13.14 441 94.7 781
927 41.2 8.11 85.4 5.63 2.54 3.08 3.30 1180 1.75 0.438 14.31 590 98.1 85.0
922 47.8 5.75 85.1 5.44 1.43 2.76 1.90 1199 1.05 0.186 7.76 370 74.9 461
101 48.7 6.99 98.6 5.59 2.18 3.45 1.86 777 1.56 0.273 12.19 594 89.8 725
929 48.8 7.41 99.5 5.63 2.10 3.72 2.79 1205 1.52 0.250 11.82 576 88.5 70.2
1011 49.5 7.31 91.6 5.44 2.55 3.94 4.59 1607 1.88 0.321 13.84 685 96.1 82.2
1015 52.3 7.95 88.8 5.33 273 4.69 5.21 1679 2.08 0.381 14.52 759 97.5 86.3
924 53.7 5.50 91.8 5.66 1.62 2.79 2.54 1426 1.13 0.209 9.11 490 80.9 54.2
106 543 6.67 954 5.48 1.85 2.18 1.67 776 1.42 0.199 10.68 579 86.5 63.5
123 54.3 14.61 94.2 5.51 1.90 4.27 3.35 1594 1.35 0.236 10.44 567 83.0 62.1
104 63.7 7.72 92.6 5.55 2.61 2.71 2.85 989 1.90 0.307 14.50 924 96.2 86.2
1129 68.9 14.27 91.9 5.59 2.85 3.82 3.96 1254 2.04 0.281 15.89 1094 98.1 94 .4
121 72.5 15.07 80.4 5.63 1.39 3.57 3.76 2445 0.99 0.169 7.81 567 70.8 46.5
1119 74.6 16.32 97.3 5.74 2.1 3.96 3.65 1567 1.50 0.264 12.08 901 86.0 71.8
1117 85.4 11.26 97.0 5.78 2.30 2.92 2.13 830 1.64 0.268 13.31 1136 92.2 79.1
1110 119.7 11.56 91.0 5.74 1.79 1.67 3.03 1544 1.27 0.206 10.23 1224 81.6 60.8
1221 125.5 10.67 98.2 5.44 2.10 3.25 4.34 1803 1.61 0.345 11.42 1434 87.9 67.9
all — — — 5.51 217 3.33 3.20 1380 1.57 0.268 11.93 — 88.3 70.9
mean — — — 0.15 0.41 0.72 0.99 424 0.31 0.067 2.25 _— 7.5 13.0
data
<95% — — _— 5.51 219 3.34 292 1256 1.60 0.263 12.03 —_ 87.9 71.5
mean — — — 0.19 0.14 0.54 0.99 433 0.12 0.039 0.76 — 21 4.5
aDC = dry chips

BEnergy in gas/energy in wood
Source: Walawender 1985, p. 917



droplets are collected inertially as a function of their

aerodynamic size. Once they have been separated by
size, there is no need to prevent the droplets from
coalescing. Quantities, and therefore distributions, are
subsequently determined by the relative masses
represented in each size grade.

7.8.5 Graphic Analysis of Size Distribution

The cumulative particle-size distribution shown in
Fig. 7-4 plots as a straight line on probability paper,
thereby indicating log normal distribution about a
mean particle diameter, dp at 50%, with a geometric
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Fig. 7-21. Flow rate effects on efficiency, heating value, and gas composition for rice hulls (Source: Compiled from Kaupp 1984b data)

Table 7-9. Examples of Size-Analysis Methods and Equipment

Particle Size, um

General Method

Examples of Specific instruments*

37 and larger

10 and larger

1-100

0.2-20

0.01-10

Dry-sieve analysis
Wet-sieve analysis

Optical microscope

Microscope with scanner and counter
Dry gravity sedimentation

Wet gravity sedimentation

Electrolyte resistivity change

Light scattering
Cascade impactor
Wet centrifugal sedimentation

Ultracentrifuge
Transmission electron microscope

Scanning electron microscope

Tyler Ro-Tap, Alpine Jet sieve
Buckbee-Mears sieves

Zeiss, Bausch & Lomb, Nikon microscopes
Millipore IIMC system

Roller analyzer, Sharples Micromerograph
Andreasen pipet

Coulter counter

Royco
Brink, Anderson, Casella, Lundgren impactors
M.S.A.-Whitby analyzer

Goetz aerosol spectrometer

Phillips, RCA, Hitachi, Zeiss, Metropolitan-Vickers,
Siemens microscopes

Reist & Burgess system

*This table gives examples of specific equipment. It is not intended to be a complete listing, nor is it intended to be an endorsement of any instrument.
. Source: Perry 1973, Table 20-33.
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Table 7-10. Sieve Number Versus Mesh Size

Sieve Number Mesh Size, um

80 180
100 150
120 125
140 106
170 90
200 75
230 63
270 53
325 45
400 38

standard deviation ¢ as indicated by the slope. Note
that both distributions shown in Fig. 7-4 have the same
overall slope. This slope is typical of large materials
that have been broken up into a wide range of smaller
particles. It will be helpful for us to express this slope

as the geometric standard deviation, o,

6, = dpaa/dyso = dpso/dpie (7-5)

where d 5 is the diameter for which 50% of the total
particles are captured. The other subscripts denote
similar cumulative percentages of particles smaller
than the respective particle diameter d.

The solid particles that pass through a cyclone can be
expected to have a mean particle diameter near the

cyclone cut diameter, d 5, with a standard deviation,
o, equal to 2.5, which is characteristic of cyclones,
gravity separators, and all Stokes’ law particle
movement.

7.8.6 Physical Size Analysis

The major methods for particle-size measurement are
shown in Table 7-9. Screening and microscopy are used
to determine linear dimensions. The Stokes’ radius, r,,
is the radius of a hypothetical spherical particle with
the same falling velocity and bulk density as the
particle. The aerodynamic diameter, d ,, is the
diameter of a hypothetical sphere of density 1 g/cm?
that will attain the same falling velocity as the particle
in question.

The number mass and area distributions all have the

same geometric standard deviation, G

Scrubber performance can be characterized similarly
by the size particle diameter which is captured at 50%.
The preciseness of the size cutoff point is characterized
by the value of 6. Various scrubbers and separators are
compared in Table 8-1 for cut diameter and sharpness.
Note how many have a standard deviation near 2.5.
Consider those that have a sharper or broader deviation
and why.
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Chapter 8
Gas Cleaning and Conditioning

8.1 Introduction

If the gas is to be used in a burner application, an
updraft gasifier can be used, and no cleanup will be
needed. However, if the fuel gas will be fed to an
engine, then a downdraft or other tar cracking gasifier
must be used; and the gas must be cleaned and
conditioned before it is fed to the engine.

The gas emerging from a downdraft gasifier is usually
hot and laden with dust, containing up to 1% tars and
particulates. If these materials are not removed proper-
ly, they can cause maintenance, repair, and reliability
problems much more costly and troublesome than
operation of the gasifier itself. In fact, it is likely that
more gasifier engine systems have failed because of im-
proper cleanup systems than for any other cause. In
particular, the gas is very dirty during startup and
should be burned at the gasifier until the system is fully
operational. (See Sections 9.3.3, 9.3.4 and 9.4.1 for
blowers, ejectors, and flares.)

In 1983, the Minneapolis Moline Engine Company be-
came the first contemporary engine manufacturer to
offer a 5000-h warranty on its engine—based on a fuel
gas at the engine containing less than 5 mg/Nm? of total
combined tars and particulates (Mahin, June 1983).
This amounts to 99% removal of all dust particles.

Prior to 1950, manufactured gas was widely distributed
to homes, and the technology for gas cleanup was used
extensively and well understood at that time. The
chemical and energy industries of today routinely use
the methods that will be described in this chapter.

In order to design effective gas cleanup systems, one
must determine the magnitude, size, and nature of the
contaminants, and then couple that information with
knowledge of methods available for their removal. This
chapter presents the principles of gas cleanup, the
available types of separation equipment and their

respective capabilities and suitability, and some
approaches for overall cleanup systems.

The basic cleanup system design strategy should be
based on the required cleanliness goals (determined by
the application), the order of removal, temperature,
and the intended deposit site for collected materials. In
addition, size, weight, cost, reliability, the need for ex-
otic materials, water consumption, effluents disposal,
the time between cleaning cycles, and the ease of
equipment servicing must be considered.

The first step toward producing clean gas is to choose
a gasifier design that minimizes production of tars and
particulates to be removed, such as a downdraft or
other low-tar gasifier, and to make sure that the gasifier
is operated in a manner that will minimize particulate
production by proper sizing (see Chapter 5). Develop-
ment of cleaner gasifiers is proceeding in the United
States and Europe at a good pace (see Chapter 5).

The next step, which simplifies the handling of cap-
tured contaminants, is to remove particulates, tars, and
water in the proper order and at the right temperature.
If the gas is immediately cooled and quenched in one
operation, then char, tars, and water all are removed at
one location to form a sticky, tarry mess. If particulates
are removed first at a temperature above the dewpoint
of the tars (~300°C), tars are removed next at inter-
mediate temperatures (above 100°C), and water is
removed last at 30°-60°C, then each separated con-
taminant can be handled much more easily. The rela-
tion between gas temperature and each operation is
shown in Fig. 8-1.

The final step of effective gas cleanup is to wisely
choose a site for depositing the collected materials.
Devices can be classified as either “in-line” or “off-
line.” In-line devices, such as fabric bags and packed-
fiber filters, cut off the gas flow as they become filled
with the tar or particulate material that they have

o
T>700°C T >300°C T>80°C =
&
e}
Startup Q
. burner <
Gasifier o
Particle Tar Water Engine
l separator Cooler separator ?"l Cooler l" separator Reheater 9
Grate Y ¥ . ¥
large particles Small Engine exhaust
particles heat

Fig. 8-1. Schematic relationship of gas temperature to contaminant removal
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captured. The pressure drop across the cleanup system
steadily rises with the accumulation of captured
materials, requiring frequent or automatic cleaning or
replacement. Collection efficiency is low for a clean,
in-line filter but climbs steadily with the increasing
pressure drop as the filter becomes plugged. Collection
efficiency measurements of in-line filters should clear-
ly indicate loading effects or be averaged over a full
cleaning cycle in order to be meaningful.

Off-line devices, such as cyclone separators, wet scrub-
bers, and electrostatic precipitators, deposit captured
materials outside of the flow path. These devices
separate the contaminants into one stream and the gas
into another stream. The pressure drops and efficien-
cies associated with these devices are predictable and
independent of the amount of captured materials,
eliminating the slow buildup of pressure drop with
use. Off-line methods are preferable in applications
where they can be used.

8.2 The Power Theory of Gas Cleanup

According to the contact-power theory of gas cleanup
(Perry 1973), for a given power consumption, as
measured by the gas pressure drop or water flow rate,
all cleaning devices give substantially the same collec-
tion efficiency, and the collection efficiency increases
with increasing power. Some improvement over con-
tact-power theory limitations can be gained by using
designs for reduced power consumption that use small
parallel streams, multiple stages in series, diffusion,
mass transfer, or condensation.

The performance and sharpness of size separation of
various cleanup components are compared in Table 8-1
and Fig. 8-2. Particle cut diameter (as written d . or
dpSO) is the particle diameter at which 50% of particles
are captured. A capture rate other than 50% may be
noted as a different subscript. A convenient
relationship exists for particle collectors with standard
deviation 2.5. Particles with diameters that are double

99.9 7 8
99.8 |- 10 g
99 2
S o 17 12 3
99 1
98 I > 3 6 {
90 |
S 80 | 4 8 13
g 70 F
S 60 | Scrubber Type
S 50 b 1 — Cyclone and cloth loaded
© 40 2 — Part loaded
5 ,/14 3 — Venturi scrubber (40in. wg)
£ 30 4 — Packed bed (6 in. deep, 30 fps.,
2 20 % in. spheres)
8 5 — Wet impingement scrubber
10 F 6 — Karbate
7 — Disintegrator
8 — Packed bed (6 in. deep, 5 fps,
S % in. spheres)
2 K 9 — Impingement scrubber
10 — Cyclone
1T 11 — Wire mesh (n = 2)
05} 12 — Wire mesh (n = 1)
13 — Wave plate
0.2F 14 — Cyclone and cloth clean
0.1 1 2 5 20 50 100
Particle size (d,) um

Fig. 8-2. Scrubber performance and sharpness comparison (Source: See reference in Table 8-1)
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Table 8-1. Sharpness of Particle Collection Methods

Cut Diameter

Sharpness of
Size Separation
Standard Deviation

Scrubber Type (dpso) um og = dpga/dps0 Reference
Gravity Settling Chamber 50 25 Perry 1973
Single Cyclone 1 25 Kaupp 1984a
Cascade Cyclones Das 1986
n in series identical
n=2 0.67* 1.8
n=3 0.53* 0.65
n=4 0.48" 0.55
Disintegrator 0.6 25 Perry 1973
Karbate 1 in. Ap 5 1.3 Perry 1973
Wire Mesh 2 layers 24 25 Perry 1973
(0.011in. wire) 3 layers 1.5 25
Packed Bed 6 in. deep Perry 1973
1/2 Spheres + Saddles
Velocity 5 fps 23 1.65
Velocity 30 fps 0.95 1.8
Venturi Scrubber 40 in. ApWG 0.3 1.8 Perry 1973
Sieve Plate 1.5 in. ApWG Calvert 1972
Hole Velocity Vi, = 75 fps 0.625 2
Impingement Scrubber Kaupp 1984a
4 mbar/Stage
1 Stage 1.5 6.7
2 Stage 0.076 8.5
3 Stage 0.015 20
Fabric Filter > 1.8 Peterson 1965
Spray Cyclone Dia. 24 in. 2 3 Perry 1973
Ap =2-10in. HoO V; = 57 fps
Droplet 40-200 um
Spray Tower 2-4 fps 5 25 Perry 1973
Droplet 500-1000 um
53 ft high
Wave Plate 90° Perry 1973
7 waves 7/16 in. radius 5 11

0.15 in. spacing

*Cyclone cut diameter is determined by design, down to 1 pm.
**Fabric filter collection efficiency is above 90% for all particle sizes.

the d,. value are captured with 80% efficiency, and
those with diameters triple the d . value are captured
with 90% efficiency. In other words,

dpgo = 2 dpsg, (8-1)
and
dygo = 3 dygg - (8-2)

For example, a cyclone rated at d -, = 10 um can be ex-
pected to capture 50% of particles having 10 pm

aerodynamic diameter, 80% of particles dp = 20 um,
and 90% of particles d, = 30 pm.

Note that gravity settling chamber, cyclone, disin-
tegrator, wire mesh separator, and spray tower all have
the same standard deviation—characteristic of inertial
and gravitational separation mechanisms. Sharper par-
ticle separation as indicated by smaller standard devia-
tion involves the benefit of other or additional capture
mechanisms such as condensation, cascading dif-
fusion, or mass transfer. Similarly, poorer separation
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sharpness indicates that particle separation is
degraded by some mechanism such as reentrainment
or fragmentation.

The minimum particle size and typical pressure drops
are shown for various scrubbers in Table 8-2. The
scrubber selection parameters are the particle size to be
removed, the desired collection efficiency, and the
maximum pressure drop.

8.3 Gas Cleanup Goals

8.3.1 Gas Contaminant Characteristics

Gas cleanup goals should be based on the degree of con-
tamination, the size, distribution, and nature of the
contaminants, as well as the degree of cleanliness re-
quired by the equipment. Both solid and liquid con-
taminants are present in producer gas. The solids are
char, ash, and soot, and they cover a wide range of sizes.
The liquid is initially a fine mist or fog composed of
droplets smaller than 1 um, but the droplets
agglomerate to increase in size as the gas cools.

8.3.2 Typical Dirty Gas

A typical specification for dirty gas might be
100 mg/Nm3 of particulates with mean diameter d 5, =
100 pm, a geometric standard deviation o, = 3.5, and

: . g
tar contamination of 1000 mg/Nm3.

8.3.3 Gas Cleanup Goals

The solids can be quite abrasive, and the tar mist can
cause the inlet valves, rings, throttle shafts, and other
moving parts to stick. Therefore, both contaminants
must be thoroughly removed for reliable engine opera-
tion (Gengas 1950; Freeth 1939; Goldman 1939; Kaupp
1984a; Kjellstrom 1981). Successful gasifier-engine
systems have required gas cleanliness standards from
10 mg/Nm3 to less than 1 mg/Nm?.

Many gasifiers can produce very clean tar-free gas
under certain conditions, However, it is best to design
the gas cleanup system with adequate capability for the
very dirty gas that is occasionally produced by every

Table 8-2. Minimum Particle Size
for Various Types of Scrubbers

Pressure Drop, Minimum Particle

in. water Size, um

Spray towers 0.5-1.5 10
Cyclone spray scrubbers 2-10 2-10
Impingement scrubbers 2-50 1-5
Packed- and fluidized-bed

scrubbers 2-50 1-10
Orifice scrubbers 5-100 1
Venturi scrubbers 5-100 0.8
Fibrous-bed scrubbers 5-110 0.5

Source: Perry 1973, Table 20-41.

gasifier, especially during startup, idling, and when
wet fuel is used.

8.3.4 Cleanup Design Target

Requirements for solid-particle removal may be deter-
mined from knowledge of average particle diameter d,,,
and the worst-case char-ash dust content (cg). This in-
formation can be gathered using isokinetic sampling
techniques to collect a representative sample of all par-
ticle sizes. If cq,, represents the maximum permis-
sible dust level for engine use, then the maximum
permissible dust penetration « is given by:

& = Cgmax X 1OOOA)/Cdmeasured (8-3)

On a probability plot of the size distribution of the dust
shown, for example, in Fig. 7-4, we then find the par-
ticle size d, where cumulative mass % less than d
equals a. T%is d, 50 is then the cyclone cut diameter
required for solids cleanup.

A useful rule of thumb is that the cut diameter (ds50)
required for a cyclone or scrubber will be about the
same as the diameter at the cumulative fraction cor-
responding to the maximum permissible penetration
{Calvert 1972).

8.4 Classification of Particles

Solid particles with diameters greater than 1 um are
called dust, and those with diameters below 1 pm are
referred to as fume. Liquid droplets over 10 pm in
diameter are called spray, and droplets with diameters
below 10 um are called mist. Aerosols are solids or
liquids suspended in a gas (Calvert 1972).

Dispersion aerosols are materials that begin as large
particles and subsequently are broken into smaller
sizes. They tend to be coarse with a wide size-range,
composed of irregular particles and aggregates (i.e.,
char-ash dust). Condensation aerosols are formed from
supersaturated vapors, such as tar and water mist from
chemical reactions, and soot formed from cracked
hydrocarbon molecules. They tend to be very fine and
of uniform size.

8.5 Particle Movement and Capture
Mechanisms

Methods for separating particulates from the gas stream
usually depend on the mass of the particles. The
simplest method allows the particles to settle under the
influence of gravity, with the gas stream flowing verti-
cally upward or horizontally. For horizontal separa-
tion, the process can be accelerated by providing
multiple horizontal plates. Particles also can be
separated from the gas on the basis of their mass by
using the centripetal force provided by a centrifugal
separator.
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The details, including the mathematics, of separation
can be found in Strauss (1975). However, it is interest-
ing to note that the relation between any separator’s
50% capture cut-particle diameter and the capture rate
for other sized particles is:

doso = (1/2) dpgg = (1/3) dpgg = (1/4) dgs (84

All collectors using the same capture mechanism will
be characterized by the same slope (Fig. 8-2) and
standard deviation (Table 8-1).

8.6 Dry Collectors
8.6.1 Gravity Settling Chambers

As long as unlimited space and materials are provided,
a gravity settling chamber theoretically can achieve any
level of particle separation down to the Stokes’ limit of
about 1 um. In fact, many of the earliest gasworks used
gigantic settling chambers. However, even though it is
effective, this method tends to be a bit cumbersome.

8.6.2 Cyclone Separators

Cyclones are simple and inexpensive dust and droplet
separators; they are widely used on gasifiers and will
be discussed in extra detail in this section.

Hot gas cyclone separators are well suited to remove
solid particles larger than 10 um as a prefilter for the
gas cooler and fine particle removal, as shown in
Fig. 8-3, for a vehicle gasifier of the 1939-1945 era.

Cyclone separators are also used widely in industrial
processes. The principles are well-developed, and
designs are easily scaled to the necessary size.
High-efficiency cyclone separators can be fabricated
readily by a sheet-metal or welding shop. Cyclone
design parameters are presented in this section and at
greater length in Perry (1973), Calvert (1972), and
Strauss (1975). Unfortunately, the small cyclones
required for small gasifiers are not available
commercially, so they must be custom designed and
fabricated.
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Fig. 8-3. Typical vehicle gasifier system showing cyclone and gas cooler (Source: Adapted from Skov 1974)
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8.6.2.1 Cyclone Operating Principles

A cyclone separator imparts a rotary motion to the
gases and thereby enhances the settling rate to many
times that induced by gravity alone. A cyclone
separator is essentially a gravitational separator that
has been enhanced by a centrifugal force component.
The cyclone separator grade efficiency curve, Fig. 8-4,
applies to all cyclone separators, as well as to inertial
and gravitational collectors.

Cyclone performance is rated in terms of particle cut
diameter or cut size. The cut size, d50, is the particle
size which is captured 50%.

The relationship between particle cut diameters for
this type separator is given by Eq. (8-4), where d | is the
particle diameter and the numerical subscript denotes
the collection efficiency of that size particle.

8.6.2.2 Cyclone Design Principles

The proportions for high-efficiency cyclone separators
are shown in Fig. 8-5.

The particle size that can be separated with 50% ef-
ficiency is predicted for general cyclones and for the
high-efficiency cyclone proportions of Fig. 8-6 by

dpe =V 9 ug b/12T N, V; (o, - po)l (8-5)

From Eq. (8-5) we can derive the relationship between
the cyclone separator’s particle cut size d5, and the

cyclone inlet width b for a given pressure drop as
shown in Fig. 8-6. Notice that the effect of temperature
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on cyclone cut size is minimal. This is due to the
counterbalancing variations of density and viscosity
with temperature.

If coarse particles are introduced into a fine-particle
cyclone separator, then two detrimental effects may
occur. First, the large particles may block the small
inlet. Second, the high velocities within the separator
may break up the coarse particles by erosion, impact,
and attrition. This latter effect can generate fine
particles that may be harder to capture than anything
previously present in the gas (Perry 1973). For this
reason it is preferable to provide a gas-disengagement
space for gravity settling within the gasifier rather than
two stages of cyclones. This will allow the coarser
particles to settle out prior to the cyclone separator.
Settling velocities are related to particle size as shown
in Figs. 7-6 and 8-7.

8.6.2.3 Cyclone Design Strategy

In our experience, the cyclones fitted to gasifiers are
too large for optimum particle removal. Therefore, we
present here an example of detailed cyclone design.

The diameter of pipe leading from the gasifier outlet to
the cyclone inlet should be selected to allow an ade-

quate solids-conveying velocity within the pipe. Typi-
cal solids-conveying velocities for light materials range
from 10 to 15 m/s (30-50 ft/s), as shown in Table 6-1.
One should select the cyclone inlet pipe width (b)
equal to the gasifier outlet pipe diameter or set the
cyclone inlet velocity equal to the pipe velocity and
design the cyclone according to the proportions in
Fig. 8-5. The equations that appeared previously then
can be used to predict particle cut size and pressure
drop.

8.6.2.4 Cyclone Design Example

For example, let us design a high efficiency cyclone for
a 10 kW (13.4 hp) gasifier engine system.

First we must determine the gas flow rate for a typical
22% engine efficiency and an assumed heating value
of 1300 kcal/Nm? (5.44 MJ/Nm3, 157 Btu/scf). Then
specific gas consumption is 2.2 Nm3/hp-h
(1.43 scfm/hp) per horsepower or 3 Nm3/kWh
(25 cfm/kW) (Gengas 1950).

A 10 kW engine will require 30 Nm3/h (20 scfm) of
producer gas, which corresponds to a gas energy out-
put of 163 MJ/h (188 kBtu/h). The volume of gas at the
cyclone inlet temperature of 300°C will be
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573.16 K

3
30Nm /h° ek

= 63 m3/h (37 ft3/min) (8-6)

We wish to reduce solid particulates to 10 mg/Nm?3
from raw gas that exits the gasifier at 700°C. The gas
cools within a few feet of pipe to 300°C, which we will
consider as the cyclone inlet temperature. Screening
analysis of airborne char (Fig. 7-4) shows a mass mean
particle diameter d = 100 um with a geometric
standard deviation ¢, = 2.5. Particulate sampling
indicates that total dust load at peak flow is
5000 mg/Nm3. For a turndown ratio requirement of 4:1,
the maximum dust penetration at peak flow rate is
P’p = 100 x [(10 mg/Nm3)/(5000 mg/Nm?3)] = 0.2%.
Derating by square root of turndown, the maximum
penetration for turndown is Pp, = 0.2%/V4 = 0.1%.

On Fig. 7-4 we follow the particle size distribution line
for an Imbert gasifier to the data point where the
cumulative fraction equals the maximum penetration
allowing for turndown P, (0.1%). The corresponding
particle diameter d, = 3 um is the cut point we will
require. d; 5o = 3 um, as shown by the dotted line.

The recommended minimum gas velocity for convey-
ing medium density dust is 15 m/s, and for heavy dust
(metal turnings) is 25 m/s.

A pipe 2.5 cm (1 in.} inside diameter should provide a
gas velocity of

_ 4Q
V= nD?2
4(63 m3/h)
3.14 (2.5 cm/100 cm/m)? (3600 s/h)

35 m/s (7000 fpm) (8-7)

which is well above the minimum.

Selecting the cyclone inlet width equal to the gas pipe
diameter, the cyclone is designed by the proportions
from Fig. 8-5. For inlet width 2.5 cm (1 in.) and inlet

height 5 cm (2 in.) then the cyclone inlet velocity will
be

_Q
V=2

(63 m3/h)(10* cm?/m?)
(2.5 cm)(5 cm)(3600 s/h)

14 m/s (2755 fpm) (8-8)

Cyclone cut size is the size particle that will be col-
lected with 50% efficiency. Then we calculate, using
the viscosity and density of producer gas from Fig. 8-8
and assuming ash density 2.0 (2000 kg/m?) and char
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density 0.2 (200 kg/m3), that the cyclone cut size will
be

i = ,\/ 9 ub
pe 2NeV; (pp - pg) ™

_ \/ 9(255 x 10~ kg/m-s)(0.025 m)
- 2(5)(14 m/s)(2000 - 0.489 kg/m3) (3.14)

= 2.5um (8-9)

for ash, and

4 - ,\/ 9(259 x 10" kg/m-s)(0.025 m)
pe 2(5)(14 m/s)(200 - 0.489 kg/m?) (3.14)
= 8um (8-10)
for char.
The cyclone pressure drop will be

(065)(pc)(V;H)A
ap = lgre2 !

(.065)(0.489 kg/m3)(14 m/s)?(0.025 m)(0.05 m)
(0.05 m)?

= 31 mm (1.22 in.) H,0

(8-11)
We can see that this cyclone should achieve the desired
particulate removal without excessive pressure drop.

However, there is still a finite possibility of large par-
ticles passing through a cyclone, so it is not advisable
to use a cyclone as the sole method of particulate

removal. Also, any tars in the gas stream must still be
removed by other means.

8.6.2.5 Other Factors in Cyclone Performance

The most common errors encountered in cyclone
design are a low intake velocity caused by an oversized
cyclone, and reentrainment of solid particles caused by
improper cone design or faulty design of the discharge
receiver.

Reducing the flow rate decreases the separator’s perfor-
mance, but it has only a slight effect on gas cleanliness
because the dust load entrained in the inlet gas to the
separator is also lower at the reduced flow rate. The
resulting effect is that this outlet dust load slowly in-
creases with a decreased flow rate by the inverse square
root of gas flow rate (Gengas 1950; Calvert 1972; Perry
1973).

It is tempting to reduce the cyclone inlet width with an
inlet vane; however, the upset cyclone proportions
have been found to reduce the effective number of gas
rotations N, to as low as two and to increase reentrain-
ment (Perry 1973). Therefore, to remove finer particles
using cyclones at a lower pressure drop, it is preferable
to reduce the individual cyclone diameter using
multiple parallel cyclones (multi-clones) if necessary.

Air leaking into the char-ash hopper at the bottom of a
cyclone separator deteriorates performance substan-
tially. Similarly, removing the gas through the bottom
improves the efficiency of the cyclone separator.
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Collecting entrained droplets from wet scrubbers with
a cyclone separator requires an outlet skirt to prevent
reentrainment of liquids that have impinged on the
outlet tube, as shown in Fig. 8-9.

8.6.2.6 Recent Cyclone Development

Recent work has been done on cyclone design specifi-
cally applied to gasifiers by LePori (1983).

8.6.3 Baghouse Filter
8.6.3.1 Principle of Baghouse Filters

Baghouse filters (such as shown in Fig. 8-10) are used
widely today to capture fine dust particles and to
separate flyash from combustion gases. A baghouse fil-
ter consists of one or more fibrous filter bags supported
on metal cages enclosed in a chamber through which
the gases must pass. A deposit of the separated par-
ticles soon builds up on the bag and establishes a dust
cake of appropriate pore size through which addition-
al particles cannot pass. As more dust is accumulated,
the pressure drop increases. When the cake is an op-
timal thickness for removal, the bag is agitated either
by gas pressure or by mechanical means, causing the
excess cake to drop to the bottom of the housing where
it is eventually removed.

8.6.3.2 Action of the Filter Cake in the Operation of
Fabric Filters

Fibrous bag filters have been found to be outstanding
in the removal of particles down to submicron sizes, as
shown in the grade efficiency curve of Figs. 8-2 and
8-11. High-efficiency capture of small particles is
surprisingly independent of the size of openings in the
filter weave. The reason for this is that the primary

capture element is the dynamic cake that forms on the
filter surface. This cake, which consists of captured
particles, presents a circuitous path that effectively
captures fine particles, while coarser captured
particles maintain an open cake structure to promote
high gas permeability. When a new filter fabric is
inserted, the main mechanism of particle collection is
physical sizing as determined by the openings in the
weave. At first, small particles may pass uncaptured
until some buildup accumulates on the filter. From this
point on, the gas must effectively pass through a packed
bed of micrometer-sized particles. Interception and
impaction then emerge as significant collection
mechanisms.

8.6.3.3 Application of Baghouse Filters

Baghouse filters have been used with good success in
many of the more successful and reliable engine
gasifier systems (Breag 1982; Kjellstrom 1981). The use
of fabric filters has virtually eliminated the corrosive
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ash that otherwise was present in condensate or scrub
water. The fabric filter is no doubt the most efficient
device for fine cleaning; but for wood gas, extensive
precautions against condensation of tar or water are
necessary (Gengas 1950).

Note in Table 8-3 that the cylinder wear is less for
producer gas with a fabric filter than for diesel oil
alone.

During operation, the previously described filter cake
grows steadily in thickness, collection efficiency
climbs, and the pressure drop across the filter rises.
When the filter cake has reached optimal thickness for
removal, the filter cake must be removed by one of the
following methods: momentary flow reversal to
collapse the bag and dislodge the cake as shown in
Fig. 8-10, a pulse jet of compressed gas or air to create
the momentary bag collapse, or dismantling and
manually shaking the bag (Breag 1982).

After cleaning, the filter efficiency is lower until the fil-
ter cake reforms. It is wise to use a conservatively
designed fabric filter (5-10 cfm/ft?), or even larger bag
area, to maximize the interval between bag cleanings
so as to maintain clean gas flow.

Bag filters are suitable only for removing dry particu-
lates; sticky or tacky materials do not release from fil-
ter bags. Therefore, special provisions and precautions
are required to maintain the bag filter temperature in
order to prevent water vapor or tars from condensing
on the filter bag. In particular, since tar-laden start-up
gas should not be drawn through a cold bag filter, the
design should locate the flare outlet upstream from the
bag filter and provide means for preheating the bag
filter assembly.

Materials that have been used for bag filters include
natural and synthetic organic fiber, glass fiber, ceramic
fiber, and stainless steel. The properties of these
materials are outlined in Table 8-4. Organic-fiber bag
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Fig. 8-11. Fractional efficiency curves of cyclone-alone and cyclone-
cloth collectors  (Source: Peterson 1965, p. 48)

filters are limited to low-temperature operations re-
quiring accurately controlled gas cooling. Various cloth
filters were used on gasifiers between 1939 and 1945,
but these proved to be a continual source of difficulty
because they could catch fire if they became too hot, or
would get wet (from condensate) if they became too
cold. Polyester felt bags, the most widely available, are
rated for 135°C continuous service temperature. Stain-
less steel, glass-fiber, and ceramic-fiber bag filters have
been used successfully at higher temperatures and
show good promise (Johansson 1980). Unfortunately,
the abrasion and flexing resistance of glass and ceramic
fibers can be low; after installation, and especially once
they have been heated, these materials should be han-
dled as little as possible. High-temperature bag
materials are not as widely available as other materials.

Table 8-3. Cylinder Wear for Gas Cleaned with Wet Cleaning and Fabrics Filters

Dual Fuel — Diesel Oil/Producer Gas

Diesel Oil Wet Cleaning System Fabric Filter
Cylinder wear
1000 h

Tractor 01 0.015mm 0.05

02 0.028 0.05

03 0.031 0.08 0.007

06 0.005-0.010 0.019

08 0.020 0.011
Qil contamination expressed 0.2% - 0.3% 0.54% - 1.97% 0.12% - 0.25%
as amount of insoluble products average 0.75% (2 tests)
in benzene after 100 h (9 tests)

Source: Kjellstrom 1981, Table 2.4.
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Table 8-4. Filter Fabric Characteristics

Operating Air
Exposure (°F) Supports Permeability? Resistance® to
Fiber Long Short  Combustion (cfm/ft?) Composition Abrasion  Mineral Acids Organic Acids  Alkali Cost Rank®
Cotton 180 225 Yes 10-20 Cellulose G P G G 1
Wool 200 250 No 20-60 Protein G F F P 7
Nylond 200 250 Yes 15-30 Polyamide E P F G 2
Orlon 240 275 Yes 20-45 Polyacrylonitrile G G G F 3
Dacrond 275 325 Yes 10-60 Polyester E G G G 4
Polypro(?ylene 200 250 Yes 7-30 Olefin E E E E 6
Nomex 425 500 No 25-54 Polyamide E F E G 8
Fibergiass 550 600 No 10-70 Glass P-F E E P 5
Teflond 450 500 No 15-65 Polyfluoroethylene F E E E 9

acfmy/ft2 at 0.5 in. water gauge

bP = poor, F = fair, G = good, E = excellent.
€Cost rank, 1 = lowest cost, 9 = highest cost.
dDu Pont registered trademark.

Source: After Kaupp 1984a, Table 51.



8.6.3.4 Safety Filter

If the filter bag ruptures, contaminants harmful to the
engine will be released. Therefore, a safety filter or
other effective warning means should always be used
in conjunction with bag filters. The safety filter acts by
plugging quickly and shutting down the system in the
event of an upstream equipment failure. A 200-mesh
screen is suitable for a safety filter, as shown in
Fig. 8-12.

8.6.4 Electrostatic (Cottrell) Precipitators

Electrostatic precipitators have a long history of in-
dustrial use to produce exceptionally clean gas. During
operation, the gas passes through a chamber (as shown
in Fig. 8-13) containing a central high-voltage (10-
30 kV) negative electrode. A corona discharge forms
around the central electrode, which imparts a negative
charge to all particles and droplets. The negatively
charged particles then migrate to the positive electrode,
which may be washed by a continuous water stream to
remove these particles. The electrostatic precipitator is
effective for all drop and particle sizes.

A small precipitator (20 cm in diameter and 1 m in
length) was operated at SERI to clean gas produced by
a 75-hp Hesselman gas generator powering a 15-kW
electric generator. The initial results were very
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Supporting mixer g
filter

From the cloth cleaner

Fig. 8-12. Flame arrestor and safety filter (Source: Gengas 1950,
Fig. 166)

dramatic, and the tar mist at the flare could be seen to
disappear instantaneously when the voltage was
applied. However, the electrodes and insulators soon
became coated with soot and tar, and formed a
short-circuit path that supported an arc. A means for
cleaning the electrodes must be provided, along with a
means to warm the insulators to prevent a
water-condensation short-circuit. These problems are
being investigated.

For tar-mist removal, wire and tube electrostatic
precipitators are preferred over the plate-type electro-
static precipitator (Strauss 1975). Typical performance
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Fig. 8-13. Electrostatic precipitator examples (Source: Strauss 1975, Figs. 10-19, 10-20. © 1975. Used with permission of Pergamon Press)
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Table 8-5. Typical Performance Data for Electrostatic Precipitators

Dust Concentrations

g/m3 at Operating Temp. Collecting Power Consumption
Type of Plant Inlet Outlet Efficiency (%) W/1000 m3/h
Coal Gas Industry
Peat gas producer 5.34 0.008 99.85 702
Cracking plant for natural gas 0.224 0.002 99.20 120.4
Producer gas from lignite briquettes 37.7 0.20 99.47 652
Producer gas from sembituminous lignite 28.7 0.10 99.7 602
Shale-gas cleaning plant 40.0 0.006 99.9 903
Coke oven town gas cleaning 24.15 0.010 99.9 903
Coke oven town gas cleaning 17.0 0.003 99.9 1605
Coke oven gas cleaning 28.0 0.078 99.8 752
Oil carburetted water gas cleaning 473 0.039 99.2 1404
Tar carburetted water gas cleaning 10.0 0.050 99.5 1805

Source: Perry 1973, Table 20-45.

characteristics for electrostatic precipitators are shown
in Table 8-5, indicating high capture efficiency.

The precipitator tube diameter should be small enough
to allow the corona discharge to be established at a
reasonable voltage and large enough so that its volume
will provide the necessary residence time with a
reasonable length. Low flow rates result in a higher
residence time and higher collection efficiency.

Multiple parallel precipitator tubes, as shown in
Fig. 8-13, permit a more compact precipitator design
and use a lower voltage than a single larger tube. The
sparking voltage for tube precipitators is shown in
Table 8-6. In practice, precipitators are operated at the
highest operating voltage without excessive sparking
(Perry 1973). One-second delays between sparks have
been found to result in effective precipitator operation.

Application of a negative (rather than positive) voltage
on the center electrode is favored because this arrange-
ment results in a more stable corona and less sparking.

The typical current through the electrode is low: 0.1 to
0.5 mA/m? of collecting surface (Perry 1973). Half-
wave rectification of a 50 to 60 Hz electric supply
provides adequate time for extinguishing sparks.

The power consumed by an electrostatic precipitator is
very low, typically 1.5 W/hp (Strauss 1975), and the
pressure drop also is very low, at considerably less than
1 in. of water. High-voltage equipment requires
rigorous safety measures. In addition, unforeseen
power failures may cause a loss of the electrostatic
precipitator’s cleaning ability, with a subsequent
release of tars to the engine.

8.7 Wet Scrubbers
8.7.1 Principles of Wet Scrubbers

As we have previously stated, particles with diameters
larger than 1 pm settle by gravity and inertia. They fol-
low Stokes’ law and can be captured by impaction,

gravitational, or centrifugal means. For particles
smallerthan 0.1 pm, motion is dominated by molecular
collisions. They follow Brownian motion principles,
behave more like a gas, and may be collected by dif-
fusion onto a liquid surface. In this section we will look
at the basic mechanisms of particle movement and
capture for wet scrubber systems.

Particles with diameters between 0.1 and 1 um fall
within the so-called “open window.” They are the most
difficult particles to capture, either by diffusion or in-
ertial mechanisms. They are too large to diffuse well
but too small to settle. However, they can be made to
grow in size, since small particles collide naturally and
agglomerate into larger particles that are easier to
capture.

One method of high-efficiency collection uses primary
collection of large particles by inertia and diffusion,
followed by an increase in fine particle size by ag-
glomeration, and finally by collection and entrainment
separation. The rate of agglomeration is proportional to
the total number of particles present. Agglomeration is
also assisted by the presence of droplets that act as
nuclei.

Particles tend to move toward a surface on which con-
densation is taking place. This phenomenon is referred
to as “Stefan motion.” Particles tend to migrate away

Table 8-6. Electrostatic Precipitator

Sparking Potentials
Sparking Potential,2 volts
Pipe Diameter, in. Peak Root Mean Square
4 59,000 45,000
6 76,000 58,000
9 90,000 69,000
12 100,000 77,000

aFor gases at atmospheric pressure, 100°F, containing water vapor, air,
CO2 and mist, and negative-discharge-electrode polarity.

Source: Perry 1973, Table 20-43.
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from a hot surface and toward a cold surface. This
phenomenon is called “thermophoresis.”

Wetted particles tend to stick together better when they
collide, thereby assisting agglomeration. Wet scrubbers
have been used widely, especially in stationary ap-
plications for cleaning and cooling the gas. A scrubber
operates by creating conditions for maximum contact
between the gas to be cleaned and a scrubbing liquid
medium.

Basic scrubber types and performance characteristics
are summarized in Tables 8-1, 8-2, and 8-7, and grade
efficiency curves are shown in Fig. 8-2. Scrubbers can
be divided into impingement-plate, packed-bed, sieve-
plate, spray tower, and Venturi scrubbers.

Small, difficult-to-capture droplets may be made to
grow in size with time until they are large enough to be
captured by simply providing adequate residence time
in the scrubber volume. Particles grow in size by ag-
glomeration and condensation. Agglomeration is par-
ticle growth through particle collision. Almost all
high-concentration clouds tend to have the same
particle concentration within 1 min after formation.

A novel method to capture 0.2-pm mist is to provide
water fog nuclei and ample residence time. Water fog

is introduced at a concentration of 0.25 L/Nm3, at
30 psig spray pressure, and in a high intensity sonic
field of frequency 600 to 800 Hz. A 12-s residence time
permits the particles to agglomerate to a size large
enough to be captured with 94% efficiency by a 5-um
cyclone (Calvert 1972).

If a condensation nucleus is absent, but the degree of
supersaturation (S) exceeds 200% to 400%, then
homogeneous self-nucleation occurs. Self-nucleation
produces extremely small droplet sizes. The droplet
growth rate is inversely proportional to the droplet
radius, so it proceeds slowly at first, accelerating with
droplet size.

Nucleated condensation dominates over homogeneous
self-nucleation when nucleation sites are present.
Vapor condenses more readily within concave sur-
faces, filling the void fraction of solid particles. Soluble
aerosol particles nucleate even more readily by boiling
point depression in solution. A small droplet grows
slowly by chance agglomeration until it reaches its
critical size; after that, it grows rapidly by acting as a
nucleation site. Soluble particles behave as nucleation
sites without having to achieve critical size. The char-
ash dust particles present in the producer gas stream at
temperatures below the tar dew point will act as nuclei

Table 8-7. Scrubber Types and Performance

Size Limit Particle
Cut Dia. (dpsq)

Pressure Drop
Water Column

Method pum cm (in.) Comments

Gravity Settling >30 Low Coarse separator; very
large and bulky

Massive Packing >5 1 cm/10cm of Free draining coarse demister

column height

Fiber Packing 1 10cm Viscous materials can cause
plugging

Preformed Spray >5 Low High water consumption

Gas Atomized Spray >5 0.002 (0.001) Good for sticky materials

Venturi and Sieve >2 5.7 (2.2)
Plate Scrubbers >1 22.6 (8.9)
Submicron 91 (35.9)

Centrifugal 1-2 7.5-20 (3-8) Compact; good for preliminary
cleanup

Baffle Plate 20 (solids) 2.5-7.5(1-3) Large coarse collector

5 (mist)

Impingement 2-3 10-50 (4-20) Recoil bounce; can reentrain

Entrainment Separator 5 32 (5) Can be clogged

Mechanically Aided 1-2 Acts as a blower High power and maintenance

Moving Bed 1 7.5-15 (3-6) Good mass transfer

Fabric Filter 0.3 13 (5) Excellent; can be clogged

Source: Compiled from data in Calvert 1972.
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for tar condensation, thereby reducing the amount of
very fine and persistent self-nucleated tar mist (Calvert
1972).

The design of a good scrubber must maximize the gas-
liquid contact area while minimizing the pressure drop
through the scrubber. For instance, the gas-liquid con-
tact area for a foam is much greater than for a spray,
given equal energy inputs.

If a gas stream enters a liquid-filled chamber at high
velocity through a small hole at the bottom of the cham-
ber, then all of the entering gas must experience the
subsequent impaction and diffusion environments.
When water enters the gas stream as a high pressure
spray, only a small fraction of the gas is close enough
to the nozzle to receive the benefit of impaction with
the high-velocity droplets. Spray droplet agglomera-
tion proceeds rapidly, causing the gas-liquid contact
area to drop off sharply within a short distance from
the nozzle. This effect seriously limits the collection
ability of spray scrubbers.

8.7.2 Scrubber Equipment
8.7.2.1 Spray Towers

The simplest type of scrubber is the spray tower (shown
in Fig. 8-14), which is composed of an empty cylinder
with spray nozzles. The optimum spray droplet size is
500 to 1000 um. Typical upward superficial gas
velocity for a gravity spray tower is 2 to 4 ft/s, and par-
ticle collection is accomplished when particles rising
with the gas stream impact with droplets falling
through the chamber at their terminal settling velocity.
The spray tower is especially well-suited as a prefilter
for extremely heavy dust loads (over 50 g/Nm3), which
would plug other less-open types of scrubbers. Full-
cone spray nozzles produce 500 to 1000 um droplets,
which fall with a settling velocity of 13 ft/s. For a spray
tower 53 ft high, the value of d 5 is 5 pm.

8.7.2.2 Cyclone Spray Scrubbers

The cyclone spray scrubber combines the virtues of the
spray tower and dry cyclone separator. It improves the
particle-capture efficiency of the spray droplets in
ordinary spray scrubbers by increasing spray-droplet
impact. The cyclone spray scrubber also has the
advantage, compared with the spray scrubber, of being
self cleaning, of collecting more particles regardless of
size, and operating at smaller pressure drops. A basic
design is shown in Fig. 8-9; others are described in
Strauss (1975). Commercial cyclone scrubbers are
better than 97% efficient at removing particles with
diameters greater than 1 um. The cut diameter for a
cyclone spray scrubber is about an order of magnitude
less than that for either a dry cyclone or spray scrubber.
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Fig. 8-14. Spray tower scrubber (Source: Calvert 1972, Fig. 3.1-4)

8.7.2.3 Sieve-Plate Scrubbers

A sieve-plate scrubber (Fig. 8-15) consists of a vertical
tower with a series of horizontal perforated sieve
plates. The scrubbing liquid is fed into the top of the
column and flows downward via downcomers from
plate to plate; the gas to be scrubbed is introduced at
the bottom of the column and passes upward through
the sieve holes counter to the liquid. Contact between
the liquid and gas is enhanced by using plates with
bubble caps, impingement plates, or sieve plates.

The sieve-plate scrubber captures large particles by im-
pingement and impaction, and small particles by dif-
fusion. Gas passes upward into the water layer through
holes in the sieve plate. The high gas velocity through
the sieve holes atomizes the scrubber liquid into fine
droplets, and most inertial particle collection takes
place just as the bubble is being formed, by impaction
on the inner surface of the bubble. Diffusive particle
collection dominates as the bubble rises. Here, surface-
active agents can reduce the collection efficiency be-
cause of Stefan motion, but a cold water scrubbing
liquid receiving a hot aerosol increases the collection
efficiency. A deeper foam reduces inertial effects and
increases collection by diffusion. Inertial collection is
only slightly increased by adding plates or increasing
the pressure drop.

86 Handbook of Biomass Downdraft Gasifier Engine Systems



BA-G0201778

"-.':'{,_(;?-—Foam

NN ...F:m:"—Plate
Ly L

T Gas out

N

Shetl~ @
Sieve tray ——1‘\(-‘{\ ‘ @/ :4— Liquid in
Downspout —— ((}E_;')

Sidestream
withdrawal

wt— Weir

L)
—e Liquid
out

Fig. 8-15. Sieve plate scrubber (Source: Kaupp 1984a, Figs. 141,
146)

A typical sieve-plate scrubber can attain 90% efficien-
cy for 1-um particles using 3/16-in. sieve holes, at a
specific velocity of 15 m/s (50 ft/s). Typical perfor-
mance characteristics of sieve-plate scrubbers are
discussed in Kaupp (1984a).

8.7.2.4 Impingement Plate Scrubbers

The impingement-plate scrubber shown in Fig. 8-16 is
similar to a sieve-plate scrubber, except impingement
plates are arranged so that each hole has an impinge-
ment target one hole diameter away from the hole. Gas
flow past the edge of the orifice produces spray
droplets that, when formed, are at rest, resulting in a
large relative velocity between dust particles and these
droplets. The gas velocity usually is above 15 m/s
(50 ft/s), and the typical operating pressure drop is
1.5 in. water gauge (4 mbar) per plate. An increased
pressure drop raises the collection efficiency. The re-
quired water flow rate is 1 to 2 gpm per 1000 cfm of gas
flow.

8.7.2.5 Venturi Scrubbers

The Venturi scrubber (Fig. 8-17) captures large particles
by impaction and impingement, and also rinses away
any deposits that might otherwise form. Some fine par-
ticles are also captured here by diffusion. High-velocity
flow through the low-pressure throat area atomizes the
droplets. The low pressure at the throat causes conden-
sation, and the high relative velocity of the droplets
with respect to the gas captures most larger particles by
impaction.

The atomized droplets present a considerable surface
area for fine particles to be captured by diffusion. Fur-
thermore, condensation in the throat improves capture
through diffusion because of the phenomenon of Stefan
motion. The atomized droplets rapidly agglomerate in
the diffuser section, where collection through diffusion
continues. Entrained droplets containing captured
contaminants are separated inertially from the cleaned
gas. Liquid recycle requires cooling and removal of
captured materials, or disposal and replenishment.

The collection efficiency and droplet size are deter-
mined by the pressure drop: efficiencies may be in-
creased by reducing the throat area to raise the pressure
drop. The efficiencies of Venturi scrubbers are
discussed in Calvert (1972).

8.7.2.6 Ejector Venturi Scrubbers

The velocity of the contacting liquid both pumps and
scrubs the entrained gas in an ejector Venturi scrubber,
as shown in Fig. 8-18. Spiral spray nozzles impart axial
and tangential velocities to the liquid jet. The contact-
ing liquid must be removed after the scrubber by a
suitable entrainment separator. Compared to a Venturi
scrubber, the ejector Venturi scrubber requires both
more liquid and more power to achieve the same par-
ticle collection and gas movement. Ejector Venturi

Gas Cleaning and Conditioning 87



Bubble cap
plate

s »s==5= Liquid
@ Y <= riate

N N

Baffle support

BA-GO201779

-+— |npingement
plate

Baffle

Fig. 8-16. Impingement plate scrubber (Source: Kaupp 1984a, Figs. 142,

scrubbers have no moving parts and are especially
well-suited for very dirty, corrosive, or abrasive
materials that might otherwise damage a blower
impeller (Calvert 1972).

8.7.2.7 Packed-Bed Scrubbers

The packed-bed scrubber (Fig. 8-19) is simple and open
in design, and uses spheres, rings, or saddles as ran-
dom packings to enhance the gas-liquid contact area.
Packed beds are more effective for both gas absorption
and liquid-gas heat exchange than they are for particle
collection. However, packed beds are excellent for cap-
turing entrained liquids. For entrainment separation,
the optimum superficial gas velocity for packed-bed
scrubbers using 1/2-in. spheres is 10 to 12 ft/s. Flood-
ing and reentrainment occur above a gas velocity of
12 ft/s. The pressure drop is 7.5 to 8.5 in. water gauge
for a 6-in.-deep bed. Performance characteristics of
packed beds are shown in Fig. 8-20. Packed beds are
free-draining; they may be irrigated to remove
accumulations with a water flow (Calvert 1972).

8.7.2.8 Entrainment Separators

Entrained liquids from the wet scrubber must be
thoroughly removed from the gas stream because they
carry a slurry of captured materials. Entrainment
droplets are typically greater than 10 pm and may be
captured using a variety of techniques. including a
packed bed, a packed fiber bed, a cyclone separator, an
impingement separator, a spray tower, or a settling
chamber. Poor entrainment separation has been a com-

143)

mon problem for wet scrubbers in gasifier systems. Gas
contaminant testing is advisable for all unproven
designs.

8.7.3 Auxiliary Equipment
8.7.3.1 Gas Cooling

Water vapor acts as an inert dilutent of producer gas,
initially lowering the gas heating value and ultimately
lowering engine power or burner rating, as shown in
Fig. 8-21. Much of this water vapor can be removed by
cooling the producer gas and subsequently condensing
out the water.

The amount of water vapor remaining after the cooling
and condensation steps can be determined readily from
the lowest temperature to which the gas has been
cooled. If condensation has occurred, then the lowest
gas temperature is of course the dew point of the gas
mixture. The water vapor content of the gas may be
determined from Fig. 8-22, or the psychrometric chart
of Fig. 7-19. Note that the moisture fraction roughly
doubles with each 10°C increase in the dew point
temperature. We can calculate that at the 70°C dew
point, water vapor represents 25% of the gas volume.
Cooling the gas to 40°C reduces the water content to
less than 8%, resulting in a substantial improvement
in gas quality.

Water vapor dilution will be minimized by using fuels
that are as dry as possible and then condensing water
vapor to remaove it from the gas.
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to 70% of the total energy contained within the initial
fuel.

The heat losses from surfaces vary from 1 to
5 Btu/ft2-h-°F, depending on the geometry of the gas
cooler and the temperatures involved. Thus, a great
deal of the cooling at higher temperatures can be
accomplished in the pipes and at the surfaces of the
gasifier itself, as well as in the cyclone separators or
other cleaning equipment. However, as the gas
approaches ambient temperature, additional cooling
surface through some form of gas cooler is required.

Gas coolers exchange heat between the gas and the sur-
rounding air, or between the gas and a liquid. A typi-
cal radiator used in vehicle applications is shown in
Fig. 8-3. Here, the motion of the vehicle increases air
flow around a gas cooler, so that more cooling air is
available at the higher speeds when the heat load is
greatest. In stationary applications, forced ventilation
is required to move air through the gas cooler.

As the gas cools, tars begin to condense at temperatures
below 350°C. As the temperature passes below the dew
point of the gas (typically 40°-60°C), water also will
condense. Water condensation helps to remove tar par-
ticles but yields a contaminated water condensate in
the process. If tars and particulates are removed from
the gas before it enters the gas cooler, then the gas cooler
will be able to operate longer between cleanings. All
heat-exchange and gas-cooling surfaces in contact with

Fig. 8-17. Venturi scrubber with centrifugal entrainment separator
(Source: Calvert 1972, Fig. 5.3.6-1)

When the gas leaves the char gasification zone at about
800°C, the sensible heat of the gas accounts for about
15% of the initial energy in the wood. If the gas is
burned while it is still hot, then the sensible heat can
be utilized. However, if the gas is to be used in an
internal-combustion engine, it must be cooled to
prevent preignition, to improve the engine volumetric
efficiency, and to facilitate gas cleanup.

The thermal energy in the raw gas may be either dis-
sipated, used for low-temperature applications such as
drying, or recycled into the gasifier by using the ener-
gy to preheat the incoming air. Each method has been
used on gasifiers. Airblast preheating was used exten-
sively in European vehicle gasifiers to improve the gas
and to permit wetter fuels to be gasified (Schldpfer
1937; Egloff 1941; Lutz 1940).

Using sensible heat to do more than preheat the air blast
or heat the fuel hopper is hardly justified because the
gas is dirty and only a small quantity of heat (15%) is
involved. If usable heat is desired, then the engine ex-
haust gas and engine coolant fluid are much cleaner
and more abundant sources of heat, representing 60%
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Fig. 8-18. Ejector Venturi scrubber (Source: Calvert 1972, Fig. 3.1-4)
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the raw gas are subject to ash and tar buildup, so ample
cleanout ports should be provided to clean these
surfaces without requiring extensive disassembly.

8.7.3.2 Gas Drying

The diluting effect of water vapor on the wet gas heat-
ing value (HV,) may be determined from the heating

value of the dry gas (HV,) and the moisture fraction
(F,,) from Fig. 8-22, where F = water vapor partial
pressure/total (absolute) gas pressure.

Then, HV,, is found from

HV,, = HV4(1 - F) (8-12)
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Fig. 8-19. Packed tower and packings (Source: Kaupp 1984a, Fig. 151)
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Fig. 8-21. Power loss due to moisture saturated producer gas for
150 Btu/scf dry basis gas
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Fig. 8-22. Water content of saturated producer gas (Source: Gengas
1950, Fig. 82)

(The moisture fraction F,, is approximately the same
value for either mass fraction or volume fraction of
water, since 1 Nm? of water and 1 Nm3 of producer gas
each weighs approximately 1 kg.) The effect that water-
vapor dilution has on the heating value of a 150-Btu gas
is shown in Fig. 8-21.

Methods of measuring gas moisture are presented in
Chapter 7. To minimize the power loss from water-
vapor dilution, the necessary cooling surface can be
roughly determined from Fig. 8-23. Ample ventilation
must be provided to cool and condense the gas with a
60°C dew point from 700°C down to 40°C.

8.7.3.3 Demisting/Entrainment Separation

A common problem with otherwise adequate gas
cleanup systems is inadequate removal of entrained
scrubber liquids.

The gas emerging from the gas cooler and from wet
scrubbers contains droplets of dirty water entrained in
the gas stream. Most engine trouble is caused when
these entrainment-borne contaminants form deposits
on the engine parts. Therefore, they must be removed
to finish the job of gas purification.

Wet cyclone separators (shown in Fig. 8-9), with a max-
imum spray velocity of 45 ft/s (15 m/s), are good for
removing large mechanical entrainment drops more
than 100 um in diameter but have low efficiency for
fine mist particles less than 10 pm in diameter.

Packed bed separators are good for finer droplet
removal. For example, Fig. 8-24 illustrates that a 6-in.-
deep bed packed with 1/2-in.-diameter spheres will
capture 50% of 2.5-pm diameter droplets from a super-
ficial gas velocity at 1.5 m/s (5 ft/s). Deeper beds and
finer packings will increase collection performance;
however, excessive gas velocity may cause reentrain-
ment, deteriorating overall performance. The mini-
mum packing size is limited by the fact that smaller
packings more rapidly become plugged by viscous tar
deposits. In these cases flow can be restored by stirring
or replacing the packing.

It was common practice during World War II to pass
the gas through wood chips, cork, or other fibrous
materials to remove tars. Some of these materials
subsequently can be used as fuel in the gasifier and thus
dispose of the pollutants. Fiberglass filters have been
used to clean gas (Johansson 1985) as has char
(Humphries 1985).

Fiber-type demisters have limited applications be-
cause viscous tar deposits on fine wire mesh do not
drain freely and are prone to plugging.

An electrostatic precipitator may be useful for entrain-
ment separation. However, these units have not yet
been proven reliable for continuous operation with
producer gas.

8.7.3.4 Preventing Further Condensation

The scrubbed gas may have a very high humidity (from
80% humidity to the saturation point). Further
condensation can be expected to occur either as the
pressure drops or when the producer gas is mixed with
combustion air.

To prevent further unwanted condensation, one may
heat the gas or secondary air (the engine intake air) with
engine exhaust heat as shown in Fig. 8-25.
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Fig. 8-23. Gas cooler surface requirements for various oulputs at
700°C (Source: Gengas 1950, Fig. 99)
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Fig. 8-24. Entrainment separator collection efficiency of packed bed for
droplet removal (Source: Perry 1973, Fig. 18-142)

8.8 Disposal of Captured
Contaminants

8.8.1 Char-Ash

The char-ash removed from producer gas is free of
dangerous materials and can be burned or safely
disposed of in a landfill. When burned to white ash,
char-ash contains valuable minerals that may be
beneficially returned to the soil. Charcoal is a valuable,
clean-burning fuel, worth several times the value of
wood. Alternate uses and possible markets definitely
should be examined.

Larger char material may be saleable for further
charcoal gasification, combustion, or briquetting.

In addition to valuable soil minerals, charcoal has been
successfully used as a soil conditioner in Japan, result-
ing in improved crop yields (Kishimoto 1985). It also
has been used as a livestock feed supplement to reduce
digestive problems and meat hormone levels (Taylor
1986). Charcoal has long been considered a premium
cooking fuel in many developing countries.

8.8.2 Tar

A gasifier that produces more than 0.5 g/Nm?3 of tar can-
not be suitably cleaned for engine applications due to
the large amounts of tar that must be captured and dis-
posed of. For a worst-case tar production scenario of
2 g/Nm? (0.2%), up to 2 g of tar per hp-h may arise with
each horsepower. Thus, a 100-hp engine would
produce up to 4.8 kg of tar in 24 hours, or about 5 L. At
room temperature, tar is a viscous, slow-flowing,
molasses-like fluid. It may contain carcinogenic sub-

stances and should be treated with due caution. Col-
lected tars that cannot be recycled to the gasifier or
burned on site may need to be treated as hazardous
wastes. They should not be dumped on the ground or
released into waterways. Prevention through low-tar
gasifier designs is the best cure for tar (see Chapters 4
and 5).

8.8.3 Condensate

The liquid condensate from producer gas may contain
substantial amounts of tars and phenols. Phenols are
well-known germicides and will kill soil bacteria if
spread on the ground. If released into waterways, the
condensate could damage lifeforms supported by those
waterways.

For condensate water, as with tars, prevention is the
best cure. Methods of minimizing condensate forma-
tion should be considered fully early in the design and
selection of the system (see Chapter 5). Gas moisture
content may be minimized by using dry fuel (the drier,
the better) and by recycling heat back into the gasifier
through an air-blast preheater. Condensation from the
gas may be minimized by limiting the amount of gas
cooling so as to use the gas above its water dew point
(40°-60°C). The loss in engine power or the costs of
larger engines may be more than offset by the savings
in the cost of condensate disposal.
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Fig. 8-25. Heating of secondary air by exhaustgas heat from the engine
(Source: Gengas 1950, Fig. 85)
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Chapter 9
Gasifier Systems

9.1 The Complete Gasifier System

The previous chapters have discussed the major opera-
tional components of a gasifier-engine system.
However, no system is stronger than its weakest link.
A complete system requires means to store and possi-
bly dry the biomass, to feed the biomass, to remove
char-ash, to push or pull the gas through the system, to
clean the gas, and to burn the gas during startup, as
shown in the front of this book and in Fig. 9-1.

A complete system also requires instruments to
measure pressure, flow rates, and temperatures at cru-
cial points, and controls to establish the required con-
ditions. Instruments and controls will be discussed in
Chapter 10.

9.2 Storing, Feeding, and Sealing
Solids _
9.2.1 Characteristics of Solids

Solids are many times more difficult to feed and seal
against gas flow than liquids and gases. Because they

can support a shear stress, solids can bridge and arch
in cylindrical channels. An important measure of the
difficulty of feeding a particular solid, the angle of
repose, is the average angle from a horizontal plane as-
sumed by individual biomass pieces when they areran-
domly piled up. For liquids, this angle is zero; for some
solids, it can be greater than 90°! For this reason,
vibrators, shakers, rakes and chains, live bottoms (on
trucks), and a host of ingenious devices are used wide-
ly in industrial and agricultural solid-feeding applica-
tions. Much time and money can be wasted reinventing
these devices, so the designer is advised to contact
others with experience in feeding the particular form
of biomass being used.

9.2.2 Storage

A closed bin, silo, or hopper must be supplied to hold
the biomass feedstock (chips, cobs, pellets, etc.), to
prevent it from getting wet. In many cases, industrial
or agricultural containers are available in appropriate
sizes at low cost.
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Fig. 9-1. Gasifier system showing means of moving solids and gases and positions for various instruments and controls
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9.2.3 Feeding Solids

During testing, fuel can be fed manually to small
gasifiers. However, there is the danger of running out
of fuel, which in turn overheats the gasifier as the
remaining charcoal burns. Level alarms or other con-
trols are strongly recommended for gasifier systems
intended for continuous operation (see Chapter 10).

Biomass fuels are only partially free flowing from a
hopper by gravity alone so bin stirrers, vibrators, or
shakers may be required for even fuel delivery. Biomass
can be moved laterally and vertically by conveyor belts,
chain drags, bucket elevators, augers, pneumatic
blowers, and vibratory feeders (e.g., Syntron type),
widely available in agricultural handling equipment.
Again, those with experience with the particular
biomass form should be contacted for feeding and
equipment suggestions.

The flow of solids in the gasifier is also subject to ir-
regularity and interruption and can cause great dif-
ficulty during gasification, resulting in such problems
as bridging, caking, channelling, and rat-holing. The
importance of uniform feeding cannot be overem-
phasized and experts in the field should be consulted
(Guzdar 1982, Miles 1982).

Char-ash must be removed from the gasifier and stored
as it is produced. An air-tight char-ash receiver should
be provided, since this char material is combustible
and may reignite spontaneously. In addition, it may be
necessary to cool the receiver. Although the weight of
the char-ash may be only 2% to 5% of the weight of the
biomass fed to the gasifier, its volume may represent a
larger fraction of the volume of the original biomass be-
cause of its lower density. The required receiver
volume must therefore be calculated from measured
char-ash bulk densities, which may range from 0.064
to 0.4 g/cm3 (4-26 1b/ft3).

It should be mentioned here that ash receivers can con-
tain explosive gas even when cold. They have been
known to ignite on startup, and precautions should be
taken against this.

9.2.4 Sealing Solid Flows

Gasifiers may operate at pressures up to 20 in. water
gauge above or below atmospheric pressure, which
makes it necessary to provide a seal through which the
biomass can pass without leaking air or gas. Proper
seals are very important, to ensure both gas quality and
safe operation (see Chapter 12).

Gasifiers from the World War II era were batch-fed
through a lid that could be sealed tightly, as shown in
Fig. 9-2. The spring-loaded lid would pop open in the
event of an internal gas explosion. As long as the
gasifier was filled quickly, the expelled smoke could
be tolerated as a nuisance.

4 i
Fig. 9-2. Batch fed gasifier with lid (Source: NAS 1983, p. 68)

There are two basic types of solids feeding and sealing
devices—mechanical seal type where the seal
mechanically prevents gas passage, and plug seal
where the fuel acts as its own plug and seal such that
fuel velocity into the gasifier is greater than gas velocity
out through the fuel plug. It should be noted that rotary
valves and gate valves are also good firestops for flash-
back or explosion prevention. Inert or air purge gas
should be used in pressurized gasifiers to offset leakage
through rotary valves. Additionally, there is the
stratified charge gasifier where air enters through the
top at atmospheric pressure such that for operation
above a minimum gasification rate, all smoke is drawn
down into the fuel bed, and a lid is unnecessary. In
many cases, the biomass feed can help to act as a seal
in a long auger or vertical pipe. However, the pressure
drop through the fuel is small, and the technique will
not work if the gasifier fuel inlet is under a positive
pressure of more than 2.5 cm (1 in.) water gauge.

Various solids feeding devices are shown in Fig. 9-3.
Star valves, which rotate to feed the fuel, are available
commercially.

Ifa gasifier is to be operated at high pressure, it becomes
exceedingly difficult to feed biomass through a single
seal. Lock hoppers that use two slide or bell valves
supplying a metering feeder as shown in Fig. 9-4 have
been used with biomass at pressures up to 30
atmospheres for making medium-energy gas for
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Fig. 9-3. Solids feeding devices (Source: Perry 1973, Fig. 20- 79. © 1973. Used with permission of McGraw Hill Book Co.)

pipeline distribution or syngas for chemical synthesis
(Reed 1982).

9.3 Fans, Blowers, Ejectors, and
Compressors

9.3.1 Importance of Gas-Moving System
Design

It is important to provide a suitable method for pulling
or pushing the gas through the gasifier, and since the
mass of gas and air being moved is much larger than
the mass of fuel being fed, considerable power may be
required. The engine being fueled can serve this
purpose.

Every internal-combustion engine is a compressor,
since it compresses the intake air and fuel to 10 to 30
times atmospheric pressure before igniting the fuel.
When an engine operates on producer gas, it can also
provide suction and compression for the gasifier.
However, an engine is a very expensive compressor for
testing new gasifiers (Arthayukti 1984; Breag 1982). It
is desirable to use some other method for moving the
gas that is less sensitive to tar, char, and soot during
testing. Full engine power also may require some type
of compressor (see Chapter 10).

The question of whether the gas should be pulled (suc-
tion operation) or pushed (pressurized operation)
through the gasifier is important, and one finds strong
advocates of each method. Gas leaks from gasifiers
operating above atmospheric pressure can be
dangerous because of the possibility of leaking carbon
monoxide out of the gasifier; air leaking into gasifiers
operating below atmospheric pressure can cause
explosions.

9.3.2 Fans

Propeller-type fan blades usually generate under
1.25 ¢m (0.5 in.) of water gauge pressure and are used
in gasifiers only to move air through heat exchangers
and radiators. They are not suitable for moving gas
against any resistance.
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9.3.3 Blowers

Centrifugal blowers (Fig. 9-5) can generate pressures on
the order of 100 cm (40 in.) of water gauge pressure and
are quite suitable for gasifier testing. To generate these
pressures, the blowers must either rotate very fast or
have a large diameter, since it is the centrifugal force
that creates the pressure. The blower can tolerate, and
in fact will remove, a certain amount of tar and particu-
lates, but a means for draining and cleaning the blower
should be provided. Blowers can be used eitherto push
the air into the gasifier or to pull the hot gas through
the system at negative pressure. Considerably more
power is required to pull the gas through the system
than to push air because there is necessarily more mass
to manipulate and the gas is less dense. In addition,
suction fans must be capable of handling a higher
temperature than fans pushing air into the gasifier.
Most blower breakdowns occur due to deposits on shaft
seal and impeller or erosion of the case. Reliability is
limited by deposits.

9.3.4 Ejectors

Ejectors are a very convenient and simple means for
moving dirty gas. No moving parts are exposed to gas
contaminants. Ejectors (Fig. 9-6) use the motion of a
small amount of one gas to move larger quantities of a
second gas, often at negative pressures. During startup,
the gas produced initially is very tarry and may quick-
ly clog cleanup and engine. Therefore, one should use
a fan or ejector during startup to send this gas to a
product-gas burner until low-tar operation is reached.
Compressed air, nitrogen, or steam can be used to drive
the ejector. Water jets can also be used to move, cool,
and clean the gas.

Ejector design is based on the principle of the conser-
vation of momentum of the driver gas as it aspirates

r Mixing _—-r_ Diffuser —-’1
section
~

i)
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Fig. 9-5. Centrifugal blowers (Source: Perry 1973, Figs. 6-37, 6-38,
6-39, 6-41. © 1973. Used with permission of McGraw Hill Book Co.)

Fig. 9-6. Ejector pump formovinggas (Source: Perry 1973, Fig. 6-73.
© 1973. Used with permission of McGraw Hill Book Co.)
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and mixes with the driven gas. Optimum ejector
dimensions are discussed in Perry (1973).

9.3.5 Turbochargers and Superchargers

The power output of internal-combustion spark and
diesel engines is directly proportional to the energy
content of the intake fuel-air mixture. A mixture of
producer gas and air has 30% less energy than a
gasoline and air mixture, resulting in a minimum 30%
power loss at any given rpm. Intake pressure can be in-
creased to overcome this power loss by a turbocharger
using engine exhaust pressure to run a turbine, or a su-
percharger operating from the engine shaft power. This
pressure boost is widely used in diesel engines and
racing cars, and is coming into wider use even for
spark-ignition engines.

Positive displacement rotary blowers (Fig. 9-7) and su-
perchargers can achieve any pressure required, but
they do so at higher capital, operating, maintenance,
and energy costs. This increased cost must be weighed
against the lower cost of using a larger engine.

9.4 Flares and Product-Gas Burners
9.4.1 Flares

Flares sometimes may be seen at oil wells or refineries
in which excess gas burns with a luminous flame. In
order to produce a nonluminous flame, it is necessary
to provide enough air and residence time to burn the
soot in a hotter, nonluminous flame. This is called a gas
incinerator.

Raw producer gas contains up to 40% carbon
monoxide and up to 20% volatile tars, making it
absolutely essential that a reliable incinerator be
available during testing to burn the gas. The incinerator
must be sized to fit the gasifier. Most of the principles
discussed below for incinerators also apply to
developing burners for producer gas.

The three essential elements necessary to combust any
gas are residence time, temperature, and turbulence
(the three “T’s” of gas combustion). Residence time re-
quires a sufficently large chamber for combustion to
proceed to completion. High temperature is achieved
by using a refractory lining on the burner. Turbulence
can be generated by high-velocity mixing of the com-
bustion air or fuel (for instance, by passing it through
anozzle) or by tangentially mixing air with the gas. The
reader is referred to books on combustion and burners
for a more complete discussion (e.g., Perry 1973).

Incinerators for toxic chemicals and gases require a
pilot flame to assure combustion operated on methane
OT propane, an ignitor to start the flame, a flame sensor
to shut down the burner if the flame is extinguished,
and a control system to regulate the air mixture and
stack temperature.
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Fig. 9-8. Product-gas burner for testing small gasifiers

A small flare, suitable for burning up to 200,000 Btu/h
of producer gas, is shown in Fig. 9-8. It is made from a
5-gal can lined with an 8-in.-ID ceramic fiber risa
sleeve. (Risa sleeves are used in metal foundries, high-
temperature flue lines, and oil-burning furnaces.) A
propane torch or suitable pilot flame injected tangen-
tially at the bottom maintains a high temperature on
the wall. Producer gas is injected tangentially upstream
from the propane flame and mixes with the propane
flame as a source of ignition. Air is drawn through the
bottom port to permit complete combustion. Smaller
or larger flares can be built on this principle, or com-
mercially available burners and gas incinerators can be
used.

9.4.2 Burners

Close coupled gasifier burners offer improved clean,
high efficiency burning of solid fuels compared with
conventional solid burners. Gas-air mixture and
mixing are more easily controlled than are convention-
al solid fuel burners, resulting in more complete com-
bustion. Equipment costs for retrofitting an existing gas
or oil unit for close coupled gasifying are generally less

than the costs of replacing the unit with a solid fuel
burner.

Close coupled gasifier burner systems are able to meet
emission requirements with no expensive pollution
abatement equipment.

Updraft producer gas is an excellent fuel for high-
quality heat applications. The high tar content does not
need to be removed, and adds to the heating value. Fur-
thermore, the sensible heat of the gas adds to the flame
temperature and overall heat output. TIPI workshop
(Das 1986) has been using updraft producer gas from
wood chips to fuel a melting furnace, melting copper
and bronze at high efficiency using 2 to 3 b wood chips
per pound of bronze poured and a rapid heat cycle (20
to 30 min) to reach pouring temperature.

The flame inside the forge is very different from any
other kind of wood burner. It has the intensity and fury
of a jet engine and the clean invisible flame of a high-
quality fuel. Direct combustion of producer gas lends
itself to a wide range of applications (e.g., ceramics,
glass, steam, drying, blacksmithing, and process heat).
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Chapter 10
Instrumentation and Control

10.1 The Need for Instrumentation and
Control

The gasifiers of World War II were batch-fed, and gas
flowed to the engine in response to the vacuum created
in the engine. As a result, usually there were no con-
trols or instruments other than those of the engine and
vehicle. The corresponding tradeoff was that a great
deal of operator expertise was required in order to
know when to shake the grate, stir the fuel bed, fill the
fuel hopper, and clean the system, as well as how to
operate the vehicle on hills and in traffic.

Just as the modern automaobile has a number of instru-
ments, sensors, warning lights, and controls, gasifiers
built today should incorporate instruments and con-
trols to improve performance, convenience, and safety.

During research and development, many measure-
ments are required to determine operating parameters
and to determine where controls should be applied.
However, a production gasifier system (like a produc-
tion car) should require relatively few key instruments
and controls.

Ideally, gasifiers of the future will operate automatical-
ly and unattended. A working gasifier system requires
the integrated operation of a number of components. It
is desirable to have automatic controls and warning
systems on production gasifiers in order to keep the
system in balance and to warn the operator when a par-
ticular component needs attention. Although such a
degree of automation would have been impossible to
implement on the gasifiers of World War II, reliable, in-
expensive sensors, instruments, and controls are wide-
ly available today in much of the world, due largely to
the development of the modern automobile and solid-
state technology.

This chapter deals with the methods and equipment
that have been used to measure each physical quantity
critical to optimal gasifier operation. Since these
measurements are commonly used by chemical
engineers, the reader is referred to chemical
engineering texts, and especially to handbooks such as
the Chemical Engineers’ Handbook (Perry 1973).

10.2 Gasifier Instruments

10.2.1 Pressure Measurement

In experimental work, pressure drops may be measured
routinely across the gasifier bed, the orifice plate flow

meters (see below), the gas cleanup system and com-
ponents, and any components that are prone to plug-
ging, as shown in Fig. 9-1. The total pressure with
respect to atmospheric pressure (known as “gauge”
pressure) may also be measured at the gasifier outlet,
the cleanup outlet, and (if the gasifier is operated at a
pressure above atmospheric pressure) the air inlet. In
locations where occasional measurements are suffi-
cient, connections should be capable of being closed
off when they are not in use.

The pressures within the gasifier will be close to atmo-
spheric pressure (except for high-pressure gas
producers) and generally will be measured in cen-
timeters (inches) of water column.* Pressure drops and
differential pressures can be measured by a U-tube
manometer filled with colored liquid. Liquid-filled
manometers are simple devices that use a fluid-filled
(usually colored water or light oil) tube to measure
pressure. For convenience and portability, they can be
made from flexible, transparent plastic tubing and a
meter stick. Commercial units are available in a wide
range of accuracies, from 0.25% to 3% of the full-scale
reading.

For more sensitive measurements, an inclined-tube
manometer may be used (Dwyer 1960). Typical
inclined-tube scales can be read to 0.1 in. water gauge,
and curved-tube manometers (e.g., the Dwyer Mark II)
provide high resolution over a wide range of readings.
Commercial units are available for as low as $10, dual-
range vertical inclined manometers start at $30, and
stationary high-resolution manometers can cost as
much as $400. Special handling is required to avoid
losing or blowing out gauge fluid because of excessive
pressure. Accurate leveling is important for all
manometers.

Diaphragm-type differential pressure gauges give a
needle-pointer reading of the differential pressure.
They are available in full-scale accuracies of 2% and
full-scale ranges from 0.25 to 150 in. water gauge (e.g.,
those manufactured by Dwyer). These gauges are
rugged and can be used in all positions with positive
and negative pressures.

For nondifferential pressures in excess of a few pounds

per square inch, the well known Bourdon gauges
should be used.

*1029 cm water = 1 atm; 406.8 in. water gauge = 1 atm; 26.4 in. water
=1 psi.
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10.2.2 Gas Flow Measurement
10.2.2.1 Floating Ball Rotometers

Clean gas flow can be measured using a rotometer (ball
and tube) flowmeter, which comes in a wide range of
sizes up to 100 scfm. The gas must be clean at the
rotometer, and therefore rotometers usually can be
used only on air or oxygen streams at the gasifier inlet.
In addition, rotometers must be calibrated for the
specific gas to be used, and the reading also depends
on the absolute gas pressure. Nevertheless, they are
very convenient flow measurement devices.

10.2.2.2 Gas Flow by Differential Pressure
Measurement

Gas flow may be measured by a number of methods that
produce a differential pressure signal. Pitot tubes,
orifice meters, Venturi meters, and flow-restriction
meters all are based on measuring the pressure drop
produced by gas flow across or through the device. For
each technique, the velocity is proportional to the
square-root of the pressure drop (Ap).

Pitot Tubes. The standard Pitot tube shown in
Fig. 10-1 is recognized as a primary standard for
velocity measurements. When properly designed, it
has a unit calibration factor, K = 1. For pipes under
10 cm (4 in.) in diameter, the average velocity is 90%
of the center-stream velocity, £5%. For air, the gas
velocity (ft/min) is given by the equation:

V =1096.7 K [Ap Tyog Peta/ (G Tstd Pres)]?’% (10-1)
where:
pc = gas density at standard conditions
Ap = the differential pressure in inches of water
P = gas pressure absolute (atmospheres)
T = gas temperature (Kelvin)

K = calibration constant

The subscripts std and test refer to standard conditions
and actual test conditions, respectively.

The standard Pitot tube shown in Fig. 10-1 introduces
the smallest pressure drop of any pressure-sensing flow
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Fig. 10-1. Standard Pitot tube (Source: ASME 1980, Fig. 1-1)
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method. However, the small openings are easily
clogged by dirty gas. The “s” type Pitot tube shown in
Fig. 10-2 is more suitable for dirty gas. Its calibration
factor is approximately K = 0.83 for velocities up to
50 ft/s. For more accurate work, the individual probe
should be calibrated against a primary standard.

The low pressure signal characteristic of Pitot
measurements requires a sensitive and delicate
readout. A typical gasifier pipe velocity of 20 ft/s will
show a pressure drop of 0.09 in. water gauge. This
pressure drop can be measured by a Dwyer
Magnehelic® 0-0.25 in. water gauge pressure gauge,
which has a resolution of 0.005 in. water gauge, a
pressure readability of 5%, and a velocity readability
of 2.7%.

The velocity of a gas in a pipe is highest at the center
and zero at the wall. The gas velocity can be used to
measure flow by traversing the velocity profile and in-
tegrating the result.

Orifice Meters. The orifice meter shown in Fig. 10-3
gives a much higher pressure reading than a Pitot tube,
but it does so at the expense of creating a slightly higher
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pressure drop in the system. If the added pressure drop
can be tolerated, then the orifice meter is usually
preferred over the other alternatives because it is low
in cost, can use more rugged, less sensitive readout, and
is not sensitive to small amounts of tars and particu-
lates. An orifice meter consists of a washer-shaped
plate with a hole diameter d that is placed in the gas
line, with a plate thickness at the orifice edge of no
more than 3% of the pipe diameter, D. If the orifice is
an accurately centered, round opening of diameter d,
the diameter ratio, d/D, should be between 0.3 and 0.6.
A means for measuring the differential pressure drop
across the plate, as shown in Fig. 10-3, must be
provided.

The calibration can be predicted from the gas proper-
ties and dimensions of the pipe and orifice using for-
mulas given here and in more detail in chemical
engineering texts and handbooks, such as the Chemi-
cal Engineers’ Handbook (Perry 1973). The location of
pressure taps on an orifice meter will significantly af-
fect the calibration. For an accurate measurement, the
orifice meter should be calibrated against a primary
standard such as a Pitot tube or dry gas test meter. Al-
though small deposits of tars will not plug the meter,
they may alter the calibration; frequent cleaning or
recalibration may be required. To prevent fouling from
char or condensate, the orifice opening may be moved
to the bottom of the pipe to permit free passage of solids
and liquids. Calibration is advisable against a standard
orifice meter or other known standard. If the gasifier
uses a cyclone, then it can be used as a convenient flow
meter by locating temperature and pressure taps at the
cyclone inlet and outlet. Calibration again may be
predicted from gas properties and cyclone dimensions
or calculated against a known standard.

Flow is related to pressure by the equation

— APTO — 2 _ApTo
Q=K \,———TG =aD \f T

(10-2)
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T, = calibration temperature

—3
«Q
|

= operating gas inlet temperature
Q = flow rate
K = overall calibration for specific flow meter

a = general constant for ratio d/D and geometry
(from Fig. 10-4)

Ap = pressure drop across flow meter
D = pipe inside diameter
d = orifice diameter

It is convenient to plot Q versus Ap for a handy
reference. Dimensions and calibration constants are
shown for several orifice meter sizes in Fig. 10-5.

Rotational flow caused by pipe fittings can cause
serious metering errors. It is good practice to provide
at least eight pipe diameters upstream and three pipe
diameters downstream of the orifice to minimize
rotational errors.

Venturi Meters. The Herschel-type Venturi flow meter
shown in Fig. 10-6 provides a higher pressure signal
with a minimum pressure drop across the meter, be-
cause the divergent downstream section of the meter
conserves gas momentum by converting velocity back
into pressure. In addition, the Venturi meter does not
present abrupt surfaces that otherwise would be sub-
ject to impact by tars and particles, so contaminant
buildup is minimal and the maintenance required for a
permanent installation is reduced. The cost of a Ven-
turi meter is the highest of any of the differential pres-
sure methods of velocity measurements presented here
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because of the amount of precise machining required
for its production.

10.2.2.3 Positive-Displacement Meters

The positive-displacement dry gas meter is a primary
gas volume standard. It measures the accumulated total
flow of a gas and is very accurate over a very wide range
of flows. It is relatively inexpensive because of its use
by the gas industry. Numerous moving parts are ex-
posed to the gas, so the gas must be clean and dry. Its
main applications are in measuring sampled dry gas in
a contaminant sampling and for calibrating other flow
meters. .

10.2.2.4 Tracer-Gas Flow Measurement

Total gas flow can be measured by injecting a small, ac-
curately measured flow of tracer gas into either the air
or producer gas stream. The tracer gas must be one not
normally present in producer gas, such as helium or n-
butane, and capable of being measured with high
precision. Alternatively, a gas that is normally present,
such as nitrogen, can be injected intermittently into the
stream, and the flow can be calculated from the chan-
ges of concentration in the product gas. Both of these
techniques of tracer-gas flow measurement require ac-
curate gas chromatographic analysis of the sample to
determine the flow rate.
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10.2.3 Solid Flow Measurement

Most World War II gasifiers were batch-fed, so it was
very easy to weigh the biomass fuel as it was fed in
order to record biomass consumption. However, this
technique results in only a long time average
measurement of the actual feed rate.

Many gasifier systems will be fed automatically, using
level controls to determine the feed rate, as shown in
Fig. 9-1. If the change in weight between levels is
calibrated, then recording the number of feed cycles al-
lows one to estimate biomass consumption. A more ac-
curate technique involves a weighing means that is
inserted along the feed train to measure the amount of
feed delivered to the gasifier.

Alternatively, the gasifier itself may be weighed to
record both the entry and consumption rates of fuel.
Flexible connections on the gasifier are needed, so that
the gasifier can be weighed free from the variable forces
of the connections.

10.2.4 Temperature Measurements

Low temperatures (up to 300°C, 572°F) can be visually
indicated with mercury thermometers or bimetallic-
stem dial thermometers. Alternatively, thermistor sen-
sors can be used in this temperature range to provide
an electrical signal that can be used for automatic
recordkeeping and/or control purposes.

Chromel-alumel (type K) thermocouples can be used
continuously to 1000°C (1800°F) and intermittently to
1200°C, and they provide an almost linear electrical
signal of 40 pV/°C. Sheathed thermocouples should al-
ways be used in connection with producer gas applica-
tions, because thermocouple alloys will react with hot
reducing gases, H, and CO, changing the output volt-
age calibration. Thermocouples are used widely in in-
dustry for temperature measurements and are available
from a number of sources.

Optical pyrometers can be used from about 300°C (in-
frared) or 700°C (visible) to 4000°C. They require a
viewport and are generally hand-operated. They are
useful for spot-checking surface temperatures.

Of these methods, thermocouples are best suited to
most gasifier measurements. Thermocouples are avail-
able with either analog or digital readouts, and the sig-
nal can be recorded electronically. Analog
thermocouple meters may cost as little as $50, but they
require a fixed lead resistance. They are used widely in
pottery and foundry trades. Digital pyrometer readouts
start at $200. They read to greater precision than
analog, and due to their high input impedance, lead
length is not critical.

10.3 Controls

For commercial application, gasifiers must be safe, de-
pendable, and convenient. Automatic and unattended
operation, an eventual necessity, will require suitable
control and warning sensors and mechanisms.

10.3.1 Fuel-Level Controls

If a gasifier continues to operate after the biomass has
been consumed, there is a danger of damaging the high-
temperature region of the gasifier because of the ex-
traordinarily high temperatures generated during char
gasification and combustion. Therefore, level controls
should be installed in the various fuel hoppers and the
gasifier itself to warn when the biomass level is getting
low.

A number of level indicators are available on the
market for signaling the level of solids and liquids in
containers. These operate on the basis of light or sound
signals, bin wall pressure, resistance to vibration or
rotation, and absorption of radiation. Suitability
depends on vulnerability to failure from clogging or tar-
ring the device. After installation, the control should
be repeatedly tested to ensure that it is functioning,
since we have seen these devices clog or not function
for many reasons not anticipated by either the user or
manufacturer.

10.3.2 Pressure Controls

Simple electromechanical switches that sense absolute
or differential pressure are available. These switches
provide on/off switch closure signal. An especially at-
tractive, inexpensive, low pressure differential switch
is the Dwyer Minitactor®.

Special pressure or differential pressure transducers
are needed to produce analog electrical signals for an
analog programmable controller, a control console, or
data acquisition.
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10.3.3 Temperature Controls

Thermistor, bimetallic, and thermocouple switches
and controllers activate a switch closure in response to
high- or low-temperature conditions to either take
corrective action or give warning signals.

Additionally, thermocouples and thermistors generate
analog electrical signals, which can be used by suitable
proportional controllers.

10.4 Computer Data Logging and
Control

Today, low-cost computers can exercise very sophisti-
cated control over most processes in response to
suitable signals, as can be seen under the hood of any
modern car. Computer data recording and control
should be considered in any research and development
program and for any commercial gasifier.
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Chapter 11
Engine Adaptation and Operation

11.1 Introduction

Producer gas technology is useful to provide clean
combustion heat, shaft power, and electricity from
widely available biomass fuels. Historically, engine
shaft power generation has been the major application
of small gas producers. Large producers, on the other
hand, made “town gas,” which was piped for cooking,
lighting, and heating.

The major use of small producers during World War 11
was for transportation, especially trucks and buses. Al-
though vehicles could become a major use of producer
gas again, especially during a liquid fuel emergency,
the added difficulty of using solid fuels for transporta-
tion does not make producer gas competitive with
gasoline today. If programs for making synthetic liquid
fuels (methanol, gasoline, or diesel) from biomass or
coal are successful, producer gas may never be needed
for transportation applications.

11.2 Producer Gas for Transportation

Transportation applications place a very heavy
demand on producer gas systems: the system must be
small, lightweight, and compact; operate at widely
varying load conditions; have fast response times; be
low in tar; be low in cost; be safe; and be convenient to
use and service.

Currently, some vehicles are being powered regularly
by gasifiers in Brazil and the Philippines (Mahin, June
1983). Sweden has maintained a stockpile of vehicle-
ready gasifiers as part of its national preparedness
strategy since the Suez oil crisis. A handful of groups
offers plans for vehicle gasifiers (Mother 1982; Nunnik-
hoven 1984; Skov 1974). Various American vehicles
have been converted and operated for demonstration

purposes; however, regular use of these vehicles is rare
(NAS 1983).

11.3 Producer Gas for Electric Power
and Irrigation

Compactness and low weight are not as important for
power and irrigation purposes as they are for transpor-
tation. Electric power production and irrigation place
minimal requirements on turndown and response time
of gas producers. The gas flow required for a
synchronous generator varies over only a 3:1 ratio as
the electric output goes from full load to no load.

11.4 Gasifier Types Suitable for
Shaft-Power Generation

Updraft and fluidized-bed gasifiers have the slowest
response times of the gasifier types and cannot be ex-
pected to follow changing loads with favorable results.
The gas from both updraft and fluidized-bed gasifiers
also contains large quantities of tars, making these
gasifier types unsuitable for engine applications.

The fastest response time is obtained from crossdraft
gas producers, but they are suitable only for low-tar
fuels such as charcoal. Although low-tar operation has
been observed in crossdraft gasifiers, nozzle spacing is
very critical and load variations can cause tarring.

Downdraft gas producers provide a low-tar gas product
from biomass and also have a rapid response time, so
they are suited for powering engines with either
varying or fixed loads. Low-tar gasifiers now under
development incorporate design changes that recycle
some of the gas with additional air to give even lower
tar values (on the order of 50 mg/Nm?3) than
conventional downdraft gasifiers (see Chapter 5 and
especially Section 5.9).

11.5 Sizing the Gas Producer to the
Engine

A common problem among gasifiers is the use of an
oversized gasifier. An oversized gasifier produces ex-
cessive tars because lower flow rates do not develop
the high temperatures necessary for good tar destruc-
tion. An undersized gasifier has excessive pressure
drop, weak gas, and excessive raw gas temperature, and
may be prone to burning out the grate. (See Sections
5.7.3 and 5.7.4.)

Begin sizing by evaluating the actual horsepower
needed. For the task at hand, do not be misled by en-
gine specifications. Vehicle applications seldom use
full power output. A better indication of operating
horsepower is the vehicle’s measured fuel economy.

The average vehicle engine power required from a
gasifier may be figured from the gasoline mileage at
cruising speeds:
Power (hp) = (11-1)
Cruising speed (mph) x Conversion (hp-h/gal)

Gasoline mileage at cruising speed (mpg)
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For a 55-mph cruising speed and a fuel-conversion fac-
tor of 10 hp-h/gal (a typical value for spark engines),
the fuel consumption rates for various sizes of liquid-
fueled engines are shown in Table 11-1.

Gasifiers typically require about 10 kg (22 1b) of wood
or 5 kg (11 lb) of charcoal to replace one gallon of
gasoline. Thus, it is possible to calculate wood con-
sumption rates from the data in Table 11-1. In addition,
the maximum gas production from Imbert gasifiers is
determined by the maximum hearth load—typically
0.9 Nm3/h of gas is produced per square centimeter of
hearth area (see Section 5.8). In this way, the power re-
quirements given above determine the hearth size, as
shown in Table 11-2. See Table 5-1 for sizing of other
gasifier types.

Retaining the liquid-fuel carburetor in conjunction
with a gas mixer for dual-fuel operation realizes the
best advantages of both fuels, saving liquid fuel for
occasional full-power bursts while making economical
use of producer gas during normal operation. A gas
producer sized to provide cruising power as shown in
Table 11-2 should cover at least 90% of expected
driving conditions.

11.6 Engine Selection

Because engines are mass produced for vehicles, the
vehicle engine is the most likely candidate for build-
ing small systems. Larger systems can use natural-gas
engines, which in many ways are more suitable but also
are more expensive.

11.6.1 Large-Vehicle Engines — Truck
Engines up to 50 kW

The largest widely available spark-ignition vehicle en-
gines are Big Lung V8 460-in.3 engines. The Ford truck
engine of this size is used industrially for stationary ap-
plications and is highly regarded because its extra-
heavy block readily and repeatedly can be rebuilt to its
full original specifications. Lighter weight engines of
this displacement from passenger vehicles (e.g., Pon-
tiac and Buick engines) have lighter blocks. In addition
to a shorter service life for these lighter engines, the
rebuilding costs and overhaul parts can be more expen-
sive than for the heavy-block engine. Also, the

Table 11-1. Fuel Consumption of
Liquid-Fueled Engines

Fuel Economy Fuel Use Rate Engine Output

(mpg) (gal/h) (hp)2
5.5 10.0 100
11.0 5.0 50
22.0 25 25
55.0 1.0 10

aFor fuel consumption 10 hp-h/gal gasoline.

cylinder-wall thickness is inadequate to accommodate
cylinder-wall sleeves, limiting the number of times that
the light-block engines can be rebuilt.

11.6.2 Small Engines

Engines suitable for long-life operation (2000 to
20,000 h) must have an air cleaner, oil-pump pressure-
lubrication, an oil filter, and thermostatic engine-
temperature regulation. Smaller engines of the
air-cooled splash-lubrication types (e.g., many engines
manufactured by firms such as Wisconsin, Briggs,
Honda, Tecumseh, and Kohler) have very short service
periods between oil changes (one small-engine
mechanic recommends an oil change every 8 h of
operation). These engines are designed for total life
spans of around 1000 h (Onan 1981). Some of the above
manufacturers have certain models that do meet long
life requirements.

11.6.3 Natural-Gas Engines

Larger displacement spark-ignition engines (larger
than 460 in.%) are used for natural-gas-fueled station-
ary engines, many of which are used for irrigation ap-
plications. Like producer gas, natural gas has a low
flame-velocity and a relatively high octane rating.
These engines operate at a low rpm. The engines also
operate at a high compression ratio and have arelative-
ly high thermal efficiency. In addition, they have a very
long service life and are particularly suited to large
(greater than 250 kW) installations.

Natural-gas engines are built by Caterpillar, Waukesha,
Minneapolis Moline, and others. These engines are
available with standard options for industrial opera-
tion on gaseous fuel. An optional top-oiler is recom-
mended, which supplies oil mist with the fuel to
lubricate the piston rings. The top-oiler may greatly
reduce the wear experienced with dry-gas fuels, and
may be helpful with producer gas on smaller engines
as well.

Large, natural-gas, industrial engines exhibit a quan-
tum leap in equipment and overhaul costs over the cost
of vehicular engines. Service parts are often available
only through the manufacturer and may involve long
downtimes as well as high costs.

11.6.4 Diesel Engines

Diesel engines also are suitable for large installations
but have special problems, discussed in Section 11.10.

11.7 Cogeneration

Cogeneration involves using part of the greater than
70% of fuel energy that is otherwise lost as exhaust heat
and engine heat during engine operation. If one has a
use for this heat, then a water-cooled engine is both
convenient and safe. The hot air from air-cooled
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Table 11-2. Gasifier Size versus Engine Output

Wood Fuel? Hearth Capacity  Imbert Throat Diameter? Stratified Hearth Diameter®
Power P (hp)  Consumption (Ib/h) Q (kBtu/h) (cm) (in.) (cm) (in.)
100 200 1400 16 6.3 40.5 ' 16.0
50 100 700 11.3 4.5 29 114
25 50 350 8 3 205 8.0
10 20 140 5 2.0 13 5.1
5 10 70 3.6 1.5 9 3.6

a1 kg (2.2 Ib) of wood is required per hp-h for a 7000-Btu/lb average fuel heating value (Gengas 1950).

bimbert throat constriction diameter is given by D = 1.6 VP cm.

CA maximum stratified bed hearth load of hmax = 1 MBtu/h-ft2 is typical for wood chips. Stratified hearth diameter is given by D = 1.6 VP in.

engines requires safety precautions for heating because
it may contain dangerously high levels of carbon
monoxide so it should be vented. Heat from the exhaust
gas can be recovered with a compact water-cooled heat
exchanger or put to direct contact uses.

A particularly attractive application for the waste heat
from engines is the drying of biomass feedstock before
using it in the gasifier. The 50% moisture contained in
fresh wood chips can be removed effectively by drying
the chips with the engine’s exhaust gas. There is drying
capacity to spare if high-efficiency drying methods are
used.

Because the engine’s exhaust gas is hot enough (600°-
700°C) to decompose biomass, it must be tempered
(substantially diluted with cooler gases) before being
used. One of the best ways to dilute is to take the warm,
humid gas leaving the outlet of the biomass dryer, add
about 10% by volume of engine exhaust gas, and
recycle the mixture to the gas entry port of the dryer.
Reusing the outlet gas from the dryer instead of using
ambient air to temper engine exhaust gas offers several
advantages: (1) Because the gas coming from the dryer
is already warm, more of it must be used to dilute (and
cool) the engine exhaust gas to a given tempered
temperature. Therefore, more gas is available for drying
the feedstock; more gas increases the capacity of the
feedstock drying process. (2) The humidity in the dryer
flume gas reduces the chance for pyrolysis of the
feedstock when the gas recycles to the dryer. (3) Last-
ly, the gas from the dryer is essentially engine exhaust
gas; as such, it is nonoxidizing and tends to quench any
incipient fires in the dryer.

11.8 Spark-Ignition Engine Conversion
11.8.1 Engine System

A typical producer gas system for spark-ignition en-
gines is shown in Fig. 11-1. The system consists of a
gas producer (described in Chapter 5), a gas cleanup
and cooling system (described in Chapter 8), a starting
blower (described in Chapter 9), a carburetor (gas
mixer), and an engine. During operation, suction from
the engine draws air into the gas producer, through the

cleaning system, and into the gas mixer where air is
mixed with the gas.

11.8.2 Gas Mixers

Maximum power is achieved with a producer gas air
mixture just slightly lean of the stoichiometric combus-
tion ratio,* as shown in Fig. 11-2. On the other hand,
gasoline delivers maximum power with mixtures sig-
nificantly rich over its stoichiometric combustion
ratio.

A gas mixer behaves in some ways like a liquid-fuel
carburetor, but in reality it is much simpler. A car-
buretor must mix the correct weight of air with the lig-
uid fuel (normally a air-fuel ratio of about 15 for
gasoline and about 6.5 for methanol). When the throt-
tle is opened suddenly, this mixture is momentarily en-
riched over the normal idling mixture by the
accelerator pump, which supplies a squirt of fuel. This
mixture is additionally enriched at high loads by the
power circuit in order to boost power and to protect the
engine from excessive temperatures. The mixture is en-
riched for starting by the choke when the engine is cold.

The producer gas mixer must mix a proper ratio of air
with the producer gas, approximately a 1:1 ratio of fuel
to air by volume. Variations in the producer gas mix-
ture cause sharper changes in engine power than do
similar variations in a gasoline mixture, as shown in
Fig. 11-2. For this reason, the gas mixture must be con-
trolled by the operator during producer gas operation.
The gas quality can vary during operation and needs
periodic adjustment; the gas mixer must allow for in-
dependent control of the gas, air, and mixed-gas flow
to accommodate such adjustment.

A number of mixers used in the past are shown in
Figs. 11-3 and 11-4. Note that, just like a carburetor,
each mixer has one control (the throttle) to meter the
total flow and another control (the choke) to adjust the
ratio of gas to air.

*The stoichiometric combustion ratio is that fuel-air mixture that al-
lows the gas to burn completely, with no surplus air remaining after
the fuel supply has been exhausted.
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Fig. 11-1. Producer gas engine for spark ignitiori system (Source: Skov 1974, Fig. 46. © 1974. Used with permission of Biomass Energy Foundation,

Inc.)

The butterfly valve often used in throttles and chokes
has been found to respond to adjustment very
nonlinearly and, therefore, can be a troublesome
method for controlling the gas-air mixture. A gate valve
provides much finer control of the air inlet than does
a butterfly valve, so the narrow power peak is
broadened out over one-half to one turn of the gate
valve. Conversely, the entire range from too rich to too
lean occupies only a few degrees of arc on a butterfly
valve and is easily missed altogether. However, a
butterfly valve can be used satisfactorily for throttle
and gas-inlet control.

If it is essential to prevent gas from being released into
the engine compartment after shutdown, then an addi-
tional gate valve should be used for positive gas shutoff.
If idling is desired, a metering idle valve leading from
the gas inlet around the throttle may be used to enrich
the mixture during idle, without changing the running
mixture, as shown in Fig, 11-4.

11.8.3 Power Time Lag

Lean gas-air ratios are normally produced for a short
time after the throttle is opened, and richer gas is
produced just after the throttle is closed. This change
in gas quality alters the gas mixture and weakens en-
gine output to a point at which the engine may even
stall. The magnitude of this momentary power loss can

be reduced by adjusting the fuel-air ratio to maintain a
peak-power mixture, but this requires operator ex-
perience. One improvement that today’s technology
might bring to gasifier designs is automatic mixture
control, employing a feedback signal from the engine
exhaust. This type of control is already in use on feed-
back fuel-injection systems, and it should be adaptable
to producer gas systems.

Gas-fueled buses, which necessarily make frequent
stops, overcome the fuel-air problem with a bag in-
flated with gas by a blower, which maintains gas flow
during stops. The surge of power needed for accelera-
tion is then provided mostly by gas from the bag.
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Fig. 11-2. Power versus gas mixture (Source: Kaupp 1984a, Fig. 198)
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Fig. 11-3. Gas air mixers (Source: Adapted from Cash 1942 and Anon 1943)

Driving through hilly country in a producer-gas-fueled
vehicle requires special skills. Often the driver races
the engine while going downhill in order to build up
the gas quality, so that high-quality gas is available
when climbing is resumed. Traffic accidents increased
in Sweden during World War Il because drivers of
producer-gas-fueled vehicles operated those vehicles
at high speeds while running at hills (Gengas 1950).

11.8.4 Engine Startup

If liquid fuel is available, the engine can be started on
fossil fuel (gasoline or propane), and the gasifier started
separately with a fan or blower (which may be hand-
or battery-operated) while flaring the gas. The blower
outlet and flare are located upstream from the cleanup
system to keep tars out of the cleanup stream during

starting procedures. When the gas is less tarry and can
support combustion on its own, the blower is turned
off, and the gas valve is opened. Then, the producer gas
air/throttle is slowly opened; simultaneously, the gas-
air mixture is adjusted to obtain maximum power out-
put. At the same time, the fossil fuel throttle is slowly
closed until the changeover is complete. Once the mix-
ture has been initially adjusted, repeated changeovers
can be made quickly. The liquid fuel supply should
then be securely closed to prevent liquid-fuel con-
sumption. This type of dual-fuel capability permits
rapid changeover between solid and liquid fuels.

One novel and simple alternative to a rotary blower for
startup is to use the engine exhaust gas as the propel-
ling gas for an aspirator ejector, as shown in Fig. 11-5.
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This technique allows a considerable saving in battery
power at the expense of only a little extra liquid fuel.

11.8.5 Ignition Timing

Producer gas burns slowly, as shown in Fig. 11-6,
giving it a high octane rating. It is usually found that
optimum engine operation is obtained by advancing
the ignition timing 5 to 15 degrees more than the
advance used with gasoline, as shown in Fig. 11-7.

The “ping-sensor” and “electronic ignition” timing
controls used in many late-model automobiles may be
helpful in automatically providing the ideal spark
advance for producer gas, while permitting dual-fuel
operation.

11.8.6 Spark Plugs

The porcelain insulator of a spark plug may be glazed
with feldspar or quartz for generator gas operation, to
permit easier deposit removal. Spark plugs for gasoline
engines are normally not glazed because lead anti-
knock compounds form a conductive glass with the
glaze. Therefore, lead-free gas should always be used
for start-up procedures with glazed plugs. The heat
range of the spark plug should be high enough to per-
mit the plugs to self-clean, but not so high as to cause
preignition. If fouling is encountered, use a lower heat
number (hotter electrode).

11.9 Two-Cycle Engine Conversion

Two-cycle engines present a special challenge for con-
version to producer gas. The crankcase must be lubri-
cated and protected from contact with gas impurities.
A number of the most novel innovations from the en-
tire World War II era appeared on the two-stroke
engines of that time.

In the Bolinder Munktell system shown in Fig. 11-8,
the lower crankcase served as an air pump to supply
air to pressurize the gasifier and supply pressurized air
to the gas mixer. In the two-cycle conversion, either the
head must be fitted with both intake and exhaust
valves, or the piston-porting system must be isolated
from the crankcase to prevent gas deposits from
interfering with lubrication.

Similarly, the Pulsator prevents producer gas from
passing through the crankcase. A side-chamber is
provided, as shown in Fig. 11-9, and connected to the
crankcase through a butterfly valve and to the gas sup-
ply and intake manifold through check valves. The pul-
sating movement of air in the crankcase alternately
sucks in and mixes air and gas, and then delivers the
mixture to the combustion chamber. A check valve
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Fig. 11-5. Exhaust pipe ejector for start-up fanning (Source: Gengas
1950, Fig. 151)
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leading from the air control into the crankcase permits
air flow through the engine. The volume of the pulsator
mixing chamber necessarily prevents gas from being
sucked all the way into the crankcase.

The third method of two-cycle conversion, the two-
cycle crank case system shown in Fig. 11-10 is the
simplest. Mixed gas is drawn through check valves into
the crankcase, where the gas enters the combustion
chamber through the piston ports. Engines popular for
use on today’s motorcycles use this same principle.

11.10 Diesel Engine Conversion

11.10.1 Diesel Operation with Producer Gas

Diesel engines are operated on producer gas in the
“aspirated” mode. The producer gas is mixed with the
intake air stream. A small quantity of diesel fuel is in-
troduced continuously through regular injectors to ig-
nite the gas mixture and provide timing. This so-called
pilot-injection mode or diesel dual-fuel mode is also
used occasionally with natural gas or alcohol fuels.
Direct-injection diesel engines are better suited for con-
version to producer gas than antechamber diesel
engines.
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The power lost when diesel engines are converted to
operate on producer gas is less than that lost by spark-
ignition engines for three reasons. First, some diesel
fuel is introduced; second, diesel engines operate at a
higher compression ratio; and third, diesel engines
operate with a large excess of air, so the energy per
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Fig. 11-8. Bolinder Munktell two-stroke engine (Source: Gengas
1950, Fig. 191)
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stroke is very close to the heating value of producer gas-
air mixtures. Furthermore, full diesel operation can be
achieved readily by simply increasing the quantity of
diesel fuel to normal levels when needed. This can be
accomplished easily by using the existing governor on
many diesel engines. The brake thermal efficiency is
higher for dual-fuel operation than it is for straight
diesel fuel operation, as shown in Fig. 11-11.

Standard diesel engines inject the liquid fuel into the
compressed-air charge in order to start combustion at
a time preset relative to the crank position. When using
producer gas, it is necessary to advance the injection
timing angle with engine speed especially at high rpm.
The optimum injection timing for one engine is shown
in Fig. 11-12. An engine with an improperly set timing
angle will run roughly at low speed and expel smoky
exhaust at high speeds.

It has been reported that when operating conditions al-
lowed producer gas-air mixtures to self-ignite without

engine knock, a single-cylinder, direct-injection, slow-
speed (1000-1500 rpm) diesel engine was able to run
on 100% producer gas for extended periods (Cruz
1984).

Ignition reliability is better with pilot injection than
with spark ignition. The minimum injector quantity for
stable pilot ignition and efficient combustion is given
by a diesel fuel-mixed gas ratio of 10 mg/kg. Increasing
this ratio will boost power output, up to a ratio of
20 mg/kg; above this injection rate, thermal efficiency
falls with no significant power increase. Partial load
operation is complicated by the fact that a full inlet
charge is required to develop ignition temperature.
Throttling for no-load operation must lean out the gas
mixture and reduce the injection quantity to as low as
5 mg/kg (Anon 1944).

Diesel engines also can be converted completely to
producer gas by reducing the compression ratio, add-
ing a spark-ignition system, and replacing the injectors
with spark plugs. The modification expense is hardly
justified, considering the successes of pilot injection,
except in situations where complete replacement of
diesel fuel is required. The compression ratio should
be reduced to 9.5-10:1. For antechamber diesel engines,
the heads must be replaced.

Pilot injection helps the producer gas-air mixture to ig-
nite. Very stable and reliable ignition has been ob-
served using pilot injection; in fact, it has been shown
to perform better than spark ignition. The power out-
put from weak gas (e.g., 118 Btu/scf) can be 90% of that
from straight diesel fuel with 40% excess air (Anon
1944). Minimum pilot injection is limited by its ability
to provide stable ignition at roughly 10% to 20% of the
regular idle injection.

11.10.2 Starting Diesel Engines

Diesel engines are started by first igniting and fanning
the gasifier with the blower until clean, burnable gas is
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Fig. 11-10. Two-cycle crankcase system (Source: Gengas 1950, Fig. 185)
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produced. The engine itself is started and warmed up
on diesel fuel. As the producer gas-air mixture is ad-
mitted, the governor automatically reduces the liquid-
fuel feed. Fuel savings up to 90% over straight diesel
operation have been reported (Kjellstrom 1983). The
governor can automatically boost the injection rate in
the event of an increased power demand or a variation
in the quality of the producer gas.

11.10.3 Throttling at Partial Load

High compression is essential for ignition, so throttling
should be applied only at the gas inlet. Throttling at the
air inlet results in excessive suction, which can cause
bad combustion and smoky exhaust. However, at
medium-to-high loads, where the gas-inlet throttle is
fully open, the air inlet throttle may be partially closed
to increase gas suction.

11.11 Increasing Power from
Producer-Gas-Fueled Engines

11.11.1 Mechanisms of Power Loss

A given volume of producer gas, when mixed with the
correct quantity of air for combustion, contains 70% of
the energy of an equal volume of gasoline or propane
mixed with air. The relative power at any given low
rpm is proportional to the mixed gas energy, as shown
in Fig. 11-13. Full-throttle power increases in direct
proportion to the rpm rate, up to the point where pres-
sure drops in the intake and exhaust valves limit the
power. Beyond that point, power drops off rapidly with
an increasing rpm rate.

The volume of intake gases for producer gas use is
roughly doubled over gasoline or propane use, and in-
creasing pressure drops reduce the peak-power rpm by
up to 30%. The combined effects of a reduced mixed-
gas heating value and a reduced peak-power rpm result
in an overall reduction in the maximum engine power
up to 50%.

11.11.2 Engine Breathing

A producer gas engine must handle roughly twice the
volume of intake gases as gasoline engines. The
engine’s breathing ability therefore begins to limit en-
gine power at lower rpm. The peak power output,
shown in Fig. 11-14, operating on producer gas, occurs
at roughly two-thirds of the rpm for gasoline fuel. Im-
provements in the intake induction system can both
raise the speed attained during the peak-power output
and decrease the overall power loss at each rpm rate.

One might ask how producer gas containing
150 Btu/scf can produce more than 150/1000ths of the
power of methane containing 1000 Btu/scf. The reason
is that engine power depends on the mixed-gas/air
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heating value, not on the pure-gas heating value. The
mixed-gas heating values are shown in Fig. 11-15.

11.11.3 Efficiency and Power Loss

The spark engine operating on gasoline achieves a ther-
mal efficiency of 25%-30%. The same engine operat-
ing on producer gas may achieve 15%-25% thermal
efficiency, depending on how well the engine is con-
verted to producer gas. A diesel engine using diesel
typically achieves 30%-35% thermal efficiency.
Operating on 90% producer gas, it can be expected to
give 25%-30% thermal efficiency. The overall efficien-
cy of the system must be computed from engine
efficiency and gasifier efficiency.

Engine friction losses are primarily a function of engine
speed. Partial-throttle operation increases the fraction
of fuel consumption devoted to fixed engine losses.
The highest shaft efficiency for any speed occurs with
the highest power consistent with complete combus-
tion. Therefore, it is primarily for the benefit of fuel
economy that various techniques are used to increase
the power output obtained from producer gas.

11.11.4 Blowers and Superchargers

The power from producer gas can be increased to that
available from naturally aspirated gasoline or propane
{or even higher) in several ways.

A Roots-type blower or turbosupercharger can be used
to increase the pressure in the cylinders above atmo-
spheric pressure, and thus to increase the air-fuel
charge. A positive pressure of 6 psig at the intake
manifold can recover full gasoline power.
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The Roots blower uses engine shaft power for compres-
sion and necessarily reduces engine efficiency. The tur-
bosupercharger uses the pressure of the exhaust gases
to operate a turbine compressor, thereby recovering
most of this lost energy.

Since engines are designed to withstand a particular
combustion pressure (BMEP, or brake-mean-engine-
pressure), power increases must be limited to restoring
the original peak-combustion pressure.
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Fig. 11-14. Power output as a function of engine speed (Source:
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11.11.5 Other Methods for Increasing
Producer Gas Power

A portion of the power loss can be recovered even for
anaturally aspirated engine by increasing the compres-
sion ratio, as shown in Fig. 11-16, advancing the spark,
and improving the intake systems. Producer gas has a
higher octane rating than gasoline, so the power can be
increased by increasing the compression ratio to be-
tween 11:1 and 14:1. Increasing the compression ratio,
however, may prevent dual-fuel operation on lower
octane fuels because of excessive knocking and detona-
tion. One method for adapting high-compression-ratio-
modified engines to operate at a partial load on lower
octane fuels is to recycle some exhaust gas back to the
intake manifold. The mixture of gasoline, air, and ex-
haust gas burns more slowly, minimizing knock and
permitting partial power operation.

Cam shaft design may be optimized for producer gas by
faster valve opening, higher lift, and no overlap, with
valve timing-duration optimized for producer gas.
Synchronous generators require engine speeds of 1800,
1200, or 3600 rpm. Cam shaft design has been
overlooked so far in the lower rpm region.

Power levels also have been increased to near-super-
charged performance by improving the intake system
through installing larger inlet valves, or especially by
using tuned ram induction and larger intake manifolds.
Additionally, intake manifolds may be improved with
flow straighteners after each bend, shell-casting tech-
niques to give smoother inlet surfaces, and carefully
tuned long ram intake manifolds to give a sharper peak
pressure at the optimum rpm rate.

The decrease in power can be made more acceptable if
one increases the engine speed by lowering the final
drive-gear ratio. Full power cannot be recovered com-
pletely by increasing the rpm rate alone. Many inter-
esting details for converting piston engines to operate

on generator gas during World War II are given in
Generator Gas (Gengas 1950]}.

Further work in engine adaptation would be beneficial.
In particular, combustion chamber design for improved
swirl and squish can lead to better combustion efficien-
cies, and microprocessor controls using signals from
pressure, temperature, ping, and oxygen sensors can be
applied to provide continuously optimized, unat-
tended engine operation, to follow changing gas quality
and variations in solid fuels.

11.12 Engine Life and Engine Wear
11.12.1 Engine Life Expectancy

There are reports from Sweden during the late World
War Il era (undocumented) and also from recent inves-
tigations (Kjellstrom 1981), of longer engine life ex-
perienced with producer-gas-fueled engines than with
gasoline-fueled engines. Note in Table 8-3 that cylinder
wear is less on producer gas with fabric filter than wear
on straight diesel fuel for two out of three tractors
tested, and that oil contaminants were also less on well-
cleaned producer gas than on diesel oil.

Discussion with rebuilders of industrial engines indi-
cates that life expectancy for fossil industrial engines
is related to engine speed. High-speed, 3600-rpm
operation is discouraged because it may cause prema-
ture failure. Lower speeds give commensurately longer
engine life expectancies (e.g., engines operated at
1800 rpm may last 1-1/2 years, 1200-rpm-operation
may allow 3-year lifetimes, and 900-rpm-operation

70 &
e é
o)
60 : 5
< . / ! &
§ 4
o 50 _0\\3\ '
x > S
4] (V
2 e‘a\c')'
Q
C as //
o O/r ‘
17 2%
E /\N‘Ooé
£
o
@]
10
0
4 5 6 7 8 9 10 11
Compression ratio (£)

Fig. 11-16. Combustion pressure versus compression ratio for various
fuels (Source: Gengas 1950, Fig. 174)
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may allow 5 years of engine life). No known property
of clean producer gas should shorten engine life. One
truck, which has been driven 100,000 km on producer
gas without any unusual maintenance problems, is
shown in Fig. 11-17.

Many of the clean-up mechanisms outlined in Chapter
8 have yet to be applied to producer gas cleanup, and
the gas in the bulk of today’s test and production sys-
tems is far from clean enough for reliable operation,
much less long-life operation. We will discuss the ef-
fects of dirty gas on engine life in the following
sections.

11.12.2 Sticking Intake Valves

Tar mists and dusts form an asphalt-like deposit
around the intake valve stems. Valve-sticking is more
a problem of accumulated deposits than of engine wear.
Sluggish valve closing decreases power until cylinder
compression is lost. The valves may be reconditioned
and cleaned, and the engine returned to service
without serious damage to the rings and bearings.
Intake valve sticking can be corrected without a
complete engine overhaul. However, sticking valves
clearly indicate the need for a more efficient cleanup
system (Chapter 8) and possibly redesign of the gasifier
for a higher hearth load to reduce tar production. (See
Chapter 5 for information on gasifier sizing and
design.)

Based on cylinder, bearing, and cam wear, a life-expec-
tancy of 5000 hours was estimated for an engine after
extended tests with gas so dirty that the valves required
cleaning eight times in the first 1000 hours (Breag
1982).

11.12.3 Oil Thickening and Contamination

In gasoline engines, gasoline vapors from blowby tend
to thin oil with time. With clean producer gas, the
volatile motor oil fractions evaporate, causing the oil

Fig. 11-17. Swedish logging truck that operated 100,000 km on
producer gas without any unusual maintenance (Source: NAS 1983,
p. 46)
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lubricating ability, but it also increases cranking
requirements.

Dirty gas can introduce tar particulates and corrosives
into the oil through cylinder blowby. Particulates in the
oil, and ash and char accumulation in crankcase oil can
increase engine wear. Particles larger than the oil-film
thickness may scratch bearing surfaces; however, char
and ash tend to be easily crushed to smaller particles,
minimizing this problem. If ash slagging occurs, harder
abrasive particles are produced that could cause severe
wear. Small amounts of sodium from ash in either the
lubricating oil or the fuel gas can lead to engine valve
corrosion. It becomes obvious that particulate removal
is very important to engine life at many levels.

11.12.4 Tar/Qil Accumulations

Tar/oil mists occur as very finely divided droplets
much smaller than char and ash. Therefore, they are
much more difficult to remove from the gas and are
more likely to cause engine problems. Tar accumula-
tion does not cause wear; rather, it causes moving parts
to stick, and plugs passages, intake valves, throttle
valves, and piston rings. Tar deposits become a viscous
fluid at engine operating temperature; on cooling and
drying, they can become a hard varnish that will
prevent or hinder engine startup, or bend a pushrod in
the event of full valve seizure. The best cure is preven-
tion through the cleaning methods of Chapter 8 and
verification of gas cleanliness regularly, using the tests
of Chapter 7.

Mild valve-sticking can sometimes be reduced or
prevented by operating the engine on liquid fuels
(especially methanol) for a short time before shutdown.
This tends to rinse tar accumulations into the
crankcase oil, where they do less damage. This tech-
nique may merely prolong the time lag between repair,
and delay use of proper cleanup techniques.

Oil should be changed when it is visibly dirty or exces-
sively thick. Crankcase oil analyses are more widely
available today than in the past and can help diagnose
causes of engine wear.

11.12.5 Engine Corrosion

Corrosive engine cyclinder wear increases with low
operating temperatures as shown in Fig. 11-18. Below
120°F, bearing corrosion, cylinder wear, and ring wear
rise rapidly with increased water condensation and
corrosive hydrolysis products due to carbonic acid.
Below 185°F, cylinder and ring wear depends on oil
consumption and oil-film thickness. Above 185°F,
wear is minimal and independent of oil-film thickness
(Mahin, June 1983). Wear is maximum after a
cold-engine startup; preheating the engine block to
120°F before startup can minimize this wear for
maximum engine life.
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11.12.6 Engine Warranty

Reputable sellers of gasifier engine systems should
offer an engine warranty, backed up with service con-
tracts, operator and maintenance training, and con-
tinuous monitoring of the performance of installed
units. In this way the maker will be the first to know of
the need for design improvements. Minneapolis
Moline has written a 6000-h warranty on a producer-
gas-powered engine (Mahin, June 1983).

11.13 Exhaust Emissions

Carbon monoxide exhaust emissions from a properly
running producer-gas-fueled engine can be expected to
be inherently lower than emissions from gasoline-
fueled engines because maximum power is achieved
from a mixture that is lean of the stoichiometric com-
bustion mixture, whereas the maximum power from
gasoline is achieved by burning a rich mixture.
Hydrocarbon emissions also can be expected to be
lower than those from gasoline because of the relative
absence of hydrocarbons in the producer gas. The
lower flame temperature of the producer gas, along
with the excellent antiknock characteristics and low
prompt nitrogen content of producer gas, suggests that
NO, emissions should be lower as well.

The major pollutant source from producer gas is the
necessary disposal of cleaning condensate, which may
be high in tars, especially phenols. This source can be
minimized by using the driest fuels possible and using
the best low tar gasifier design available to prevent tar
and condensate production. If the fuel is dry enough to
render condensate removal unnecessary, then the gas
can be used above its dew point, eliminating conden-
sate collection altogether. This scheme has been used
in a few recent designs. Another approach to conden-
sate removal is evaporation, coupled with recycling the
residue to the gasifier. Still another approach applied
to an especially tarry system is incineration of the con-
densate tars outside of the gas producer. However, this
approach consumes additional fuel. Prevention

remains the most attractive cure. (See the discussion
on low tar designs and drying in Chapter 5.)

11.14 Other Devices for Producer-Gas
Power Generation

11.14.1 Gas Turbines

Historically, producer gas has been used in internal-
combustion piston engines of both the spark-ignition
and diesel-powered types. Accordingly, these engines
remain the major area of interest at present. However,
the gas turbine may be an attractive application for
producer gas for electric generation, since it operates at
a relatively high efficiency (25%-35%) in intermediate
sized units (200-10,000 kW). Gas turbines can be
started and stopped in a short time, so they are
especially useful for producing peaking power.

Gas turbines were developed after producer gas use
was abandoned, and turbines still have not been
operated on producer gas. Producer gas could eventual-
ly become a major turbine fuel for electric power
generation. Turbines offer simplicity, long life, and
reliability. The simplicity of one moving part is un-
matched. Long bearing life with minimal wear is
achieved through an even, nonimpulsive bearing load
rotating at high speeds to establish a stable and
continuous lubricant film.

Turbines now in use operate with intake gas pressures
of 75 to 150 psi. Converting existing turbines to
producer-gas use would require operating the gas
producer at pressures up to 150 psi in order to avoid
compressing the gas for the turbine combustor.

Gas must be exceptionally clean and particularly free
from alkali metal content for turbine operation, be-
cause the blades operate at high temperatures and
velocities, and are easily eroded and coated. Based on
tests of a 4250-hp turbine fired with powdered coal,
erosion and blade deposits were eliminated using high-
temperature cyclonic inertial cleanup, which captured
99% of 20-um and 92% of 10-um particles (Yellott
1955). The Aerospace Corporation has developed a
wood-fired turbine system operating at 5 MW under a
DOE contract. Initial results show that ash from the
wood, though relatively bulky, has not presented
problems that were expected (PNL 1985).

11.14.2 Fuel Cells

Another potentially interesting use of producer gas is
to generate electric power using fuel cells. A fuel cell
is an electrochemical device that converts chemical
energy into electricity directly without moving parts.
Fuel cell development is very active at present, and
some fuel cells may soon be available for use with
producer gas.
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11.14.3 External-Combustion Devices

A variety of external-combustion devices can produce
power from producer gas. The Stirling-cycle engine
uses an external heat exchanger and external combus-
tion. The main drawback is the lack of widespread
availability of inexpensive Stirling-cycle engines. The

steam engine also uses external combustion in a boiler
to produce high-pressure steam for use in piston or
turbine steam engines.

When clean producer gas is available, external-
combustion devices are not necessary.
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Chapter 12
Safety and Environmental Considerations

12.1 Introduction

During the emergency development of gasifiers during
World War II, various dangers were discovered in con-
junction with gasifier operation. These dangerous areas
were divided into toxic hazards, and fire and skin burn
hazards. In addition, we have since recognized the
potentially damaging effects that our activities can
have on the environment, and we shall point out areas
that may be affected by widespread gasifier use.

12.2 Toxic Hazards
12.2.1 Carbon Monoxide

The principal fuel component of generator gas is car-
bon monoxide (CO), a deadly poison that ties up
hemoglobin in the blood and prevents the transport of
oxygen to the tissues. Death from CO is death by suf-
focation. Lesser exposures cause headaches, nausea,
dizziness, and irritability. CO is an insidious poison
because it is odorless and tasteless.

Exposure during pregnancy, even at levels too low to
show symptoms in the mother, may affect development
of the fetus, lower its birth weight, and increase the risk
of abortion and stillbirth. There are no indications that
CO causes mutations or cancer (Kjellstrom 1981).

Seventeen people were killed in Sweden between
December 1939 and March 1941 because of careless
gasifier operation (Foley 1983). More recently, two re-
searchers at a Midwestern university died from CO in-
halation when they climbed inside of a gasifier fuel bin.

CO occurs widely in our industrial civilization in small
quantities. Smokers typically inhale concentrations of
several thousand parts per million, and some smokers
can have as high as 20% hemoglobin blood saturation
(Kjellstrom 1981). Gasoline-fueled automobile ex-
hausts used to contain as much as 5% CO, and after a
decade of improved pollution controls on cars, CO is
still a major contributor to pollution in our cities.
However, until the advent of natural-gas pipelines, CO
was the primary fuel component of manufactured city
gas, coal, and blue water gas. These were widely dis-
tributed and used around the world and are still used
in many countries, evidence that CO can be handled
safely when proper procedures are followed.

12.2.1.1 Acute Carbon Monoxide Poisoning

The symptoms of carbon monoxide poisoning are
shown in Table 12-1.
First aid treatments for CO poisoning follow.

* Move the poisoned person to the open air or a room
free of CO. Prevent the victim from exerting himself.

Table 12-1. Symptoms of Carbon Monoxide Poisoning

% Saturation of the Blood Symptoms
with Carbon Monoxide At Rest During Physical Exertion

0-10 None None

10-20 None During exertion, dizziness, heart
pounding, and difficulty in breathing
may occur.

20-30 Headache may occur. In case of exertion, pressure at the
forehead. Mild headache.

30-40 Headache in the forehead or back of the head, In case of exertion, dizziness, fainting,

pulse increase, heartbeat, nausea.

possibly unconsciousness are added.

40-50 All symptoms more pronounced, nausea, vomiting,
dizziness,increased tendency for unconsciousness.

50-60 Deep unconsciousness with increased breathing

and pulse rate.

60-70 Deep unconsciousness with slow pulse and low

breathing rate; possible death.

70-80 Respiratory failure and death.

Source: Gengas 1950.
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Loosen clothing around the neck and throat, and
check for the presence of an adequate airway in the
throat.

o If the person is unconscious, administer artificial
respiration immediately.

» Administer oxygen or mixed resuscitation gas (7%
CO, in oxygen) as soon as possible; see Fig. 12-1.

¢ Summon medical aid but not at the expense of leav-
ing the victim unattended.

¢ Keep the victim warm.

e Keep the victim under surveillance, as relapses often
OCCUT.

Exertion presents two profoundly dangerous effects on
CO poisoning victims. First, increased exertion in-
creases the speed of CO absorption into the blood
stream, as shown in Fig. 12-2. Second, physical exer-
tion increases the impact of CO that has already been
absorbed, as shown in Table 12-1. Removing a person
suffering from CO poisoning to fresh air should be done
as quickly as possible, but preferably without the
victim’s active participation. If possible, the person
should be carried. Over-exertion on the part of the
rescuers should also be avoided.

Every second is valuable, and the difference may have
a life-saving effect. Any producer gas installation
should include two bottles of mixed resuscitation (Kar-
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Fig. 12-1. Treatment of carbon monoxide poisoning (Source: Gengas
1950, Fig. 265)

bogen) gas (O, + 7% CO,) or pure oxygen, with two face
masks and shoulder harnesses to free the hands of the
rescuers. The Karbogen gas removes CO as much as
three times as fast as pure oxygen alone since the CO,
in Karbogen gas stimulates the vagus nerve, causing
more rapid breathing and, hence, faster CO removal.

Chronic CO poisoning symptoms, which often “sneak
up” on the victim, cause him to become tired, uncom-
fortable, and irritable, and also induce sleeping dif-
ficulty. Sex drive virtually disappears, and urinary and
heart problems have been common. Memory and
eyesight may be temporarily impaired. More extensive
mental symptoms may occur, including impaired
memory, reduced concentration and perseverance, and
possibly brain damage. The effects of chronic CO
poisoning may go undiagnosed or be attributed to other
causes by the uninformed victim.

12.2.1.2 Carbon Monoxide Exposure Standards

It is curious that no safe threshold CO exposure limit
was recognized during the development and use of
Swedish World War II-era gasifiers. Such a threshold
limit would recognize CO levels below which one
could expect to be free from the effects of chronic CO
poisoning. The current threshold limit value (TLV) in
the United States is 50 ppm CO in the work place time
weighted average for an 8-h work day. Short-term ex-
posure limit (STEL) is 400 ppm (MSA and ACGIH
[OSHA 1982]). The international standards are shown
in Table 12-2. Atmospheric concentrations above 1%
(10,000 ppm) CO may cause unconsciousness in only
a few minutes.

12.2.1.3 Safe Operating Procedures

From a safety standpoint, the best gasifier systems
operate at negative pressure (suction gasifiers), so that
leaks result in air being drawn into the system (possibly
causing gas explosions, which the equipment should
be designed to handle without harm) rather than CO
being expelled into the workshop. If the gasifier is
pressurized or if there is an outlet blower, then the gas
will be under positive pressure and leaks will release
CO. The importance of achieving a leak-tight system
cannot be overemphasized.

The entire system should be leak-tested upon installa-
tion and regularly with engine service thereafter. The
pipes carrying producer gas should be provided with
fittings to close off and pressurize the system to 4 in.
water gauge, and the entire system should be checked
with soap bubbles, especially at fittings, valves, welds,
lids, and seals.

Where possible, generators should be used outside,
with adequate ventilation. Indoor installations should
provide adequate ventilation that effectively changes
the inside air supply every 1 to 2 minutes. Alarms are
available for measuring and signaling excessive CO
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Fig. 12-2. Absorption of carbon monoxide in the blood (Source: Gengas 1950, Fig. 264)

levels in ambient air. Mini (Peeper) CO models [, I, and
III (MSA catalog nos. 465398, 465530, and 466523,
with ranges of 0-100, 0-500, and 0-250 ppm, respective-
ly) indicate CO levels, while model [V (MSA catalog
no. 468572} indicates the CO level and sounds an alarm
(MSA). These precautions should be observed and en-
forced for the protection of all who run the risk of
inhaling gas.

12.2.1.4 Startup and Shutdown

Carbon monoxide gas release also occurs during start-
up and shutdown. Gas is released when the gasifier is
fanned or when the engine is started on producer gas.
These two CO releases can be eliminated by installing
a propane-enriched burner at the fan outlet and by start-
ing the engine on liquid fuel.

When a gas generator is shut off, CO continues to be
evolved, sometimes for hours. In this case, it is par-
ticularly important to have adequate ventilation. Since
suction is no longer present within the gasifier, pres-
sure builds up, and CO and smoke can leak out. Under
no circumstances should a generator mounted on a
vehicle be driven into a closed garage and shut off.

These concerns were the impetus for vehicle gasifier
regulations between 1939 and 1946 that required the
gas-mixer air inlet to be extended out of the engine
compartment. In this way, the opening was a safe dis-
tance downwind from doors and windows. Similar
regulations applied to the gasifier air inlet.

12.2.2 Creosote

Although downdraft gasifiers usually convert less than
0.1% of the input into tars and oils, these heavy
products still must be scrubbed from the gas and dis-
posed of. In earlier days, they were probably flushed
down the sewer or buried. Today, such practices have
come under close scrutiny, and neither can nor should
be tolerated.

Man has always lived with the smoke and tars from
fires, and we know that these materials can be tolerated
in reasonable quantities. We also know now that the
smoke from wood cooking can cause cataracts and that
some tars contain dangerous carcinogens, so they
should be handled with care.

Furthermore, the tars contain phenols that are potent
bacteriocides. Relatively small amounts can interfere
with the proper operation of septic tanks and
municipal sewage systems. In Sweden during World
War II, the maximum permissible phenol content of
water released to sewers was 10 g/m® (10 mg/L), ap-
proximately 10 ppm. The phenol content of typical
gasifier or gas-cooler condensate is from 1500 to
3000 mg/L. Dumping these condensates onto the
ground or into the sewers or waterways is not accept-
able. It should be mandatory that we determine safe
disposal for these materials.

The best way to solve this disposal problem is preven-
tion at the source; in other words, gasifiers must be
designed to convert the maximum amount of tars to gas.
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Table 12-2. Maximum Carbon Monoxide Concentrations in Work Places — International Standards

Fraction of

Country mg/m3 ppm Year Hemoglobin (%)
Australia 55 50 1973 8
Belgium 55 50 1974 8
Bulgaria 20 17 1971 3
BRD 55 50 1978 8
Czechoslovakia 30 26 1976 45
Denmark 40 35 1978 5.5
DDR 55 50 1978 8
Finland 55 50 1972 8
Hungary 30 26 1974 4.5
ltaly 55 50 1975 8
Japan 55 50 1975 8
Netherlands 55 50 1973 8
Norway 40 35 1978 55
Poland 30 26 1976 45
Romania 30 26 1976 45
Soviet Union 20 17 1977 3
Sweden 40 35 1978 55
USA 55 50 1979 8
Yugoslavia 58 51 1971 8

Source: Kjellstrom 1981

(See Section 5.9.) Some gasifiers now incorporate
catalysts that claim to eliminate the problem entirely
(Ekstrom 1985). Others employ a high degree of tar
recycle to eventually crack the tar thermally (Susanto
1983; Groeneveld 1980b; Kaupp 1984a).

Other methods for disposing of these materials depend
on the method of gas cleanup. In some cases, wood
chips or burnable filter materials are used to filter the
tars. The chips used in this type of process can be dried
and used as a fuel; if the tar is collected on solid filters,
the filters can be incinerated. If the tars are collected in
water, waste heat can be used to concentrate the tars
and they can then be gasified with the gasifier fuel.
Well-dried biomass fuel minimizes condensate
production and eliminates the need for condensate
removal for the purpose of improving gas quality. In
this case, the gas can be used above its dew point, thus
largely eliminating condensate collection.

Finally, it should be noted that there are neither ac-
counts nor any evidence of damage occurring from
draining the nearly one million vehicle gasifiers that
were in operation worldwide between 1939 and 1945
(Kjellstrom 1983).

12.3 Fire Hazards

In the early days of gas generators, there were a large
number of garage fires. These hazards were reduced
through education and regulation to discourage indoor
gasifier filling. Gases inside the hopper can flash when
the lid is opened during filling, which in turn can ignite
nearby flammable materials.

Flammable and explosive gas mixtures are usually
present inside a cold gasifier. When a flame is intro-
duced to start the gasifier, care should be exercised to
prevent explosions. Proper precautions include the
following:

e Always fan a cold gasifier before igniting it, to
remove residual producer gas; in this way, one
ensures that the gasifier contains only fresh air.

e Always stand to one side when igniting the gasifier;
never look into the ignition opening while applying
a flame.

o If the gasifier has a tight-fitting lid, it should be
equipped with a safety release valve that will harm-
lessly lower the pressure resulting from a gas
explosion.

¢ A flame arrestor, as shown in Fig. 8-12, should be
placed at the gas mixer to prevent explosions. The
flame arrestor can also serve as a safety filter, since it
will plug rapidly when the cleanup system fails.

Openings through which fuel is loaded should be
designed to provide spill shields that will prevent
spilled fuel from falling onto hot surfaces and possibly
causing a fire. During installation, one should install
insulation, heat shields, or warning signals for workers
around the hot surfaces of a gasifier. Hot metal surfaces
can cause nasty skin burns at temperatures well below
those that will cause the metal to glow.

Widespread gasifier use would require insurance and
local fire inspectors to advise users on proper precau-
tions that might not be evident to the inexperienced.
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12.4 Environmental Hazards

The environmental problems of gasifier tars have been
discussed in previous sections. As gasifiers come into
wide use again, there will be an increased demand for
wood and agricultural residue fuels. This can have
either negative or positive environmental effects,
depending upon the approach to the problem. Selec-
tive wood cutting and use of some residues can im-
prove the forests and fields. Indiscriminate cutting and
use of all residues can lead to poor wood lot stands,
erosion, and soil depletion.

“It is essential that a careful survey of available sources
of biomass fuel, and analysis of alternative uses and
existing competing markets of these fuels, be carried
out prior to large-scale introduction of producer gas
plants.” (Kjellstrom 1983). Permanent deforestation
and desertification, which already threaten much of
our planet, can be initiated by indiscriminate use of
producer gas (Giono 1976). The use of wood for gasifier
fuel should be introduced only in regions of

fast-growing vegetation, accompanied by effective
programs of forest management and reforestation
(Kaupp 1984a).

Removing excessive quantities of biomass from
agricultural land depletes the soil, removing not only
nutrients but also reducing tilth, water permeability,
and storage ability, and leaving the bare soil exposed
to erosion by wind and water. Good soil conservation
practice requires careful regional determination of
maximum acceptable biomass removal rates, coupled
with cultivation methods that make best use of the
biomass left on the field (Lowdermilk 1975).

Erosion is more destructive than drought because
erosion destroys the soil’s water permeability. Typical-
ly, 99% of rainfall soaks into healthy soil, but only 50%
of rainfall may soak into deteriorated soil. The other
50% becomes runoff, thus making the soil twice as arid
with the same rainfall and simultaneously accelerating
the erosion process (Carter 1974).
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Chapter 13
Decision Making

13.1 Introduction

Gasifiers are technically practicable. But other criteria
also factor into decisions about their use. These criteria
include gasifier application, the availability of suitable
equipment, biomass fuel availability and fuel-source
reliability, regulations, operator availability, and of
course cost and financing.

13.2 Logistics Assessment

The decision to install, invest in, or finance a gasifier
electric project in a particular location requires a
favorable assessment of the factors that affect the prac-
ticability of the concept over the operational life of the
project.

13.2.1 Gasifier Application

Gasifier system application can range from a fuel or
chemical source for in-house or external use to a vast
complex that produces heat and electric energy as well
as inert byproducts. Generally, the system becomes
more cost-effective as the range of products increases.
Gasifier system planning at least should consider sys-
tems that are larger than needed for internal use (with
sell off of surplus product) and systems providing
cogeneration.

13.2.2 Equipment Selection Factors

Equipment needs vary with application (e.g., heat only,
fuel only) as well as the gas and location of the gasifier.
All gasifier installations, however, require control sys-
tems (however rudimentary), feedstock storage,
teedstock feeding mechanisms, and, obviously, a
gasifier. Devices for making use of the gas are also
needed. Combined with internal-combustion engines,
for example, equipment add ons for gasifiers are en-
gines and gas-cleanup equipment. In addition, electric
generators and electric power conditioners will be
needed where electricity is to be produced.

The larger issues in equipment selection are project
scope (i.e., number of uses), size, and composition (i.e.,
equipment origins and mix). Decisions depend on
available financing, on whether the equipment even-
tually will be used for purposes exceeding immediate
needs, and on the talent of available labor.

Various factors control gasifier system size. Since this
document covers only small gasifiers, the largest
project size of interest to readers of this text is limited
to a few gasifiers, each producing a maximum of
20 MBtu/day—sufficient to meet the needs of a small

housing development (fewer than 20 homes), a
moderately sized industrial complex, or a good-sized
farm.

The equipment may consist of all “home-made” com-
ponents, all individual purchases of manufactured
parts, a combination of “build and buy,” or a turnkey
purchase. Money, the availability of suitable equip-
ment, use of the equipment, and local talent enter into
decisions affecting equipment makeup and mix. Where
there’s need for an uninterrupted gas supply, choices
may include multiple small gasifiers or fewer large
ones with attendant gas storage.

13.2.3 Feedstock Supply

More than price is involved in assessing biofuels
availability. Adequate resources should be available
within a 50-mile radius to minimize concerns for an
uninterrupted supply. The types of biofuels available
are also important, and there should be alternates com-
patible with the gasifier design. Since the resource may
ultimately attract other customers, one or more long-
term contracts guaranteeing a supply is prerequisite.
Also the reputation of the suppliers to meet commit-
ments should be thoroughly verified. (As the reliability
of the supply diminishes, the size of the biofuels
storage increases.)

13.2.4 Regulations

Local and federal regulations may influence decisions
regarding the type, form, and size of a gasifier installa-
tion. The federal government’s PURPA legislation
should cause planners to recognize the potential to be
earned by selling energy to utilities and then balance
advantages against liabilities. Other regulations (e.g.,
environmental) may mandate use of particular (e.g.,
ash-disposal) equipment.

13.2.5 Labor Needs

Most individuals, given minimal training, can operate
a batch-fed gasifier that is not tied into other equip-
ment. Mechanically or electrically competent labor is
required where the gas is used to produce electric
power. Where automatic control is extensive, there will
be additional need for competence with controllers and
computers.

13.2.6 Final Logistics Considerations

Before the first piece of equipment is ordered or
financed, the success of the project should be secured
by the following:
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¢ All equipment should be checked as being of proven
design, which can be underwritten by long-term ser-
vice contracts and warranties for prime power
application.

s Systems engineering should be certified by a
reputable engineer.

» A long-term fuel purchase agreement at a firm price
with assured supply (multiple sourcing is preferred)
should be secured as critical to the project’s success.

e All zoning questions should be resolved, and all
permits, licenses, and approvals acquired.

e Liability insurance should be available and assured.

» A long-term power sales contract should be in place,
with a levelized power-price or fuel-cost escalator, if
the project includes utility buyback.

13.3 Economics

Logistics is only part of the equation determining the
practicality of gasifier implementation. The other part
is economics.

Determining the economic feasibility of a gasifier
project in a specific situation involves realistic and
site-specific estimates of capital, feedstock, labor, and
maintenance costs; the value of the electricity and heat
produced; net fossil-fuel and energy savings; and ex-
perience with similar systems. Naturally, costs can be
measured by using low-cost and no-cost burnable
wastes (provided they are available on a reliable basis),
low-cost equipment options, renewable energy tax
credits, reduced-rate financing, automatic operation,
and reduced maintenance and longer overhaul cycles.

The intent of the information that follows is to provide
a basis for understanding the elements entering into an
economic decision about investing in gasification
facilities. Potential users of gasifiers are referred to a
number of excellent references for economic assess-
ment of gasifiers (Hodam 1983; NYSERDA 1980; EIA
1983a; EIA 1983b). The circumstances affecting
economics, however, are in constant flux, and only
general rules are noted here.

13.3.1 Costs

An assessment of the overall economic feasibility of
using biomass to generate electric power should con-
sider the following factors when considering a specific
project:

e the cost of biomass fuel

e the cost of the gasifier system, including fuel storage
bins, fuel feeding devices, gas cleanup systems, a
utility connection, switchgear, and installation costs

e the cost of money at prevailing interest rates
s the cost of operating labor
e overhaul and replacement costs

e design life of the equipment between overhauls or
replacement

o the value of the power produced

e the cost of ash and tar disposal after cleanup
e retail cost of buying electricity

¢ benefits of using renewable energy.

13.3.2 Calculating Energy Costs

The cost of energy from biomass should be compared
with the costs for all other fuels with which it might
compete. The convenient fuels (electricity, gas, and oil)
are more expensive than the solid fuels (coal and
biomass). Furthermore, a reliable infrastructure is
available for transport and delivery of electricity, gas,
and oil, and reliable, inexpensive equipment is
available for their use. Biomass use in the absence of a
well-developed infrastructure is highly dependent on
the specific situation. The fact that biomass use has
doubled in the past 10 years, however, suggests that the
infrastructure and equipment for biomass use are
growing steadily.

The cost of various forms of biomass should be com-
pared using a common base. Since neither moisture nor
ash contributes to fuel value, biomass cost is mostly
quoted in dollars per ton MAF ($/ton MAF), where
MAF denotes moisture-and ash-free basis (i.e., AS IF
the biomass had its moisture and ash removed).

_ Actual Cost

Biomass Cost ($/MAF ton) = 1-M-A) (13-1)

where M is the fraction of moisture and A is the fraction
of ash in the biomass. Freshly harvested biomass often
contains a moisture fraction of 0.5. The internal ash
content of most wood is less than 0.01 (1%), but
as-delivered it may contain extraneous matter in
fractions up to 0.05 (5%). Agricultural residues contain
0.05 to 0.20 ash fractions, so the normalization
equation clearly is important in calculating biomass
costs.

The actual costs of various forms of biomass, shown in
Table 13-1, vary from a negative $20/dry ton (depend-
ing on landfill tipping fees) for landfilled burnable
residues and municipal waste to more than $100/ton
for firewood delivered in a city. Costs, in fact, depend
on many factors, which include

e quantities available

e whether the biomass is a byproduct or principal
product

o distance that the biomass must be hauled

e amount of pretreatment, sizing, drying, and storage
needed

e tax benefits for biomass use.
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Table 13-1. Typical Fuel Costs for Various Forms of Biomass (1986 $)

Cost Cost

Form Shape ($/ton) ($/MBtu)
Wet chips from tree service Chips with twigs and bark $0-20 $0-1.25
Demolition hammer milled dry Chips with slivers 10-20
Dry chips - whole tree Chips, uniform size, no twigs or slivers 20-40 1.25-2.50
Densified biomass {pellets, cubes) Uniform cubes 40-60 2.50-3.75
Cord wood Large 2 ft long irregular diameter 100-150 6.25-10.00
Municipal solid waste Very irregular shape with high-ash, Credit Credit

mixed composition -(0-20) -(0-1.25)
Refuse-derived fuel Irregular shape, composition more 0-20 (0-1.25)

more predictable than MSW

MSW or refuse-derived fuels (RDF) appear to be the
most inexpensive biomass fuels. However, they are
technically the most difficult fuels to gasify (or burn)
because of their high ash, heavy metal, and plastic
content.

13.3.3 Equipment Cost

A gasifier is basically only an empty can, so the gasifier
by itself can be a very low-capital-cost device, ranging
from $2000 to $10,000/MBtu ($40-$200/kWh). On the
other hand, a gasifier system for generating process
heat or power may cost two to eight times this amount,
depending on the auxiliary equipment required,
including

o fuel storage bins

o fuel drying, screening, and pretreatment systems

¢ devices to deliver and meter fuel fed to the gasifier
o the gasifier itself

e ash removal

* a gas cleanup system

o gasifier operational controls.

These factors are specific to the application, and they

must be evaluated economically and technically for
each application.

In many cases, a gasifier might be considered simply
as a retrofit to provide low-cost gas in place of more

expensive fossil fuels because a boiler or other
fuel-requiring device already exists. In these cases, the
economic analysis is greatly simplified because the
operation and economics of existing equipment are
already well understood.

If the gas is to be burned directly, then an equipment
comparison should be based on the cost per million Btu
per hour. A compilation of local energy costs similar to
those in Table 13-2 gives a good idea of the economic
attractiveness of a gasifier-for-heat project. Many ofthe
factors discussed for power generation will also apply
to heat generation.

The cost of a gas producer and the cleanup portion of
the gasifier system primarily should be the cost of
fabricating these units from sheet, plate, bar, and tube
stock. As the production volume for a particular design
increases, there may be some benefit from using cus-
tom stampings, spun domes and cones, and custom
castings. However, tooling costs are significant for spe-
cialized, single-purpose components, so the effect of
mass production methods can be considered minor.

Fabrication costs can be estimated by tabulating the
costs of material (e.g., pounds of sheet metal, as well as
accessory hardware and fittings), fabrication (based on
factors such as the length of cuts, number of welds and
bending operations, and amount of assembly), and
overhead expenses (Perry 1973). In addition, there will
be extra costs for completely automatic systems and for

Table 13-2. Typical Costs of Various Fuels in Colorado (1983)

Energy Typical Cost
Fuel Equivalence Typical Cost ($/MBtu)
Natural Gas 0.1 MBtu/therm $0.50/therm $5.00
Oil 6.2 MBtu/bbl $30/bbl 4.80
Electricity 3412 Btu/kWh $0.05/kWh 14.60
Coal 24 MBtu/ton $40/ton 1.66
Biomass 16 MBtu/ton $32/ton 2.00
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the engineering and development costs of the initial
design.

13.3.4 Conversion Efficiency and Fuel
Consumption

The efficiency with which the gasifier converts
biomass fuel to a final product is an important factor
for calculating operating costs. Overall efficiency is the
product of a several efficiencies, including those for
drying the biomass, operating the gasifier, and use of
the product gas.

Drying efficiency varies widely, depending on the
equipment design and heat source (see Chapters 3 and
8). A typical fuel consumption for wood with 20%
moisture is roughly 2 lb/hp-h (3 1b/kWh) (Gengas
1950).

13.3.5 The Cost of Operating Labor

Batch-fed gasifiers can be used in the lowest cost
system designs. Batch-fed gasifiers are suitable for
many situations, especially in the context of the
workplace where change of shifts, lunch, and breaks
serve as natural intervals for fueling and ash removal.
The cost of operator labor for refueling can be
determined from the equation in Table 13-3 or from the

graph in Fig. 13-1. If the gasifier requires continual
monitoring by a skilled attendant, this cost introduces
a significant economy-of-scale factor against very small
systems.

It should be possible in many situations to operate
batch-fed gasifiers with minimal attention. However, if
round-the-clock operation or minimal attendant labor
is desired, then automatic operation may be more
suitable.

Automatic fuel feeding and ash removal require addi-
tional equipment. The necessary materials and equip-
ment and reliable controls can add to the equipment
costs on a one-time basis. The costs of automatic
material handling can be compared with the expected
savings in operator labor costs using Fig. 13-1 to
evaluate whether to install the automatic equipment.

13.3.6 Maintenance Costs

Maintenance costs (as well as original-equipment
costs) increase rapidly as engine size increases. For
instance, a major overhaul on Caterpillar and Onan
100-kW natural-gas engines costs from $6000 to $9000,
representing more than a threefold increase in overall
cost per kilowatt hour over a 50-kW system.

Table 13-3. Sample Calculation of Electric Production Costs

For example at $1,000/kW equipment cost, 15% interest, operating at 80% duty cycle

Interest = C

Equipment Cost $/kW)(Loan Interest %/yr)

(1000)(15)

int = " (Duty Cycle %)(365 days/yr)(24 h/day)

(80)(365)(24) = 2.14¢/kWh

For wood chips at $24/ton, 10% H20 + Specific Fuel Consumption of 3 Ib/kWh, we get Fuel Cost

{Fuel Price $/ton){Specific Fuel Consumption Ib/kWh)

(24)(3) = 4.0¢/kWh

C =
fuel (2000 Ib/ton)[1-(moisture %/100)]

(2000)[1-(10/100)] ~

The cost of engine wear for a 50 kW engine with 2000 h engine life and $1000 rebuild cost

Rebuild Cost$  $1000
(KW)(engine life) ~ 50(2000)

Cwear -

= 1.0¢/kWh

Labor Cost for 1/2 h per 8 h shift at $5/h wage rate

Clabor =~ (Capacity kW)(Hours/Shift

(Wage Rate $/h)(Attention Hours/Shift)  (5)(0.5)  0.625¢/kWh
(50)(8)

Normal Maintenance $6.25 for 5 qt oil capacity, $15 oil analysis, $15 plugs, $8 points, 1 h labor, 200 h maintenance

interval

Parts + Labor + Qil Analysis

(6.25 + 15 + 8) + (5) + (15)

Cmaint = - - =
maint = “(LW)(maintenance interval)

(50)(200)

= 0.5¢/kWh

Total Cost of Electric Generation is the sum of the above components of production cost.

Ciotal = Cint + Ctuel + Cwear * Clabor + Cmaint

Ciotal = 2:14 +4.0 + 1.0 + 0.625 + 0.5 = 8.525 /kWh
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Furthermore, the supply of parts and service is more
limited, usually only through dealers.

13.4 Cost Benefits

Using a gasifier may be justified on the basis of a single-
purpose, in-house need. If, additional uses can be
found, the gasifier may be more attractive. Some fac-
tors to consider are sale of excess electricity and
cogeneration.

13.4.1 Value of Power Produced

PURPA requires utilities to buy back power generated
from biomass at a price equal to the utility’s full,
avoided cust for generating power. Plants with excess
capacity may be worth considering if the extra capacity
can be sold at a profit. Unfortunately, the electric utility
climate for PURPA buyback is one of confusion, misin-
formation, and frequent changes. The PURPA law does
not clearly define avoided costs and leaves interpreta-
tion to the individual states, with consequent
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Fig. 13-1. Electricity generation costs: graphic calculators for interest, wear, maintenance, fuel, and labor
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disparities between states. It is prudent to seek long-
term power purchase agreements only when and where
buyback rates are favorable.

If sale of power to utilities is being considered, it is im-
portant to determine the optimum size for a gasifier
electric-genseration project. If the buyback rate is low
and the retail rate is high, then a prospective gasifier
should be downsized. If, however, the buyback rate is
high enough, the decision may be to choose a much
larger project with expectations of making money by
selling power. Certain system sizes may be allowed
more favorable buyback rates; this therefore should be
explored. For instance, some states offer full retail
value to renewable electricity projects under certain
sizes.

Thus, it is necessary to investigate thoroughly the
PURPA and state power-generation structure before
making binding decisions on an electric power project.
The state utility commissions are listed in Table 13-4.

Optimum-size considerations for a biomass gasifier
electric project also should include available fuel sup-
ply, fuel transport equipment, and the PURPA climate.

13.4.2 Cogeneration Possibilities

The “waste” engine heat in the exhaust gases and en-
gine block coolant represents from 2/3 to 3/4 of the
energy supplied by the fuel. Using this waste heat from
an engine (cogeneration) allows a much higher degree
of energy utilization and is sometimes eligible for
additional tax credits.

The size of a cogeneration system depends, of course,
on the size of the heat load. Each kilowatt from an en-
gine generator yields arund 15,000 Bti/h in waste heat.
This heat can be applied for such applications as space
heating, greenhouse heating, grain drying, and absorp-
tion cycle refrigeration. Figure 13-2 indicates a heat
budget for cogeneration heat recovery.

13.5 Financing

Several potential funding sources exist for implement-
ing a gasifier project.

13.5.1 Government Subsidies in the Form of
Tax Incentives

The U.S. government in the past has provided tax in-
centives for using renewable-energy power. So have
some states. The situation is in flux and existing situa-
tions should be checked with an accountant or lawyer
specializing in energy issues. More general (energy) tax
breaks (e.g., accelerated depreciation) may exist even
where no special tax privileges exist for renewable-
energy projects, and these possibilities, too, should be
explored.

13.5.2 Financial Institutions

If a gasifier/generator system is being financed, inves-
tors and the financial community will require a com-
plete assessment of the project with performance
guarantees and benefits and risks clearly identified.

Interest rates in the 8% to 20% range place a high
demand on the economic yield of an installation. For
instance, 15% interest on a $1000/kW installation costs
2.5¢/kWh in interest for continuous operation. Actual
interest costs can be determined from either the equa-
tions in Table 13-3 or the graphs in Fig. 13-1. The sen-
sitivity of the generating cost to engine life, fuel cost,
and labor is indicated in graphic form in Fig. 13-1.

13.6 Other Considerations

The factors considered in this chapter directly
influence the viability of the project. Other factors,
although peripheral, are worth noting because such
important considerations as bank-loan approvals can
hinge on their perceived value. These factors include
job creation and economic benefit to the community.
New jobs can be expected because of a biomass gasifier
electric project. Also most of the operating expenses of
biomass gasifier electricity generation involve
payment directly to individuals in the local
community through fuel purchases (including
collection, preparation, handling, and transport) and

Recoverable energy
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equipment operation and maintenance. Engine
overhaul and investment earnings also may be retained
in some local economies. In such cases, virtually all the
expenses of electricity generation may generate local

wages and income to the benefit of the local economy.
From 30% to 70% of the value of electricity sales could
result in local wages in addition to earning a 15%
return on equipment investment.

Table 13-4. State Utility Commissions

Phone

Contact

State Agency
Alabama PSC
Alaska PUC
Arizona ACC
Arkansas PSC
California PUC
Colorado PUC
Connecticut DPUC
Delaware PSC
District of Columbia PSC
Florida PSC
Georgia PSC
Hawaii PSC
Idaho PUC
inois ICC
Indiana PSC
lowa ISSC
Kansas KCC
Kentucky KURC
Louisiana PSC
Maine PUC
Maryland PSC
Massachusetts DPU
Michigan PSC
Minnesota PSC
Mississippi PSC
Missouri PSC
Montana PSC
Nebraska

Nevada PSC
New Hampshire PUC
New Jersey BPU
New Mexico PSC
New York PSC
North Carolina uc
North Dakota PSC
Ohio PUC
Oklahoma OcCC
Oregon PUC
Pennsylvania PUC
Rhode Island PUC
South Carolina PSC
South Dakota PUC
Tennessee PSC
Texas PUC
Utah PSC
Vermont PSB
Virginia SCC
Washington uTcC
West Virginia PSC
Wisconsin PSC
Wyoming PSC

Institute for Local Self Reliance

205-832-3421

Wallace Tidmore

907-276-6222 Judy White
602-255-4251 Jim Apperson
501-371-1792 Dana Nixon
415-557-1159 John Quinley

303-866-4300
203-827-1553
302-736-3233
202-727-3062
904-487-2740
404-656-4141

Mike Homeac
Research Division
Leon Ryan

Bonnie Davis
Sam Weaver

808-548-3990 Leroy Yuen
208-343-3456 William Drummond
217-785-0326 Joseph Gillan
317-232-2711 William Boyd
515-281-5701 Robert Latham
913-296-5468 Eva Powers

502-564-3940
504-342-1403
207-289-3831
301-659-6021
617-727-9748
517-373-8171

Richard Heman
Arnold Chauviere

Paul Daniel
Fuel Charge Div.
Donald Johns

612-296-8662 Stuart Mitchell
601-354-7265 Keith Howle
314-751-3234

406-449-2649 Ted Otis

No Authority

702-885-3409

603-271-2437 Sarah Voll
201-648-3448 Steve Gable
505-827-3361 Tom Halpin
518-474-6515 Craig Indyke
919-733-2267 Tim Carrere
701-224-4078 Steven Kahl
614-466-7750 Alan Pound
405-521-2335 Jim Winters
503-378-7998 Leon Hagen
717-783-1373 Tim Clift
401-277-3500 Doug Hartley
803-758-5632 Randy Watts
605-773-3201 Walter Washington
615-741-2125

512-458-0202 Mike Williams
801-533-3247 Douglas Kirk
802-828-2880 Perer Zamore
804-786-4932 Bill Stevens
206-753-1096 Dick Bostwick
304-348-2174 Rich Hitt
606-266-5620 Jennifer Fagen
307-777-7472 Dave Walker
202-232-4108 David Morris
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Appendix
Units and Conversions

Although it was our intention to present the collected
data within a consistent framework of acceptable
metric units, this goal could be only partially achieved.
The still widespread use of English units did not in all
cases allow transfer of the reported data to metric units.
The internationally established gram (g), meter (m),
second (s), and joule (J) system (SI) is therefore oc-
casionally replaced by units that may be more familiar
and more convenient to the reader. Conversion factors
are from the Handbook of Chemistry and Physics, 67th
edition, CRC Press.

Weight

1 kg = 2.204 Ib = 15,432 grains = 32.105 oz {troy)
= 0.984 x 1073 ton (long) = 1.1023 x 1073 ton (short)

Pressure

1 atm = 1.0133 bar = 101.33 kPa = 14.7 psia
=29.921 in. Hg = 1419 in. H,0 = 760 mm Hg

Velocity
1 m/s = 3.281 ft/s = 3.6 km/h = 2.237 mph

Energy

1 Btu = 1.055 kJ = 252 Cal = 778.2 foot-pound-force
1 kWh = 3.6 MJ = 3413 Btu

1 Cal =4.187]

Density
1 g/cm3 = 1000 kg/m3 = 62.43 1b/ft3
1 Ib/ft3 = 0.01602 g/cm3 = 16.02 kg/m?3

Power

1 watt = 1 J/s = 3.43 Btu/h = 0.2389 Cal/s = 3.6 kJ/h
=1.341x 103 hp

Temperature
K="C+273.15

‘R="°F +459.67 =1.80 K
‘F=1.8C+ 32

Concentration
1 grain/ft3 = 2.571 g/m?3

Length
1in.=2.54 cm

1 micron (micrometer) =1 ym = 105 m

Volume compressible gas (1 atm dry)

1 Nm3 (0°C) = 38.55 scf (77°F) = 37.32 scf (60°F)
= 37.90 scf (68°F)

Volume noncompressible

1 m3 = 35.315 ft3 = 1000 liters

1 ft3 = 0.02831 m3 = 7.48 gal

1 gal (U.S.) = 3.785 liters = 0.1336 ft3 = 231 in.3

1 liter = 0.353 ft3 = 0.2642 gal = 33.82 fl 0z = 61.02 in.?

Flow gas
1 Nm3/h = 0.632 scfm (68°F)

Area
1 m? = 10.76 ft2 = 1550 in.2 = 1.30 yd?

Hearth Load (for 130 Btu/scf gas)

0.9 Nm3/h-cm? = 537 scfm/ft? = 3.73 scfm/in.2
= 4.2 MBtuw/h-ft3

Gas Energy Content

1 Btu/scf (68°F) = 9.549 kCal/Nm? (0°C)
=39.98 kJ/Nm? (0°C)

Fuel Energy

1 Btu/lb = 0.5555 Cal/g = 2.326 J/g

1 Cal/g = 1.8 Btu/lb = 4.187 J/g

Gasification Rules of Thumb — approximately true
Fuel Consumption

1 hp-h = 2 Ib biomass = 2 scfm gas

1 kWh = 3 1b biomass

1 ppm = 1 mg/m3
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Standard conditions for gas measurements vary wide-
ly depending on the organization setting the standards.
Since gas properties can be measured to a precision
considerably finer than the error introduced by a
misunderstanding in standard conditions, expressions
of gas volume and heating value must specify the
measurement conditions.

Two often-used references (Kaupp 1984a; Gengas 1950)
use gas measurement conventions different from com-
mon gas practice. The following table is provided to

indicate the sources of gas measurement conventions.
However, the reader should exercise extreme caution
in all conversions using the term "scf." If the gas
measurement conditions are not specified or cannot be
safely presumed, then gas heating value is subject to
3% uncertainty.

We have used 1 atm =101.32 kPa = 760 mm = 29.92 in.
Hg for a standard pressure and a temperature of 20°C =
68°F in our data reporting.

Standard Measuring Conditions for Gases

Standard Conditions

Unit Where Used Pressure Temperature Reference
scm/scf EPA Method 5 Dry gas 760 mm = 1 atm 68°F = 20°C 0
scf American Gas Association 762 mm = 30 in. Hg dry gas 60°F = 15 5/9°C (2)
scf Compressed Gas Institute 760 mm - 1 atm 29.92 in. Hg 68°F = 20°C
Nm3 NTP = STP 760 mm = 1 atm dry gas 0°C
Nm3 Standards Council of Canada 1 atm dry 15°C (34)
ft3 Environment Canada Air 29.92in. Hg 537°R (5)
Pollution Control 78°F
Directorate
scf ANSI Standard Z132.1 Saturated with water 14.73 psia 60°F (6)
for gas calorific value
Molecular weight standard 29.92 in. Hg = 101.325 kPa 0°C = 32°F (7)
conditions 760 mm =1 atm dry gas
scf Kaupp and Goss 1 atm 77°F = 25°C (8)
Nm3 Generator Gas 1 atm dry 0°C dry (9)
scf 1 atm dry 70°F = 21.11°C

(1) Environmental Protection Agency, “Determination of Particulate Emissionsfrom Stationary Sources,” Codified Federal Register 40, Pt. 60, Appendix

A,Method 5.

2) Dictionary of Scientific and Technical Terms, 3rd Ed. New York: McGraw-Hill, 1984.
3) International Standards Organization, Measurement Conditions for GaseousFuels, 1ISO STD 5024-TC28, 1979.

4) ENFOR Project C172, DDS File No. 41SS, KL229-1-4117, 1979.

7) Determining the Properties of Fine Particulate Matter, ASME PTC-28, 1965.
8) Kaupp, A. and Goss, J.R. State-of-the-Art for Small (2-50 kW) Gas Producer-Engine Systems. Final Report to USDA, Forest Service, 1981.

9

2

3)

4)

(5) Environment Canada Air Pollution Directorate, EPS 1 AP.74-1, 1974.

(6) Calorific Value of Gases in Natural Gas Range by Continuous RecordingCalorimeter. ASTM D 1826, 1977.
7

(8)

@

Generator Gas: The Swedish Experience from 1939-1945. SERI/SP-33-140, 1979 (Solar Energy Research Institute, Golden, Colo).
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